WAITE INSTITUTE

# BIOCHEMICAL ASPECTS OF DENITRIFICATION

IN THIOBACILLUS DENITRIFICANS

by

Veena Sawhney M.Sc. (Hons.)

A thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy

Department of Agricultural Biochemistry Waite Agricultural Research Institute The University of Adelaide

> May 1978 Awarded August 1978

#### PREFACE

Some of the results presented in this thesis have been published in following scientific journals:-

 Sulphite- and NADH-dependent nitrate reductase from Thiobacillus denitrificans

V. Sawhney and D.J.D. Nicholas,

J. Gen. Microbiol. 100, 49-58 (1977).

- 2. Some properties of a nitrate reductase from Thiobacillus denitrificans.
  V. Sawhney and D.J.D. Nicholas,
  Proc. Aust. Biochem. Soc. 10, 13 (1977).
- Sulphide-linked nitrite reductase from Thiobacillus denitrificans with cytochrome oxidase activity: Purification and properties.
   V. Sawhney and D.J.D. Nicholas,

J. Gen. Microbiol. 106, 119-128 (1978).

- 4. Studies on sulphide-linked nitrite reductase from *Thiobacillus denitrificans*.
  V. Sawhney and D.J.D. Nicholas, *Proc. Aust. Biochem. Soc. 11*, 23 (1978).
- 5. Fegulation of NADH-linked nitrate reductase by NAD<sup>+</sup>
  and 5'-AMP in *Thiobacillus denitrificans*.
  V. Sawhney and D.J.D. Nicholas,
  FEMS Letters (in press).

#### ACKNOWLEDGEMENTS

I wish to express my deepest and sincerest thanks to my supervisor, Professor D.J.D. Nicholas, Chairman of the Department of Agricultural Biochemistry, W.A.R.I., University of Adelaide, for his constant encouragement and inspiring guidance throughout the progress of this work and for the valuable advice during the preparation of the manuscript.

Assistance of Mr Geoffrey Clark in the culturing of the bacterium and Mr David Hein for his help in the experiments with Mass Spectrometer is hereby acknowledged. I would also like to express my appreciation to Ms Tommie Siekmann for the skillful typing of the thesis.

Finally I am grateful to my husband Dr S.K. Sawhney for his patience and unfailing encouragement during the course of this investigation.

# DECLARATION

I hereby declare that the work presented in this thesis has been carried out by myself between January 1975 to December 1977. To the best of my knowledge no material given herein has been submitted in any form for any degree in any university. The abbreviations for chemicals and symbols in general follow either the tentative rules of the IUPAC-IUB Commission on Biochemical Nomenclature (*Biochem. J.* (1966) 101, 1-7) or the Instructions to Authors for the *Biochemical Journal (Biochem J.* (1973) 131, 1-20).

## Chemicals

5'-AMP	adenosine 5'-phosphate
3 - AMP	adenosine 3'-phosphate
ADP	adenosine 5'-diphosphate
ATP	adenosine 5'-triphosphate
5'-CMP	cytidine 5'-phosphate
5'~GMP	guanosine 5'-phosphate
5'-TMP	thymidine 5'-phosphate
5 <b>'-</b> UMP	uridine 5'-phosphate
CTP	cytidine 5'-triphosphate
GTP	guanosine 5'-triphosphate
UTP	uridine 5'-triphosphate
BVH	benzyl viologen (reduced)
DEAE-cellulose	diethylaminoethyl cellulose
DNase	deoxyribonuclease
DCIP	2,6-dichlorophenol indophenol
FAD	flavin-adenine dinucleotide
FMN	flavin mononucleotide
GSH	glutathione (reduced)
HQNO	2-heptyl-4-hydroxyquinoline-N-oxide
NAD	nicotinamide-adenine dinucleotide, oxidized

iv.

NADH	2	nicotinamide-adenine dinucleotide, reduced
NADP <sup>+</sup>		nicotinamide-adenine dinucleotide phosphate, oxidized
NADPH		nicotinamide-adenine dinucleotide phosphate, reduced
NEM		N-ethylmaleimide
p-CMB		p-chloromercuribenzoic acid
PMS		phenazine methosulphate
SDS		sodium dodecyl sulphate
-SH		sulphydryl
TMPD		N,N,N',N'-tetramethylparaphenylenediamine

Symbols and Units

A	absorbance
A <sup>O</sup>	angstrom
ΔA	change in absorbance
cm	centimetre
Е	extinction
E' 0	standard electrode potential at a given pH
a	unit of gravitational field
HCl	hydrochloric acid
hr	hour(s)
kc	kilocycle
kcal	kilocalories
ĸm	Michaelis constant
۶.	litre
м	Molar
mg	milligram
min	minute(s)

v.

m,L	mililitre
mmole	millimole
mM	millimolar
mV	millivolts
MVH	methyl viologen (reduced)
N .	normal
nm	nanometer
nmole	nanomole
8	per cent
K <sub>3</sub> Fe(CN) <sub>6</sub>	potassium ferricyanide
p.s.i.	pound per square inch
sec	second(s)
µl ·	microlitre
µmole	micromole
μΜ	micromolar
U.V.	ultraviolet
v	volts
S10	supernatant fraction left after centrifuging broken cells at $10,000 \times \underline{g}$ for 30 min
S18	supernatant fraction obtained after centrifuging disrupted cells at $18,000 \times \underline{g}$ for 45 min
S100 and P100	supernatant and pellet fractions, respectively, obtained after centrifuging S10 or S18 at 100,000x <u>g</u> for 90 or 120 min

Temperatures are expressed as degrees Centigrade (<sup>0</sup>)

vi.

TABLE OF CONTENTS

PREF	TACE			195	i
ACKN	IOWLEI	GEMENTS	S		ii
DECI	ARATI	ION			iii
ABBF	REVIA	TIONS			iv
TABI	LE OF	CONTENI	"S		vii
LIS1	IS OF	TABLES			xiii
TTC		FTCIIDEC	-		vui
	15 01	FIGUNES			VAT
SUMM	MARY		а т		xx
1.	INTRO	DUCTION	3		1
	1.1	Chemoli	thotrophic bacteria		1
		1.1.1	Sulphur oxidizing bacteria - Thiobacillus		2
	1.2	Denitri	ification		5
		1.2.1	Reduction of nitrate to nitrite		8
		1.2.2	Regulation of nitrate reductase		13
		1.2.3	Dissimilatory nitrite reductase		15
		1.2.4	Nitric oxide reductase and nitrous oxide reductase		17
	1.3	Aim of	thesis		18
2.	MATE	RIALS AN	ND METHODS		19
	2.2	Prepara	ation of solutions and buffers		19
	2.2	Biolog:	ical materials		19
		2.2.1	Bacterium		19
		2.2.2	Enzymes		19
	2.3	Chemica	als and other materials		20
		2.3.1	Nucleotides and marker proteins		20
		2.3.2	Other chemicals		20
		2.3.3	Chromatographic materials		20

Page

viii.

2.4	Cultur	e of the bacterium	2]
	2.4.1	Growth of the organism	21
	2.4.2	Maintenance of the culture	21
	2.4.3	Harvesting	21
2.5	Enzymi	c techniques	21
	2.5.1	Preparation of acetone-dried cells	21
	2.5.2	Preparation of homogenates and cell-free extracts	23
		2.5.2.1 Cell-free extracts of fresh cells	23
		2.5.2.2 Cell-free extracts of acetone-dried cells	23
		2.5.2.3 Preparation of S100 and P100 fractions	24
	2.5.3	Solubilization of nitrate reductase	24
	2.5.4	Enzyme assays	25
		2.5.4.1 Nitrate reductase	25
		2.5.4.2 Nitrite reductase	26
		2.5.4.3 Cytochrome oxidase	27
	2.5.5	Determination of Michaelis - Menten constants (K <sub>m</sub> )	28
	2.5.6	Determination of molecular weights	28
		2.5.6.1 Nitrite reductase	28
		2.5.6.2 Cytochrome c <sub>554</sub>	29
	2.5.7	Isolation of cytochrome $c_{554}^{}$ and flavin from nitrate reductase	29
	2.5.8	Dissociation of nitrite reductase into $cytochrome c$ and haem $d$ components	30
2.6	Genera	l techniques	30
	2.6.1	Oxygen electrode	30
	2.6.2	Irradiation of the samples with U.V. light	31
	2.6.3	Absorption spectra	31.
	2.6.4	Redox-potential of cytochromes	32
	2.6.5	Paper chromatography of flavins	33
	2.6.6	High-voltage electrophoresis of flavins	34
	2.6.7	Mass spectrometry	34
	2.6.8	Preparation of columns	35
		2.6.8.1 DEAE-cellulose	35
		2.6.8.2 Sephadex and Sepharose 6B	35
	260	Mombrane ultrafiltration	26

# Page

				Page
2.7	Chemic	al determ:	inations	36
	2.7.1	Protein		36
	2.7.2	Nitrite		37
	2.7.3	Copper a	nd iron	37
	2.7.4	Haems $d$ a	nd c	38
RESU	LTS			39
3.1	Nitrat	e reducta:	se	39
	3.1.1	Distribut reductase	tion of sulphite-linked nitrate	39
	3.1.2	Enzyme ad	ctivity with various electron donors	39
	3.1.3	Compariso and NADH	on of the properties of sulphite- -linked nitrate reductase	39
	i. I	3.1.3.1	Nitrate reduction with either sulphite or sulphite plus NADH as the electron donor	39
		3.1.3.2	Distribution	42
		3.1.3.3	Incubation time, stability and pH optima	42
		3.1.3.4	K <sub>m</sub> values	46
	3.1.4	Characte: chain in	rization of the electron transport volved in nitrate reduction	50
		3.1.4.1	Effects of inhibitors	50
	1	3.1.4.2	Effect of U.V. irradiation	58
	3.1.5	Spectral linked n	properties of sulphite- and NADH- itrate reductase	62
	3.1.6	Characte chain li	rization of the electron transport nked to oxygen	62
		3.1.6.1	Various electron donors	62
		3.1.6.2	Comparison of oxygen uptake in various cell fractions with sulphite or NADH as the reductant	s 62
		3.1.6.3	Effects of various inhibitors	66
	3.1.7	Solubili reductas	zation of-sulphite-linked nitrate e	66
c.	3.1.8	Inactiva on solub activity	tion of NADH-linked nitrate reductase ilization and restoration of its by PMS and menadione	70
	3.1.9	Purifica reductas purified	tion of sulphite-linked nitrate e and properties of the partially enzyme	73

3

# ix.

Page

x.

		3.1.9.1	Purification	73
		3.1.9.2	Properties	77
	3.1.10	Character: flavin iso	ization of cytochrome $c$ and plated from nitrate reductase	81
		3.1.10.1	Cytochrome c	81
		3.1.10.2	Flavin	86
	3.1.11	Regulation	n of NADH-linked nitrate reductase	90
		3.1.11.1	Inhibition by NAD <sup>+</sup> and its reversal by NADH and menadione	90
		3.1.11.2	Effect of various adenine nucleotides	92
		3.1.11.3	Effect of adenylate charge on NADH- and sulphite-linked nitrate reductase activities	97
		3.1.11.4	Effects of cations on the reversal of inhibition of NADH-linked nitrate reductase by NAD <sup>+</sup> and AMP	97
3.2	Nitrite	e reductas	e	99
	3.2.1	Distribut reductase	ion of sulphide-linked nitrite	99
	3.2.2	Various e	lectron donors	99
		3.2.2.1	Sulphur compounds	99
		3.2.2.2	Other reductants	99
	3.2.3	Optimum c	onditions for enzyme activity	103
		3.2.3.1	Effect of incubation time	103
		3.2.3.2	Effect of enzyme concentration	103
		3.2.3.3	Effect of pH	107
	54	3.2.3.4	Heat stability	107
× =	3.2.4	${\tt K}_{\tt m}$ values	for sulphide and nitrite	107
	3.2.5	Character chain inv	ization of the electron-transport olved in nitrite reduction	107
		3.2.5.1	Effects of inhibitors	107
		3.2.5.2	Spectral properties	117
	3.2.6	Character chain lin	ization of the electron-transport ked to oxygen	117
		3.2.6.1	Comparison of oxygen uptake and nitrite reduction with various electron donors	117
		3.2.6.2	Effects of various inhbitors on oxygen uptake with sulphide as the reductant	120

			Page
3.2.7	Purificat reductase	tion of sulphide-linked nitrite	124
	3.2.7.1	Solubilization	124
	3.2.7.2	Purification	124
3.2.8	Propertie	es of the purified enzyme	129
	3.2.8.1	Molecular weight	129
	3.2.8.2	Copper and iron contents	132
	3.2.8.3	Relative concentrations of haems $c$ and $d$ in the purified nitrite reductase	132
	3.2.8.4	Sequence of reduction and oxidation of cytochromes $\boldsymbol{c}$ and $\boldsymbol{d}$	132
	3.2.8.5	Difference in redox-potential of cytochromes $c$ and $d$	135
	3.2.8.6	Electron donors	135
	3.2.8.7	Effects of mammalian and bacterial cytochromes $c$ on nitrite reductase activity	139
3.2.9	Cytochron nitrite	ne oxidase activity of the purified reductase	143
	3.2.9.1	Comparison of nitrite reductase and cytochrome oxidase activities	143
3.2.10	Spectral reductase	properties of purified nitrite	146
	3.2.10.1	Absorption spectra	146
	3.2.10.2	Effects of cyanide and carbon monoxide	152
	3.2.10.3	Pyridine haemochromogen	152
3.2.11	Dissocia into cyto	tion of purified nitrite reductase pochromes $c$ and $\overline{d}$	155
	3.2.11.1	Spectral properties of dissociated cytochrome $c$	155
	3.2.11.2	Spectral properties of haem $d$	159
3.2.12	Products enzyme	of nitrite reduction by purified	162
3.2.13	Effect of reductase	f NO gas on purified nitrate e	162
	3.2.13.1	Formation of NO-enzyme complex	162
	3.2.13.2	Dissociation of NO-enzyme complex	167
	3.2.13.3	Effects of pH, carbon monoxide and cyanide on the formation and dissociation of NO-enzyme complex	167
	3.2.13.4	Effect of NO on dissociated cytochrome $c$ and haem $d$	169

4. DISCUSSION
4.1 Nitrate reductase
4.2 Nitrite reductase
5. BIBLIOGRAPHY
193

Page

# LIST OF TABLES

Τĉ	ble		Page
	1.	Free energy production with various compounds as terminal electron acceptors in bacteria	6
	2.	Electron donors, electron carriers and products of dissimilatory nitrate reductase in various organisms	10
ō	3.	Composition of culture solutions of <i>Thiobacillus</i> denitrificans	22
	4.	Distribution of sulphite-linked nitrate reductase in various cell fractions	40
	5.	Nitrate reductase activity with various electron donors	41
	6.	Nitrate reductase activity with either sulphite, NADH or sulphite plus NADH as the electron donor	43
	7.	Distribution of sulphite- and NADH-linked nitrate reductase in various cell fractions	44
	8.	Stability of sulphite- and NADH-linked nitrate reductase during storage and the restoration of the NADH-dependent activity with GSH	47
	9.	Effects of inhibitors of -SH groups on nitrate reductase activity with either sulphite or NADH as the electron donor	53
3	10.	Effects of flavin inhibitors on nitrate reductase activity with either sulphite or NADH as the electron donor	55
]	1.	Effects of inhibitors of electron transport chain on nitrate reductase activity with either sulphite or NADH as the electron donor	56
J	L2.	Effects of metal chelators and other inhibitors on nitrate reductase activity with either sulphite or NADH as the electron donor	57
ן	L3,	O <sub>2</sub> uptake in S10 fraction in the presence of various electron donors	64
]	14.	O <sub>2</sub> uptake by various cellular fractions with either sulphite or NADH as the electron donor	65
3	L5.	Effects of sulphydryl group inhibitors on O <sub>2</sub> uptake with sulphite as an electron donor	67

xiii.

Table		Page
16.	Effects of inhibitors of the electron transport chain on $O_2$ uptake with sulphite as an electron donor	68
17.	Effects of metal chelators and other inhibitors on $O_2$ uptake with sulphite as an electron donor	69
18.	Effects of sodium deoxycholate treatment on nitrate reductase activity with either sulphite or NADH as an electron donor	74
19.	Restoration of the NADH-linked nitrate reductase activity in the deoxycholate treated Pl00 fraction	75
20.	Purification of sulphite-linked nitrate reductase	80
21.	Stability of the partially purified enzyme on storage for 36 hr at $5^{\circ}$ , $0^{\circ}$ and $-15^{\circ}$	83
22.	Effect of monovalent and divalent cations on the inhibition by NAD <sup>+</sup> and 5'-AMP of NADH-linked nitrate reductase activity	98
23.	Distribution of sulphide-linked nitrite reductase	100
24.	Nitrite reductase activity with various sulphur compounds as the electron donors	101
25.	Nitrite reductase activity with various electron donors	102
26.	Effects of sulphydryl group inhibitors on nitrite reductase activity	113
27.	Effects of inhibitors of the electron transport chain on nitrite reductase activity	114
28.	Effects of arsenite, azide and cyanide on nitrite reductase activity	115
29.	Effect of various metal chelators on nitrite reductase activity	116
30.	Comparison of various electron donors on $O_2$ uptake and nitrite reduction	119
31.	Effects of inhibitors of -SH groups on $O_2$ uptake with sulphide as an electron donor	121
32.	Effects of inhibitors of electron transport chain on sulphide-linked O <sub>2</sub> uptake	122
33.	Effects of metal chelators and other inhibitors on sulphide-linked O <sub>2</sub> uptake	123

mable

Table	e e	Page
34.	Purification of nitrite reductase	131
35.	Iron contents of purified nitrite reductase (cytochrome $cd$ complex), dissociated cytochrome $c$ and haem $d$	133
36.	Relative concentrations of haem $c$ and haem $d$ in the purified nitrite reductase	134
<del>,</del> 37.	Difference in redox-potential of cytochromes $c$ and $d$ of the purified nitrite reductase	137
38.	Effect of various electron donors on nitrite reductase activity in purified enzyme	138
39.	Effect of various electron mediators (dyes) on nitrite reductase activity in purified enzyme	140
40.	Effect of cytochrome <i>c</i> from various sources on nitrite reductase activity	141
41.	Cytochrome oxidase activity at various purification stages of nitrite reductase	144
42.	Comparison of nitrite reductase and cytochrome oxidase activities in purified nitrite reductase	145
43.	Effect of pH on the absorbance characteristic of cytochrome $d$ component of sulphide-reduced nitrite reductase	151

xv.

# LIST OF FIGURES

Figure		Page
l.	Effect of incubation time on nitrate reduction	45
2.	Thermal stability of nitrate reductase	48
3.	Effect of pH on nitrate reductase activity	49
4.	Double-reciprocal plot of the effect of nitrate concentration on enzyme activity	51
5.	A Lineweaver - Burk's plot of the effect of various concentrations of electron donor on nitrate reductase activity	52
6.	Double-reciprocal plot of the effect of potassium chlorate at varying concentrations of nitrate on nitrate reductase activity	59
7.	Effect of U.V. irradiation on nitrate reductase activity in SlO and PlOO fractions	60
8.	Effect of various concentrations of menadione on nitrate reductase activity in the U.V. irradiated Pl00 fraction	61
9.	Difference spectra of P100 fraction	63
10.	Effect of various concentrations of sodium deoxycholate on the solubilization of the sulphite-linked nitrate reductase in P100 fraction	71
11.	Effect of incubation time on the solubilization of the sulphite-linked nitrate reductase from PlOO fractio by sodium deoxycholate treatment	72 n
12.	Restoration by menadione of NADH-linked nitrate reductase activity in the deoxycholate treated P100 fraction	76
13.	Elution profile of nitrate reductase from Sepharose 6B column	78
14.	Effect of protein concentration on the sulphite-linked nitrate reductase in partially purified enzyme	80
15.	Thermal stability of partially purified sulphite-linked nitrate reductase	84

xvi

Spectra of dithionite-reduced cytochrome c554, 16. 85 separated from solubilized nitrate reductase 17. Effect of various concentrations of cytochrome 87  $c_{554}$  on nitrate reductase activity Effect of different concentrations of mammalian 18. 88 cytochrome c on nitrate reductase activity Paper chromatography of flavin isolated from nitrate 19. 89 reductase 20 High-voltage electrophoresis of isolated flavin 89 Effect of various concentrations of NAD<sup>+</sup> on NADH-21. 91 linked nitrate reductase activity Effect of NAD //NADH ratio on NADH-linked nitrate 22. 91 reductase activity Effect of NAD<sup>+</sup> on the NADH-linked nitrate reductase 23. 93 at various concentrations of NADH Double-reciprocal plot of the effect of NAD  $\tau$  at 24. 94 varying concentrations of NADH on nitrate reductase activity 25. Reversal by menadione of the inhibition of the NADH-95 linked nitrate reductase by NAD<sup>+</sup> Effect of 5'-AMP on nitrate reductase activity 26. 96 27. Effect of various concentrations of menadione on 104 nitrite reductase activity with NADH as an electron donor 28. Effect of incubation time on the sulphide-linked 105 nitrite reductase activity 29. Effect of varying protein concentrations on the 106 sulphide-linked nitrite reductase activity 30. Effect of pH on the sulphide-linked nitrite reductase 108 31. Thermal stability of the sulphide-linked nitrite 119 reductase A Lineweaver - Burk's plot of the effect of varying 32. 110 sulphide concentrations on nitrite reductase activity 33. Double-reciprocal plot of the effect of varying 111 nitrite concentrations on enzyme activity

Figure

Page

xviii.

Figure

Page

34.	Sulphide-reduced versus oxidized difference spectra of P100 fraction	118
35.	Effect of sodium deoxycholate treatment on reduced versus oxidized spectra of S100 fraction prepared from acetone-dried cells	125
36.	Elution profile of nitrite reductase from a DEAE- cellulose column	127
37.	Dithionite-reduced versus oxidized difference spectra of cytochromes $c_{551}^{}$ and $c_{554}^{}$	128
38.	Elution profile of nitrite reductase from Sephadex G-100 column	130
39.	Rate of reduction of cytochromes $c$ and $d$ of $cd$ complex of purified nitrite reductase	136
40.	Rate of oxidation of cytochromes $c$ and $d$ of $cd$ complex of purified nitrite reductase with nitrite	136
41.	Effect of various concentrations of cytochrome $c_{551}$ on the activity of the purified nitrite reductase	142
42.	Reduced cytochrome $c_{551}$ and mammalian cytochrome $c$ as electron donors for the activity of purified nitrite reductase	142
43.	Absolute and difference spectra of purified nitrite reductase	147
44.	Time course of reduction of the cytochrome <i>cd</i> complex of purified nitrite reductase with sulphide	148
45.	Time course for the reduction of cytochrome <i>cd</i> complex of purified nitrite reductase with ascorbate plus PMS	148
46.	Oxidation of sulphide-reduced <i>cd</i> complex of purified nitrite reductase with nitrite	150
47.	Effect of cyanide on the spectra of purified nitrite reductase	153
48.	Effect of CO on the spectra of purified nitrite reductase	153
49.	Spectra of purified nitrite reductase in the presence of alkaline pyridine and sodium dithionite	154
50.	Absolute and sulphide-reduced <i>versus</i> oxidized spectra of SDS-treated nitrite reductase	154

xix.

C	Iguie	<u>-</u>	aye
	51.	Separation of cytochromes $c$ and $d$ on Sephadex G-100 column	156
	52.	Elution profile of cytochromes $c$ and $d$ on Sephadex G-100 column	157
	53.	Absolute and dithionite-reduced versus oxidized, difference spectra of dissociated cytochrome $c$	158
	54.	Absorption spectra of sulphide-reduced dissociated cytochrome $c$	160
	55.	Difference spectra of alkaline pyridine haemochromogen of cytochrome $c$ , dissociated from cytochrome $cd$ complex	161
	56.	Absolute and dithionite-reduced versus oxidized spectra of haem $d$	163
	57.	Sulphide-reduced versus oxidized difference spectra of haem $d$	164
	58.	Effect of incubation time on the production of various nitrogenous gases from NO $_2$ by purified nitrite reductase	165
	59.	Effect of NO gas on the spectra of purified nitrite reductase reduced with sulphide	166
	60.	Effect of sulphide and dithionite on the spectra of purified nitrite reductase treated with NO gas	166
	61.	Effect of incubation time on the formation of complex between NO and sulphide-reduced purified enzyme	168
	62.	Formation of complex between NO and sulphide-reduced purified nitrite reductase at pH 10.2	168
	63.	Effect of pH on the formation of complex between NO and purified nitrite reductase	170
	64.	Effect of CO on the formation and dissociation of complex between NO and sulphide-reduced purified enzyme	170
	65.	Effect of cyanide on formation and dissociation of complex between NO and sulphide-reduced purified nitrite reductase	171
	66.	Effect of NO on the spectra of sulphide-reduced cytochrome c, dissociated from cd complex of purified nitrite reductase	171

Figure

#### SUMMARY

- This thesis embodies results of an investigation on some biochemical aspects of denitrification in a chemolithotrophic bacterium, *Thiobacillus denitrificans*, grown under anaerobic conditions.
- 2. In this bacterium nitrate reductase, nitrite reductase and O<sub>2</sub> uptake were all associated with membrane fractions and their activities were coupled to the oxidation of inorganic sulphur compounds.
- 3. In the pellet fraction (Pl00), nitrate reductase utilized either sulphite or NADH as a reductant. Both the sulphite- and NADHlinked nitrate reductase activities were affected by inhibitors of the respiratory chain. Reducing equivalents from either sulphite or NADH were transferred to nitrate via an electron transfer system consisting of flavins, quinones and cytochromes b and  $c_{554}$ .
- 4. A sulphite-dependent nitrate reductase was solubilized with sodium deoxycholate. The NADH-dependent activity which was lost after this treatment was restored on adding menadione.
- 5. A requirement for cytochrome  $c_{554}$  for nitrate reduction was established from the spectral changes for this cytochrome in the Pl00 fraction on reduction with either sulphite or NADH and its reoxidation with nitrate. A cytochrome  $c_{554}$  and FAD were isolated from the solubilized nitrate reductase. This cytochrome had a molecular

weight of 13,000 daltons, a redox-potential of +.223 V at pH 7.0 and it functioned as an effective electron donor for nitrate reduction.

- 6. Data are presented for the involvement of a common respiratory chain for the reduction of nitrate and the uptake of  $O_2$ . A scheme is presented for the transfer of electrons from sulphite and NADH to both nitrate and  $O_2$ .
- 7. The NADH-linked nitrate reductase was inhibited by NAD<sup>+</sup> and 5'-AMP and their effects were reversed with menadione. The inhibition of NADH-linked nitrate reductase by NAD<sup>+</sup> was dependent on the NAD<sup>+</sup>/NADH ratio. The positive cooperative influence of NADH on enzyme activity was increasingly suppressed by higher concentrations of NAD<sup>+</sup>. It is proposed that NAD<sup>+</sup> and 5'-AMP regulate the NADHlinked nitrate reductase activity to enable the bacterium to conserve NADH for various anabolic reactions.
- 8. Nitrite reductase activity in this bacterium is linked to the oxidation of sulphide. The nitrite reductase was dissociated from cell membranes in acetone-treated cells. The solubilized enzyme was purified twenty-fold. Sulphide-linked O<sub>2</sub> uptake was detected in the Pl00 fraction as well as in the purified enzyme.
- 9. Spectral studies of the Pl00 fraction and the purified enzyme confirmed the involvement of cytochromes c and d for the reduction of nitrite and for  $O_2$  uptake. Neither the reduction of nitrite nor the uptake of  $O_2$  was affected by the inhibitors of the electron transport chain.

- 10. In the purified nitrite reductase, cytochromes c and d were present in a molar ratio of 1:1. The sequence of electron transfer from sulphide to nitrite via the cytochrome cd complex was established. The redox-potential of cytochrome d was +22 mV more positive than that of cytochrome c.
- 11. Cytochromes  $c_{551}$  and  $c_{554}$  were isolated from this bacterium. Cytochrome  $c_{551}$  stimulated both nitrite reductase and  $O_2$  uptake by the purified enzyme whereas either cytochrome  $c_{554}$  from *T. denitrificans* or mammalian cytochrome *c* was without effect. Reduced cytochrome  $c_{551}$  functioned as an electron donor for the reduction of nitrite and  $O_2$  in the purified enzyme. It is proposed that this cytochrome functions in the sequence of electron transfer from sulphide to nitrite and  $O_2$ .
- 12. NO and N<sub>2</sub>O were identified as the products of the purified nitrite reductase.
- 13. An absorption band at 573 nm which appeared during the oxidation of the sulphide-linked nitrite reductase, was identified as a complex between NO and the enzyme. The effects of NO on the spectra of the purified enzyme were also determined.
- 14. Cytochromes c and haem d were dissociated from the purified nitrite reductase by SDS treatment. These components were separated by chromatography on a Sephadex G-100 column. Spectral studies were made on the dissociated cytochrome c and haem d.
- 15. On the basis of these investigations an integrated scheme of electron transfer for the reduction of nitrate and nitrite, coupled to the oxidation of sulphite and sulphide respectively, is presented.

xxii.

## 1. INTRODUCTION

The generation of metabolically utilizable energy, required for the diverse anabolic reactions, is of fundamental importance for the survival and growth of every living organism. Different groups of bacteria, the simplest prokaryotic microorganisms, fulfil their energy requirements through a number of diverse and varied mechanisms. Depending on their primary source of energy and their mode of life, bacteria have been classified as chemolithotrophs, chemoorganotrophs, photolithotrophs and photoorganotrophs (Rittenberg, 1969; Kelly, 1971; Doelle, 1975). Among these various groups, chemolithotrophs and photolithotrophs are autotrophic organisms because they are capable of fixing  $CO_2$  and thus do not rely on any organic source of carbon. In contrast, chemoorganotrophs and photoorganotrophs are heterotrophs and depend on an exogenous supply of carbon compounds, which are more reduced than  $CO_2$ , for their growth and development.

## 1.1 Chemolithotrophic bacteria

Chemolithotrophs are unique since they derive their energy needs via the oxidation of simple inorganic compounds. The occurrence of chemolithotrophic bacteria was first reported by Winogradsky in 1890. Since then several microorganisms belonging to this group have been identified. These autotrophic bacteria incorporate CO<sub>2</sub> into their cellular constituents and are able to thrive on a simple inorganic salt medium containing an appropriate oxidizable substrate, e.g. reduced sulphur compounds for *Thiobacilli* (Kelly, 1968; Hutchinson *et al.*, 1969; Roy and Trudinger, 1970), ammonium for *Nitrosomonas* (Peck, 1968; Wallace and Nicholas, 1969; Aleem, 1970) and ferrous iron

1.

WAITE INSTITUTE LIBRARY for Ferrobacillus ferrooxidans (Schlegel, 1960) and Thiobacillus ferrooxidans (Peck, 1968). The physiology and energy metabolism of these bacteria have been extensively reviewed by Peck (1968), Rittenberg (1969), Kelly (1971) and Aleem (1970, 1977).

#### 1.1.1 Sulphur oxidizing bacteria - Thiobacilli

Thiobacilli, a group of sulphur oxidizing bacteria, were first discovered by Nathansohn (1902). These rod-shaped, motile, non-sporulating, gram-negative bacteria are found in soil, sea and fresh water. The genus thiobacillus includes *T. novellus* (Vishniac and Santer, 1957), *T. perometabolis* (London and Rittenberg, 1967), *T. ferrooxidans* (Tabita and Lundgren, 1971; Shafia *et al.*, 1972) and *T. denitrificans* (Taylor *et al.*, 1971; Taylor and Hoare, 1971). The physiology and various biochemical aspects of thiobacilli have been reviewed by Vishniac and Santer (1957), Lees (1960, 1962), Peck (1962) and Kelly (1968, 1971).

Thiobacilli are capable of oxidizing a number of reduced sulphur compounds including elemental sulphur, sulphide, sulphite, thiosulphate and tetrathionate into sulphate (Roy and Trudinger, 1970). The mechanisms involved in the oxidation of inorganic sulphur compounds have been reviewed by Peck (1968), Trudinger (1967, 1969), Roy and Trudinger (1970) and Aleem (1975).

Under natural conditions the usual substrate for their growth is either elemental sulphur or thiosulphate, the latter being preferred. Two pathways have been proposed for the oxidation of thiosulphate to sulphate in thiobacilli. Thiosulphate is either oxidized to tetrathionate which is converted, *via* polythionate, to sulphate (Nathansohn, 1902; Vishniac, 1952; London and Rittenberg, 1964; Pankhurst, 1964) or as proposed by Peck (1962, 1968), it is first cleaved into two one-sulphur moieties which are then oxidized to sulphate. Cell-free extracts of T. thiooxidans (Tano et al., 1968) and T. concretivorus (Moriarty, 1969) were reported to oxidize thiosulphate to trithionate and tetrathionate. However, in T. denitrificans, thiosulphate was metabolized to sulphide, polysulphide, tetrathionate and sulphite. In the presence of GSH, thiosulphate was mainly oxidized to sulphite. It was also shown that on replacing GSH with nitrate, sulphide and sulphite were the initial products of thiosulphate oxidation and these were converted into tetrathionate and sulphate (Aminuddin, 1974). Peeters and Aleem (1970) reported that thiosulphate oxidation in T. denitrificans was linked to nitrate reduction. They showed that under anaerobic conditions, cytochromes b, c, a and o, reduced by thiosulphate were reoxidized by nitrate.

The oxidation of sulphide to sulphate, which involves a release of eight electrons, is linked to phosphorylation through the electron transport chain. In *T. thioparus* (Vishniac and Santer, 1957; London and Rittenberg, 1964) and *T. thiooxidans* (London and Rittenberg, 1964), apart from sulphate, thiosulphate and polythionates were identified as the other products of sulphide oxidation. It is proposed that the initial product during the oxidation of sulphide is a polysulphide (Trudinger, 1967, 1969; Aminuddin and Nicholas, 1973; Saxena and Aleem, 1973). Aminuddin and Nicholas (1973) also showed that under anaerobic conditions, in the presence of nitrate, sulphide was oxidized to sulphate with concomitant reduction of nitrate to nitrite. However, in the presence of nitrite, the final product was only polysulphide.

Sulphite, the other product of thiosulphate cleavage, is oxidized via either an AMP-dependent APS reductase (Peck, 1961; Peck et al., 1965; Bowen et al., 1966; Hemfling et al., 1967; Lyric and Suzuki, 1970B; Aminuddin and Nicholas, 1974A) or by an AMP-independent sulphite oxidase (Adair, 1966; Charles and Suzuki, 1966; Kodama and Mori, 1968; Moriarty and Nicholas, 1969; Lyric and Suzuki, 1970A). APS reductase catalyses the production of adenosine-5'-sulphatophosphate (APS) from sulphite and AMP. APS produced in this reaction subsequently generates ATP through the combined action of ADP-sulphurylase and adenylate kinase (Peck, 1960, 1962). In T. denitrificans, the purified enzyme utilized ferricyanide or 0, as an electron acceptor (Bowen et al., 1965). The other sulphite oxidizing enzyme, viz. sulphite oxidase does not require In T. novellus (Charles and Suzuki, 1966) AMP for its activity. and in T. thioparus (Lyric and Suzuki, 1970A), sulphite oxidase, which functioned with either ferricyanide or cytochrome c as an electron acceptor, is a soluble enzyme. However, sulphite oxidase from T. thiooxidans (Adair, 1966; Kodama and Mori, 1968), T. concretivorus (Moriarty, 1969) and T. denitrificans (Aminuddin and Nicholas, 1974A) was shown to be associated with cell-membranes. This particulate sulphite oxidase from T. denitrificans used either nitrate, 0, or ferricyanide as an electron acceptor but the purified enzyme functioned only with ferricyanide (Aminuddin and Nicholas, 1974A).

Thiobacilli are mainly strict aerobic bacteria requiring  $O_2$  as a terminal oxidant for cell respiration. The notable exception is *T. denitrificans* which can respire even in the absence of air by using nitrate as an alternate terminal electron acceptor. Several other

obligate or facultative anaerobes are known to thrive in the absence of  $O_2$  by coupling their respiration to the reduction of inorganic compounds. The various compounds that can substitute for  $O_2$  are: sulphate, as in *Desulphovibrio* and *Desulphotomaculum* (Postgate, 1965), nitrate as in the denitrifying bacteria (Payne, 1973) and  $CO_2$  as in *Methanobacterium* (Stadtman, 1967). The mean energy yields for the transfer of a molar equivalent of electrons from organic compounds to these terminal electron acceptors (Payne, 1973), are presented in Table 1.

It is evident that the energy yield with nitrate as the oxidant is significantly higher as compared to the other inorganic compounds and its evolutionary advantage is reflected by the relatively larger number of nitrate reducers than those which utilize either sulphate or  $CO_2$  as the terminal oxidant (Payne, 1973).

Among thiobacilli, only *T. denitrificans* can grow under anaerobic conditions with nitrate as the alternate terminal electron acceptor. Since the products of this nitrate-dependent respiration are nitrogenous gases, it has been classified as a denitrifying bacterium (Beijernick, 1904; Baalsrud and Baalsrud, 1954; Ishaque and Aleem, 1973). Although the denitrification process results in a net loss of nitrogen from the soil, it plays an important role in the nitrogen cycle in nature by maintaining the dinitrogen level in the biosphere at a steady level (Alexander, 1971).

# 1.2 Denitrification

Nitrate is utilized by microorganisms in two distinct ways: (a) assimilation, whereby nitrate is reduced to ammonia and incorporated

Table 1. Free-energy production with various compounds as terminal electron acceptors in bacteria (adapted from Payne, 1973).

Terminal electron acceptor	∆F (kCal/mole)	Reference
0 <sub>2</sub>	-26.5	Payne (1970)
NO <sub>3</sub>	-18.0	McCarty (1972)
$so_4^{2-}$	-3.4	McCarty (1972)
co <sub>2</sub>	-2.4	McCarty (1972)

into cellular nitrogen compounds and (b) dissimilation, whereby nitrate serves as an alternative electron acceptor to O<sub>2</sub> and is reduced to nitrite. The term denitrification is used when nitrogenous gases are produced. Nitrate may be dissimilated either to nitrite only, as in Escherichia coli (Taniguchi et al., 1956) and Hemophilus parainfluenzae (White, 1962) or reduced further to N<sub>2</sub> gas through a series of reactions as in T. denitrificans (Baalsrud and Baalsrud, 1954; Ishaque and Aleem, 1973). The microbial, physiological and biochemical aspects of nitrate respiration have been reviewed by Najjar and Chung (1956), Kluyver (1959), Nason (1962), Payne (1973), Gracia (1975), Delwiche and Bryan (1976), Payne (1976), Stouthamer (1976), Haddock and Jones (1977) and Thauer et al. (1977).

The reduction of nitrate to  $N_2$  gas during denitrification is associated with the following changes in the free energy (Thauer *et al.*, 1977).

$$NO_{3}^{-} + H_{2} \longrightarrow NO_{2}^{-} + H_{2}O \qquad -39 \text{Kcal/mol} \qquad (1)$$

$$2NO_{2}^{-} + 3[H_{2}] + 2H^{+} \longrightarrow N_{2}^{-} + 4H_{2}O \qquad -190 \text{Kcal/mol} \qquad (2)$$

The energy released during the reduction of nitrate to nitrite and nitrite to dinitrogen, is utilized for the production of ATP (Thauer *et al.*, 1977).

It is established that the reduction of nitrate to nitrite is mediated by nitrate reductase. The further reduction of nitrite to nitrogen gas proceeds in a sequence of reactions (3-5) with nitric oxide and nitrous oxide as the intermediates (Payne, 1973).

$$2NO_{2}^{-} + H_{2}^{-} + 2[H^{+}] \longrightarrow 2NO + 2H_{2}O -35.1Kcal/mol (3)$$

$$2NO + H_{2} \longrightarrow N_{2}O + H_{2}O -73.2Kcal/mol (4)$$

$$N_{2}O + H_{2} \longrightarrow N_{2}^{-} + H_{2}O -81.6Kcal/mol (5)$$

#### (Thauer et al., 1977).

Only a limited number of denitrifying bacteria reduce nitrate through to dinitrogen. Alcaligenes odorans reduces nitrite but not nitrate to N<sub>2</sub> gas (Chatelain, 1969) while in *Pseudomonas fluorescens* (Greenberg and Becker, 1972) and *Corynebacterium nephridii* (Hart et al., 1965; Renner and Becker, 1970) the final product of the denitrification process is nitrous oxide.

Most of the denitrifiers are non-fermentative bacteria which grow anaerobically only when supplied with nitrate as an alternative electron acceptor to  $O_2$ . The only obligate anaerobic bacterium which exhibits denitrification is *Propionibacterium pentosaceum* (Gent-Ruijters *et al.*, 1975).

# 1.2.1 Reduction of nitrate to nitrite

Nitrate reductase is the first enzyme involved in both assimilatory and dissimilatory nitrate reduction. The assimilatory nitrate reductase is usually a substrate inducible enzyme and is generally located in the soluble fraction of cell (Payne, 1973). The synthesis and activity of this enzyme is repressed by ammonia. Its activity is also inhibited by chlorate and it has been designated a B-type nitrate reductase by Pichinoty (1970). The reduction of nitrate to ammonia, which involves 8-electron transfer, in the assimilatory

pathway is not coupled to phosphorylation (Hadjipetrou and Stouthamer, 1965).

In contrast, the dissimilatory nitrate reductase functions as a respiratory enzyme in bacteria growing under anaerobic conditions. The synthesis and the functioning of this enzyme is regulated by the partial pressure of O<sub>2</sub> and the concentration of nitrate in the environment (Pichinoty and D'Ornano, 1961; Van't Riet *et al.*, 1968). Reduced nitrogenous compounds like ammonia and amino acids do not affect either the enzyme activity or its synthesis. The enzyme reduces chlorate, bromate but not iodate and has been classified by Pichinoty (1964) as type A. Dissimilatory nitrate reductases are generally associated with cell membranes, with the exception of those in *Spirillum itersonii* (Gauthier *et al.*, 1970) and *Clostridium perfringens* (Chiba and Ishimoto, 1973).

Some properties of dissimilatory nitrate reductases from various bacteria are summarized in Table 2. Dissimilatory reduction of nitrate to nitrite is linked to a particulate electron-transport chain consisting of dehydrogenases and electron carriers with nitrate reductase as the terminal enzyme. In many bacteria, NADH serves as an electron donor for nitrate reductase, e.g. *E. coli* (Medina and Heredia, 1958; Iida and Taniguchi, 1959; Itagaki *et al.*, 1962; Cole and Wimpenny, 1968), *Pseudomonas aeruginosa* (Fewson and Nicholas, 1961B), *Pseudomonas denitrificans* (Radcliffe and Nicholas, 1970), *Bacillus stearothermophilus* (Downey, 1966; Kiszkiss and Downey, 1972A), *Aerobacter aerogenes* (Knook *et al.*, 1973), *Proteus mirabilis* (DeGroot and Stouthamer, 1970) and *Micrococcus denitrificans* (Lam and Nicholas, 1969B). However, depending upon the type of organism and also on growth conditions, the reduction of nitrate may also be linked to various

Table 2.

reductase in various organisms (Taken from Thauer et al., 1977).

Organism	Electron donor	Electron carrier	Reference
(A) OBLIGATE ANAEROBES			24
Clostridium perfringens $(NO_3^- \rightarrow NO_2^-)$	Pyruvate	Ferredoxin	Chiba and Ishimoto (1973, 1977)
Propionibacterium pentosaceum $(NO_3 \rightarrow N_2)$	NADH, lactate, glycerol- phosphate	Cytochrome b	Schwartz (1973), Sone (1974), Gent-Ruijters <i>et al</i> . (1975)
Staphylococcus aureus $(NO_3 \longrightarrow NO_2)$	Lactate	MK, cytochrome b	Chang and Lascelles (1963), Sasarman <i>et al.</i> (1974), Burke and Lascelles (1975)
(B) FACULTATIVE ANAEROBES			
Aerobacter aerogenes $(NO_3^- \rightarrow NO_2^-)$	NADH	UQ, cytochrome b	Forget and Pichinoty (1964), Knook and Planta (1971B), Van't Riet and Planta (1969, 1975)
Bacillus stearothermophilus ( $NO_3^- \rightarrow NO_2^-$ )	NADH, succinate	MK, cytochrome b	Downey (1962), Schulp and Stouthamer (1970), Kiszkiss and Downey (1972A, B)
Escherichia coli $(NO_3^- \rightarrow NO_2^-)$	NADH, lactate, formate, glycerol- phosphate	UQ, cytochrome b	Itagaki <i>et al.</i> (1962), Forget (1974) MacGregor <i>et al.</i> (1974), Haddock and Kendall-Tobias (1975), Clegg (1976)
$\begin{array}{ccc} \text{Micrococcus denitrificans} \\ (\text{NO}_3^- \longrightarrow \text{ N}_2) \end{array}$	NADH, succinate	UQ, cytochrome b	Scholes and Smith (1968), Lam and Nicholas (1969A, B, C), Forget (1971), Sapshead and Wimpenny (1972)
Psuedomonas aeruginosa $(NO_3 \rightarrow N_2)$	NADH	Cytochrome b and c	Fewson and Nicholas (1961)
Pseudomonas denitrificans $(NO_3 \longrightarrow N_2)$	Formate, NADH	Cytochrome b	Radcliffe and Nicholas (1970)
Proteus mirabilis $(NO_3^- \rightarrow NO_2^-)$	NADH, formate	UQ cytochrome b	DeGroot and Stouthamer (1970)
Spirillum itersonii ( $NO_3^- \rightarrow NO_2^-$ )	8	Cytochrome b	Gauthier et al. (1970)
$\begin{array}{ccc} \mbox{\it Phiobacillus denitrificans} \\ (\mbox{NO}_3^- \longrightarrow \ \mbox{N}_2) \end{array}$	Sulphite, sulphide, NADH	Cytochrome <i>b</i> and <i>c</i>	Baalsrud and Baalsrud (1954), Peeters and Aleem (1970), Adams <i>et al</i> . (1971), Aminuddin and Nicholas (1973, 1974B)

Abbreviations : MK, Menaquinone-6 (vitamin K<sub>2</sub>); UQ, ubiquinone.

organic acids. Thus, in E. coli (Taniguchi and Itagaki, 1960), P. mirabilis (DeGroot and Stouthamer, 1970) and P. denitrificans (Radcliffe and Nicholas, 1970) formate is an effective electron donor (Table 2). The enzyme from T. denitrificans is unique since its activity is coupled to the oxidation of inorganic sulphur compounds (Aubert et al., 1959; Baalsrud and Baalsrud, 1954; Peeters and Aleem, 1970; Adams et al., 1971; Aminuddin and Nicholas, 1974B). Several workers have found that the purified preparations of dissimilatory nitrate reductases from various bacteria do not utilize either NADH or formate but either reduced flavins or dyes are effective electron donors for nitrate reduction (Fewson and Nicholas, 1961B; Lam and Nicholas, 1969B; Radcliffe and Nicholas, 1970; Kiszkiss and Downey, 1972A; Rosso et al., 1973).

In several of the early studies, FAD and FMN were shown to be electron carriers for nitrate reduction in bacteria (Nason, 1962). Recently, however, ubiquinones have also been implicated in nitrate respiration in gram-negative bacteria (Table 2). Extraction of ubiquinones from the membrane fractions of *E. coli* (Itagaki *et al.*, 1962; Itagaki, 1964; Enoch and Lester, 1974) and *A. aerogenes* (Knook and Planta, 1971A, B), resulted in a loss of nitrate reductase activity with the physiological electron donors. In gram-positive bacteria, however, naphthoquinones are probably involved in the functioning of dissimilatory nitrate reductases (Downey, 1962). A requirement of quinone for nitrate respiration was also demonstrated in the menaquinonedeficient mutants of *Staphylococcus aureus* (Sasarman *et al.*, 1974).

All the nitrate respiring bacteria studied so far contain at least one *b*-type cytochrome which participates in nitrate reduction (Najjar and Chung, 1956; Itagaki *et al.*, 1961; Nason, 1963; Downey, 1966; Lam and Nicholas, 1969B; Ruiz-Herrera and DeMoss, 1969;

DeGroot and Stouthamer, 1970; Radcliffe and Nicholas, 1970; Van't Riet *et al.*, 1972; Knook *et al.*, 1973; Stouthamer, 1976). Under anaerobic conditions, the electrons from the respiratory chain are diverted to nitrate at the level of cytochrome *b*. The enhanced production of cytochrome *c* in the anaerobic cultures of nitrate grown *E. coli* and *A. aerogenes* indicated its role in nitrate respiration (Gray *et al.*, 1963; Fujita and Sato, 1966). Further evidence for the participation of cytochrome *c* in nitrate reduction has been obtained in the cell-free preparations of *E. coli* (Gray *et al.*, 1963), *M. denitrificans* (Fewson and Nicholas, 1961A), *P. aeruginosa* (Fewson and Nicholas, 1961B), and *T. denitrificans* (Aubert *et al.*, 1959; Peeters and Aleem, 1970; Aminuddin and Nicholas, 1974B) where nitrate was shown to re-oxidize reduced cytochrome *c*. It has been suggested that in some bacteria different forms of *c*-type cytochrome mediate the transfer of electrons to nitrate and O<sub>2</sub> (Payne, 1973).

Besides the above-mentioned components of the electron transport chain, molybdenum and iron have also been shown to be the constituents of purified nitrate reductase from E. coli (Taniguchi and Itagaki, 1960; Forget, 1974; MacGregor et al., 1974), M. denitrificans (Forget, 1971) and M. halodenitrificans (Rosso et al., 1973). Lam and Nicholas (1969B), using radioactive molybdate had shown earlier that molybdenum was a component of 118-fold purified nitrate reductase from M. denitrificans. Labile sulphide has been found in purified enzyme preparations from E. coli (Forget, 1974), M. halodenitrificans (Rosso et al., 1973) and M. denitrificans (Forget, 1971). Forget and Dervartanian (1972) observed that electron spin resonance (ESR) signals due to Mo(V) and Fe(III) were eliminated on reduction of the enzyme and this was accompanied by the appearance of the signals due to Fe(II)-S. Addition of nitrate to the reduced enzyme restored the ESR signals attributed to Mo(V) and Fe(III). These

investigations indicated that molybdenum and iron function as electron carriers in the reduction of nitrate by nitrate reductase.

# 1.2.2 Regulation of nitrate reductase

Both the activity and synthesis of dissimilatory nitrate reductase is controlled by nitrate and the concentration of 0, in the Although the enzyme is generally nitrate-inducible, growth medium. anoxia is also important for the onset of the synthesis of the enzyme as reported for A. aerogenes (Pichinoty and D'Ornano, 1961), E. coli (Showe and DeMoss, 1968), P. mirabilis (DeGroot and Stouthamer, 1970), and Pseudomonas perfectomarinus (Payne et al., 1971). In accordance with its respiratory function under anaerobic conditions, the reduction of nitrate by dissimilatory nitrate reductase is coupled to the electron transport-dependent oxidative phosphorylation of the cell. A free energy change of -38.5 Kcal per pair of electrons has been calculated for the reduction of nitrate to nitrite (Decker et al., 1970). Thus due to a relatively high redox-potential of  $NO_3^2/NO_2^2$  couple (E' = +421 mV), the reduction of nitrate with most of the reductants, except succinate, is sufficiently exergonic to allow for the formation of 2 moles of ATP per mole of nitrate reduced (Kroger, 1977). ATP formed in this dissimilatory pathway is utilized for biosynthetic reactions of the cell. Since the principal function of the respiratory process is to furnish metabolic energy, it is conceivable that the cellular respiration, whether linked to 0, or nitrate, be regulated by the energy status of Thus in Mycobacterium tuberculosis, NADH-oxidase which the organism. is linked to oxidative phosphorylation, was stimulated by AMP and this effect was reversed by NAD<sup>+</sup> (Worcel et al., 1965). In Mycobacterium phlei, the activity of the respiratory NADH-oxidase was controlled by the levels of NAD<sup>+</sup>, AMP and inorganic phosphate (Bogin et al., 1969).
NADH-dehydrogenase from *E. coli* was competitively inhibited by NAD<sup>+</sup> and the extent of inhibition was dependent on the NAD<sup>+</sup>/NADH ratio (Dancey and Shapiro, 1976). These investigations indicate that in bacteria the aerobic respiration may be controlled through the energy charge and redox state of the cell. However, no information is available about the factors which might control the rate of nitrate respiration.

In all the chemolithotrophic bacteria, CO2 fixation supplies their carbon requirements and ATP needed for the assimilation of CO2 is generated during the oxidation of inorganic substrates via the electron transport chain. It has been suggested that the rate of energy generation through the electron transport is controlled by  $\rm CO_2$  and its reductive assimilation (Aleem, 1977). The mechanism for the generation of NADH, the reductant essential for CO2 fixation in chemolithotrophic bacteria, was elucidated by the pioneering work of Aleem et al. (1963). These workers demonstrated that in Nitrobacter, Nitrosomonas and Ferrobacillus, NADH was produced via an ATP-driven reverse flow of electrons through the respiratory chain. Subsequent work with T. novellus (Aleem, 1966A, C), Nitrosomonas europaea (Aleem, 1966B), Nitrobacter agilis (Sewell and Aleem, 1969), T. denitrificans (Peeters and Aleem, 1970) and T. neapolitanus (Aleem, 1969; Saxena and Aleem, 1972; Roth et al., 1973) has established that this phenomenon is widespread in the chemolithotrophic bacteria. Although the reverse-electron flow for the generation of NADH operates in lithotrophic bacteria, according to Kelly (1971) it is not reasonable to assume that the oxidation of NADH by the normal mode of electron flow does not occur In contrast, Smith et al. (1967) have proposed in obligate lithotrophs. that failure to oxidize NADH was the prime cause of obligate lithotrophy. However, several workers have noted the presence of NADH-oxidase activity in various obligate chemolithotrophs including *T. neopolitanus* (Trudinger and Kelly, 1968), *T. thiooxidans* (Tano and Imai, 1968) and *Nitrobacter winogradskyi* (Gool and Laudelout, 1967).

To summarise, in chemolithotrophic bacteria, NADH is largely produced *via* an ATP-driven reverse electron flow and NADH can also be oxidized by the same electron transport chain. To enable NADH to be conserved for biosynthetic reactions, it is essential that some mechanism should operate to regulate its oxidation *via* the electron transport chain.

#### 1.2.3 Dissimilatory nitrite reductase

The synthesis of dissimilatory nitrite reductase is also controlled by the concentrations of  $O_{2}$  and nitrate in the growth medium (Delwiche and Bryan, 1976). No exceptional substrates have been reported for the dissimilation of nitrite. Several oxidizable substrates are utilized in culture but in vitro this enzyme is active with either NADH or reduced flavins (Najjar and Chung, 1956; Fewson and Nicholas, 1961A; Walker and Nicholas, 1961; Nason, 1962; Lam and Nicholas, 1969C). Nitrite reductase from P. aeruginosa, reduced nitrite with reduced forms of either flavin, pyocyanine, methylene blue or 1,4-naphthoquinone but not with NAD(P)H or reduced cytochrome c (Walker and Nicholas, 1961). Nitrite reductase from P. perfectomarinus utilized either NADH or NADPH for its activity (Payne, 1973). The enzyme from T. denitrificans differs from the other dissimilatory nitrite reductases because it functions with sulphide as the reductant (Aminuddin and Nicholas, 1974B).

Nitrite reductase from denitrifying bacteria such as *P. aeruginosa* (Horio et al., 1961A; Walker and Nicholas, 1961; Kuronen and Ellfolk, 1972;

Gudat et al., 1973), Alcaligenes faecalis (Iwasaki and Matsubara, 1971) and M. denitrificans (Lam and Nicholas, 1969C; Newton, 1969) have been purified and their properties examined. The enzyme from these sources contained two different haem compounds, namely cytochromes c and d.

A crystalline preparation of nitrite reductase from P. aeruginosa, which was previously referred to as cytochrome oxidase (Yamanaka and Okunuki, 1974), has been studied extensively. This enzyme, besides accepting electrons from several reductants, functioned with either reduced cytochrome  $c_{551}$  or the reduced copper-containing protein, azurin, as a donor. Both the reductants were isolated from P. aeruginosa (Horio, 1958; Horio et al., 1958, 1961A). Nitrite reductase from M. denitrificans was also shown to function with reduced cytochrome  $c_{552}$  which was isolated from the same organism (Newton, 1969). As in P. aeruginosa, the purified nitrite reductase from M. denitrificans (Lam and Nicholas, 1969C; Newton, 1969) and A. faecalis (Iwasaki and Matsubara, 1971) also exhibited cytochrome oxidase activity. Both the nitrite reductase and cytochrome oxidase activities of the purified enzyme preparations from these organisms were inhibited by cyanide but CO affected only the reaction with O2. The molecular weight of nitrite reductase from P. aeruginosa (Gudat et al., 1973) and M. denitrificans (Newton, 1969) was 120,000. Nagata et al. (1970) calculated the molecular weight of purified nitrite reductase from P. aeruginosa to be 65,000 on the basis of its amino acid composition. The molecular weight of 90,000 was reported for the crystalline enzyme from A. faecalis (Iwasaki and Matsubara, 1971).

Investigations by Radcliffe and Nicholas (1968) and Matsubara

and Mori (1968) indicate that nitrite reductase from *P. denitrificans*, which had a molecular weight of 70,000, was distinctly different in structural properties from the enzyme from *P. aeruginosa*. Thus, unlike the enzyme from the latter organism, which has cytochromes *c* and *d* as the prosthetic groups, nitrite reductase from *P. denitrificans* contained cytochrome  $c_{552}$  or  $c_{553}$  and copper as its prosthetic groups. The enzyme from *Achromobacter cycloclastes*, with a molecular weight of 65,000 was, however, reported to be a copper protein and did not contain either flavins or cytochromes (Iwasaki and Matsubara, 1972). There is no evidence for siroheme as a prosthetic group of dissimilatory nitrite reductases as is usually the case with the assimilatory nitrite reductases (Murphy *et al.*, 1974; Vega *et al.*, 1975).

With most of the purified dissimilatory nitrite reductases studied thus far, nitric oxide has been identified as the product of the reaction (Renner and Becker, 1970; Barbaree and Payne, 1967).

#### 1.2.4 Nitric oxide reductase and nitrous oxide reductase

The further reduction of nitric oxide to nitrous oxide is catalyzed by nitric oxide reductase (Payne *et al.*, 1971; Matsubara and Iwasaki, 1971; Miyata, 1971; Cox and Payne, 1973). Only a limited information is available about the properties of this enzyme. In *P. perfectomarinus* (Cox and Payne, 1973) nitric oxide reductase was reported to be a soluble enzyme whereas in *A. faecalis* (Matsubara and Iwasaki, 1971) and *P. denitrificans* (Miyata *et al.*, 1969), the enzyme was shown to be membrane bound. The purified nitrite reductase from *A. faecalis* exhibited nitric oxide reductase activity as well. However its suggested physiological role was for the reduction of nitrite because of the presence of a separate particulate nitric oxide reductase which had much lower K<sub>m</sub> for nitric oxide (Matsubara and Iwasaki, 1972).

Nitrous oxide, the product of nitric oxide reductase, is finally reduced to dinitrogen by nitrous oxide reductase (Matsubara and Iwasaki, 1971; Cox and Payne, 1973; Matsubara, 1970, 1971, 1975). This particulate enzyme is very labile and has not been characterized. In vitro, the enzyme functions with reduced form of either viologen dyes or dichlorophenol indophenol but the physiological reductant has not A possible role of transition metals in this step of been identified. denitrification has been indicated by the inhibition of nitrous oxide reductase by carbon monoxide (Matsubara, 1971), cyanide and azide (Matsubara, 1975). In addition, Matsubara (1971, 1975) have suggested the involvement of cytochromes b and c in the reduction of nitrous oxide. Payne et al. (1971) have also observed that in P. perfectomarinus, a c-type cytochrome was tightly associated with membrane fractions containing nitric oxide reductase.

#### 1.3 Aim of thesis

The objective of these investigations was to examine some biochemical aspects of denitrification in the chemolithotrophic bacterium, *Thiobacillus denitrificans*. Work on the following lines was carried out:-

- To compare the properties and determine the relationship between the sulphite- and NADH-linked nitrate reductase activities.
- (2) To explore the mechanism by which the utilization of NADH for nitrate reductase is regulated.
- (3) To identify the components of electron-transport chain involved in the reduction of nitrate, nitrite and  $O_2$ .

(4) To study the properties of the sulphide-linked nitrite reductase.

#### 2. MATERIALS AND METHODS

#### 2.1 Preparation of solutions and buffers

Double glass-distilled water was used to dispense all the reagents used in this work. Sodium sulphite  $(Na_2SO_3.9H_2O)$ , sodium sulphide  $(Na_2S.9H_2O)$  and NADH were dispensed in 0.05M Tris-HCl buffer (pH 7.5) just prior to use. Crystals of sodium sulphide were washed with distilled water and carefully dried several times with filter paper before weighing.

The stock solutions of 0.1M of Tris-HCl, potassium phosphate and Tris-maleate buffers were prepared by the method of Gomori (1955) and stored at  $2^{\circ}$ . The pH of all the buffers were adjusted at room temperature (20-25°) using a Beckman H5 pH meter which was standardised frequently. The pH of Tris-HCl buffer was adjusted at the temperature at which it was to be used subsequently (Sigma Technical Bulletin 106B, 1967).

#### 2.2 Biological materials

#### 2.2.1 Bacterium

Thiobacillus denitrificans (9547, NClB strain AB5) was purchased from the National Collection of Industrial Bacteria, Torrey Research Station, Aberdeen, Scotland.

#### 2.2.2 Enzymes

Alcohol dehydrogenase, snake venom pyrophosphatase and DNase were supplied by Sigma Chemical Co. (St. Louis, Missouri, U.S.A.) while inorganic pyrophosphatase was from Calbiochem (Los Angeles, California, U.S.A.).

#### 2.3 Chemicals and other materials

#### 2.3.1 Nucleotides and marker proteins

ATP, ADP, AMP, NADH, NADPH, NAD<sup>+</sup>, NADP<sup>+</sup>, FMN and FAD were purchased from Sigma Chemical Co. (St. Louis, Missouri, U.S.A.) as were the following marker proteins: cytochrome c (horse heart, type VI), bovine serum albumin (fraction V), ovalbumin (grade V),  $\gamma$ -globulin and myoglobin.

#### 2.3.2 Other chemicals

Rotenone, amytal, 2-heptyl-4-hydroxyquinoline-N-oxide (HQNO), antimycin A, p-chloromercuribenzoic acid (p-CMB), phenazine methosulphate (PMS), glutathione (reduced) and Tris[hydroxymethyl]aminomethane (Tris) were purchased from Sigma Chemical Co. (St. Louis, Missouri, U.S.A.); sodium diethyldithiocarbamate (DIECA) was from Merck (Darmstadt, Germany). Bathocuproin and blue dextran were from Fluka (Buchs, Switzerland) and Pharmacia Fine Chemicals (Uppsala, Sweden) respectively. Atabrine was from Imperial Chemical Industries (Marlston, South Australia). <sup>15</sup>Nnitrite (27 atom % excess) was obtained from Office Industrial De L'Azote, France.

All the other chemicals were of the highest purity available.

#### 2.3.3 Chromatographic materials

Whatman DEAE-cellulose (type 32) and Whatman 3MM paper were supplied by H. Reeve Angel, London EC4 and Sephadex G-25, G-100, G-150 and Sepharose 6B were from Pharmacia Fine Chemicals (Uppsala, Sweden).

#### 2.4 Culture of the bacterium

#### 2.4.1 Growth of the organism

Thiobacillus denitrificans was grown anaerobically in a medium described by Adams *et al.* (1971). The composition of the medium used is given in Table 3. Batch cultures of the organism were grown at  $28^{\circ}$  in 55l carboys. The pH of the culture medium during growth was maintained at 7.0 by an automatic titration with sterile 20% (w/v)  $K_2CO_3$  by means of a pH stat unit (Radiometer, Titrator 11 pH meter 8, Copenhagen, Denmark). Optimum growth was obtained within 48-60 hr.

#### 2.4.2 Maintenance of the culture

T. denitrificans, maintained in a liquid culture described in Table 3, was sub-cultured three times a week. Microscopic examination and plating of the culture were carried out periodically to check that there were no contaminants present.

#### 2.4.3 Harvesting

The bacteria were harvested in a Sorvall  $\cdot$ RC-2 refrigerated centrifuge fitted with a continuous flow rotor (SS-34). The cells were washed twice with cold 0.05M Tris-HCl buffer (pH 7.5) and when not used immediately, were stored at -15<sup>°</sup>.

#### 2.5 Enzyme techniques

#### 2.5.1 Preparation of acetone-dried cells

The washed cells, suspended in 0.025M Tris-HCl buffer (pH 7.5) (500 mg wet weight/ml), were added slowly to ten volumes of cold acetone at  $-15^{\circ}$  with continuous stirring. After 10 min, the mixture was quickly filtered through a Buchner funnel, fitted with Whatman No. 41

#### Table 3. Composition of the culture solution for T. denitrificans.

The solution of  $\text{KH}_2\text{PO}_4$  (200 gm/l) was adjusted to pH 6.5 with KOH, autoclaved at 17 lb/in<sup>2</sup> for 30 min, cooled and then added at the rate of 10 ml/l of the culture medium. The following salts:  $\text{Na}_2\text{S}_2\text{O}_3.5\text{H}_2\text{O}$ ,  $\text{KNO}_3$ ,  $\text{MgSO}_4.7\text{H}_2\text{O}$ ,  $\text{CaCl}_2.2\text{H}_2\text{O}$  and  $\text{NH}_4\text{Cl}$  were dissolved and autoclaved together; FeSO<sub>4</sub> and KHCO<sub>3</sub> were sterilized separately. Appropriate amounts of the stock solutions of these three fractions were combined after cooling and made up to the required volume with sterile double-distilled water just before inoculation.

100 ml of fresh cell-suspension (0.5%, w/v) was used for inoculating each litre of the medium.

	Chemical	<ul><li>Concentration</li><li>(g/l of culture solution)</li></ul>
	Na2 <sup>S</sup> 2 <sup>O</sup> 3.5H2 <sup>O</sup>	5.0
	КОН	0.2
	KHCO3	1.0
	KNO3	2.0
	NH4C1	0.5
6	MgS0 <sub>4</sub> .7H <sub>2</sub> 0	0.5
	FeSO4	0.01
	CaCl <sub>2</sub> .2H <sub>2</sub> O	0.1
	KH <sub>2</sub> PO <sub>4</sub>	2.0

filter paper. The temperature during filtration was maintained below  $0^{\circ}$  by packing an ice-salt mixture around the Büchner funnel. The cells, after washing twice with three volumes of cold acetone, were dried under a stream of N<sub>2</sub> gas at  $4^{\circ}$ . They were then dried completely under vacuum in a dessicator over 36N H<sub>2</sub>SO<sub>4</sub> and stored under vacuum at  $4^{\circ}$  over NaOH pellets.

#### 2.5.2 Preparation of homogenates and cell-free extracts

All the operations described in this Section were carried out at 4°.

#### 2.5.2.1 Cell-free extracts of fresh cells

The washed cells, suspended (25%, w/v) in 0.05M Tris-HCl buffer (pH 7.5) were disrupted by passing them through a chilled Aminco French pressure cell (American Instrument Co., Maryland, U.S.A.) twice at 20,000 p.s.i. (Hughes *et al.*, 1971). The homogenate was then centrifuged either at 10,000x g for 30 min or at 18,000x g for 45 min in Sorvall-SS3 centrifuge (SS-34 rotor) to sediment the intact cells. The supernatant fraction (S10 or S18 respectively), referred to as the crude extract was used either as such or centrifuged further as described in Section 2.5.2.3.

#### 2.5.2.2 Cell-free extracts of acetone-dried cells

The acetone-dried cells were ground to a fine powder in a cold pestle and mortar and then suspended (10%, w/v) in 0.025M Tris-HCl buffer (pH 7.5). After adding DNase (0.1 mg/10 ml cell suspension), the preparation was stirred with a magnetic stirrer for 3-4 hr at  $4^{\circ}$  in an ice bath. It was then filtered through two layers of cheese-cloth and the filtrate was passed through a chilled Aminco French pressure cell twice at 20,000 p.s.i. This homogenate was centrifuged at 18,000x g for 45 min as described in Section 2.5.2.1. The resultant supernatant fraction (S18), which is referred to as the crude extract, was the source of the sulphide-linked nitrite reductase.

#### 2.5.2.3 Preparation of S100 and P100 fractions

The crude extracts (S10 and S18 of the fresh cells or acetonedried cells, respectively) were centrifuged at 100,000x g for 90-120 min in a Spinco Ultracentrifuge, Model L (rotor type 30) to obtain a supernatant (S100) and a pellet or membrane fraction (P100). The pellet fraction (P100) was suspended in 0.05M Tris-HCl buffer (pH 7.5) and dispersed uniformly by grinding in a glass homogenizer (Kontas Glass Co., New Jersey, U.S.A.).

#### 2.5.3 Solubilization of nitrate reductase

The following procedure was employed to dissociate nitrate reductase from the cell membranes:-

The pellet fraction (Pl00), suspended in 0.05M Tris-HCl buffer (pH 7.5), was diluted with the same buffer to a final protein concentration of 30 mg/ml. It was subjected to an ultrasonic treatment (MSE Ultrasonicator; 60W output) for 30 sec and then incubated with sodium deoxycholate (0.06 mg/mg protein) at  $30^{\circ}$  in a reciprocating waterbath (100 oscillations/min). After 30 min, this preparation was centrifuged at 100,000x <u>g</u> for 60 min and the supernatant fraction thus obtained, was passed through a Sephadex G-25 column (1.25 x 15 cm) which had been equilibrated with 0.05M Tris-HCl buffer (pH 7.5). The resulting preparation is referred to as the solubilized enzyme.

#### 2.5.4 · Enzyme assays

#### 2.5.4.1 Nitrate reductase

Nitrate reductase activity with either sulphite or NADH as the electron donor, was determined by measuring the rate of nitrite production.

The enzyme activity, with sulphite as a donor, was assayed in open test tubes (1.3 x 10 cm). The reaction mixture contained: Tris-HCl buffer (pH 7.5), 90 µmoles;  $KNO_3$ , 1 µmole;  $Na_2SO_3$ , 1 µmole and 0.05 ml of enzyme in a total volume of 2 ml. Tubes which contained either the boiled enzyme or those without sulphite served as the control. After equilibrating the tubes in a reciprocating water bath (100 oscillations/min) at 30<sup>o</sup> for 5 min, the reaction was started by adding a freshly prepared solution of  $Na_2SO_3$ . After 15 min incubation, nitrite produced was determined as described in Section 2.7.2.

The reaction with NADH as the electron donor was also conducted in open test tubes (1.3 x 10 cm). The reaction mixture contained the following: Tris-HCl buffer (pH 7.5), 80 µmoles;  $KNO_3$ , 1 µmole; NADH, 1 µmole and 0.05 ml enzyme preparation. Control tubes contained the boiled enzyme preparation. The reaction was initiated by adding NADH and after incubating for 15 min at  $30^{\circ}$ , it was stopped by adding 0.1 ml of crystalline alcohol dehydrogenase (0.3 mg protein/ml 0.05M Tris-HCl buffer, pH 7.5) and 0.1 ml of N acetaldehyde to oxidize the residual NADH which interferes in the diazotization reaction (Medina and Nicholas, 1957; Stulen, 1970). The amount of nitrite produced was determined as described in Section 2.7.2.

Specific activity of nitrate reductase is expressed in nmoles  $NO_2^-$  produced/15 min/mg protein.

#### 2.5.4.2 Nitrite reductase

Nitrite reductase activity was determined by following the rate of nitrite utilization. The enzyme activity in the Pl00 fraction was assayed in test tubes (1 x 7.5 cm) stoppered with rubber subaseal caps (Size 17, Wm Freeman, Barnsley, York, U.K.). Nitrite reductase activity of the purified enzyme preparation and also of the various fractions obtained during the purification of the enzyme was, however, assayed in Warburg flasks fitted with stopcocks and with their sidearms stoppered with subaseal caps.

For determining the enzyme activity with sulphide as an electron donor, the reaction mixture contained the following: Tris-HCl buffer (pH 7.5), 35 µmoles; NaNO2, 1 µmole; Na2S, 2 µmoles and 0.1 ml of enzyme, in a final volume of 1 ml. The test tubes or the Warburg flasks were thoroughly evacuated and filled with a high purity  $N_2$  gas. After 5 min preincubation at  $30^{\circ}$ , the reaction was started by injecting, with an air-tight glass microsyringe (Scientific Glass Engineering Co., Melbourne, Australia), a freshly prepared solution of sulphide through the subaseal caps. The reaction was terminated after 15 min by removing the subaseal caps and adding 1 ml of 10% (w/v) zinc acetate in 10% (v/v) ethanol to the reaction mixture. This procedure precipitated sulphide which would otherwise interfere with the diazotization reaction. The contents of the Warburg flasks were then transferred to test tubes and centrifuged at 2,000x g for 15 min in a bench centrifuge (M.S.E.). Nitrite, in an appropriate aliquot of the supernatant fraction, was determined as described in Section 2.7.2.

Except for the reductant, the composition of the reaction mixture and the experimental details for determining nitrite reductase activity with NADH plus menadione as the electron donating system, were the same as described above for the assay with sulphide as the reductant. The reaction mixture, which in addition contained 2 µmoles menadione, was dispensed into either test tubes or Warburg flasks. These were then evacuated and filled with pure  $N_2$  gas. The reaction was started by injecting 2 µmoles of NADH through the subaseal caps. After incubating at  $30^{\circ}$  for 15 min, the reaction was terminated with 1 ml of 10% (w/v) zinc acetate in 10% (v/v) ethanol. After centrifuging, an appropriate aliquot of the supernatant was taken for determining the residual nitrite.

For ascorbate plus PMS as the electron donor system, the reaction mixture and the assay procedure were as described for sulphide except that ascorbate plus PMS replaced sulphide as the reductant. The reaction mixture in addition contained 1 µmole of PMS. The reaction was initiated by injecting 2 µmoles of neutralized solution of ascorbate *via* the subaseal into the reaction mixture. The reaction was terminated after 15 min as described previously and after centrifuging, nitrite was determined in an appropriate aliquot of the supernatant by the procedure described in Section 2.7.2.

Specific activity of nitrite reductase is expressed in nmoles  $NO_2^-$  utilized/15 min/mg protein.

#### 2.5.4.3 Cytochrome oxidase

The activity of cytochrome oxidase was determined from the rate of oxygen uptake measured by Rank Oxygen Electrode as described in Section 2.6.1. After adding 1.8 ml of 0.05M Tris-HCl buffer

(pH 7.5) to the reaction vessel, the contents were equilibrated for 5 min at 30°. Then 0.1 ml of the enzyme preparation was injected into the reaction vessel with a Hamilton gas-tight microsyringe. After a further equilibration for 2 min, the reaction was started by injecting 2 µmoles of freshly prepared solution of the electron donor.

The specific activity is expressed as nmoles  $O_2$  utilized/min/mg protein.

#### 2.5.5 Determination of Michaelis - Menten constants (K<sub>m</sub>)

For determining the  $K_m$  values, the enzyme activity was measured at varying concentrations of the substrate, in the presence of saturating concentrations of the other substrate. The  $K_m$  value was then estimated from a double-reciprocal plot of the velocity of the reaction against the substrate concentration as described by Lineweaver and Burk (1934).

#### 2.5.6 Determination of molecular weights

Molecular weights of nitrite reductase and the cytochrome  $c_{554}$ , isolated from *T. denitrificans* (Section 2.5.7), were estimated by gel filtration through Sephadex G-150 or G-75 columns by the procedure of Andrews (1964, 1965).

#### 2.5.6.1 Nitrite reductase

Sephadex G-150 column (2.5 x 50 cm), prepared as described in Section 2.6.8.2, was equilibrated for 2 days with 0.05M Tris-HCl buffer (pH 7.5) containing 0.1M KCl. Void volume of the column was determined from the elution of blue dextran. The column was calibrated with a mixture of the following marker proteins of known molecular weights:  $\gamma$ -globulin (160,000), bovine serum albumin (68,000), ovalbumin (45,000) and myoglobin (17,800). Fractions, 3 ml each, were collected at 4<sup>0</sup> using L.K.B. Ultrasac fraction collector. The elution of these proteins from the column was followed by absorbance at 280 nm and at 410 nm for myoglobin. The purified nitrite reductase was then loaded onto the column either alone or along with the marker proteins. The elution of nitrite reductase from the column was monitored by assaying the enzyme activity in the various fractions with sulphide as an electron donor, as described in Section 2.5.4.2. Molecular weight of nitrite reductase was calculated according to the following equation of Squire (1964):

$$M^{1/3} = 151[1.47 - (V_e/V_o)^{1/3}]$$

where  $V_e$  = elution volume of the protein  $V_o$  = void volume of the column.

## 2.5.6.2 Cytochrome c<sub>554</sub>

Experimental details were as described for the estimation of molecular weight of nitrite reductase except that a Sephadex G-75 column (2 x 35 cm) was used and all the operations were carried out at room temperature.

### 2.5.7 Isolation of cytochrome c<sub>554</sub> and flavin from nitrate reductase

Nitrate reductase was dissociated from the cell membranes (P100 fraction) by the procedure described in Section 2.5.3. The solubilized enzyme was placed in a boiling water bath for 15 min. After cooling, the precipitated proteins were removed by centrifuging at 100,000x g for 30 min. The supernatant fraction thus obtained, was loaded onto a DEAE-cellulose column (DE 32; 3 x 12 cm) which had been prepared and

equilibrated with 0.05M Tris-HCl buffer (pH 7.5) as described in Section 2.6.8.1. The column was eluted sequentially at room temperature with 100 ml each of 0.05M Tris-HCl buffer (pH 7.5) containing 0.1 and 0.2M NaCl respectively and 3 ml fractions were collected using L.K.B. Ultrasac fraction collector. Red-coloured fractions containing cytochrome c were recovered with the buffer containing 0.1M NaCl whereas yellow-coloured fractions were obtained on subsequent elution of the column with the buffer containing 0.2M NaCl. The red and yellow fractions were pooled separately, desalted by passing through a Sephadex G-25 column (1.25 x 15 cm) and concentrated in a rotatory vacuum evaporator at  $30^{\circ}$ .

## 2.5.8 <u>Dissociation of nitrite reductase into cytochrome c and</u> haem d components

The purified nitrite reductase was dissociated into cytochrome c and haem d components by incubating it with 1 - 1.5 mg sodium dodecyl sulphate (SDS) per mg protein at 30<sup>°</sup> for 30 min in an oscillating water bath. The detergent-treated enzyme was loaded onto a Sephadex G-100 column (2.5 x 45 cm) which had previously been equilibrated with 0.025M Tris-HCl buffer (pH 7.5) as described in Section 2.6.8.2. The sample was eluted, at room temperature, with the equilibrating buffer and 3 ml fractions were collected in an L.K.B. Ultrasac fraction collector. The absorbance at 280 nm and 405 nm of each fraction was monitored.

2.6 General techniques

#### 2.6.1 Oxygen electrode

Oxygen uptake was measured polarographically with a Rank oxygen electrode (Bollisham, Cambridge, U.K.), connected to a Rikadenki recorder (Model B181). The electrode was standardized in the presence of 0.05M Tris-HCl buffer (pH 7.5) by the method of Dixon and Kleppe (1966). The full span of the chart paper corresponded to 0.25  $\mu$ moles 0<sub>2</sub>/ml buffer.

The reaction vessel was maintained at 30° by circulating water, from a temperature-controlled water bath, through the outer jacket. The reaction mixture in a final volume of 2 ml, contained either 0.05M or 0.025M Tris-HCl buffer (pH 7.5), the enzyme preparation and an appropriate reductant. The reaction was started by injecting the reductant from an air-tight glass micro-syringe, through an inlet port in the reaction vessel. The contents were continuously stirred by a small magnetic bar. The decrease in oxygen concentration in the reaction mixture was recorded.

#### 2.6.2 Irradiation of the samples with U.V. light

The enzyme preparation (1 ml) dispensed in 2 ml of 0.05M Tris-HCl buffer (pH 7.5), was poured into a small Petri-dish (5 cm diameter) and placed on a layer of aluminium foil in an ice bucket. The sample was exposed to U.V. light (HPW 125W, Phillips) from above at a distance of 15 cm. The control sample, also kept under identical conditions was, however, protected from the U.V. light by covering the Petri-dish completely with an aluminium foil.

#### 2.6.3 Absorption spectra

For spectral studies, a Shimadzu Multipurpose Recording Spectrophotometer, Model MPS-50L (Shimadzu Seisakusho Ltd, Kyoto, Japan) was used. The instrument was operated with photomultiplier at 400 volts

and automatic slit control. The spectra were recorded at room temperature in 3.5 ml quartz cuvettes (1 cm light path) in the absorbance range of either 0-1 or -0.1 to +0.1.

For absolute spectra, the sample cuvette contained 0.05M Tris-HCl buffer (pH 7.5) and the enzyme preparation in a final volume of 2.5 ml. Its spectrum was recorded against the buffer in the reference cuvette.

For difference spectra, both the sample and the reference cuvettes contained the enzyme preparation and 0.05M Tris-HCl buffer (pH 7.5) in a final volume of 2.5 ml. After adding a crystal of  $K_3Fe(CN)_6$  to the reference cuvette and an appropriate reductant to the sample cuvette, the reduced *versus* oxidized spectra of the sample was recorded.

Difference spectra of the enzyme preparations under anaerobic conditions were recorded in a special Thunberg cuvette fitted with a side-arm. In all these studies, an ordinary quartz cuvette was used for the reference solution. The enzyme preparation and 0.05M Tris-HCl buffer (pH 7.5), in a final volume of 2.5 ml, were dispensed into both the sample and the reference cuvettes. The reductant was placed in the side-arm of the sample cuvette. After thoroughly evacuating the sample cuvette, with a water pump, it was filled with high purity,  $O_2^-$  free  $N_2$  gas. This procedure was repeated 3 times. The reductant was then tipped into the cuvette from the side-arm and the contents were mixed thoroughly. A crystal of  $K_3$ Fe(CN)<sub>6</sub> was added to the reference cuvette and the difference spectra recorded.

#### 2.6.4 Redox-potential of cytochromes

Mid-point potential of cytochrome  $c_{554}$ , isolated from nitrate reductase (Section 2.5.7) was determined by the procedure of Davenport and Hill (1952), except that 1 cm open glass cuvettes were used.

Ferro-ferricyanide couple, E' of +0.423V (O'Reilly, 1973), was used for the reduction and oxidation of cytochrome  $c_{554}$ . The redox-potential was calculated from the Nernst equation:-

$$E_{O}^{*} = 0.423 - 0.059 \log \frac{K_{4}Fe(CN)_{6}}{K_{3}Fe(CN)_{6}} \times \frac{Ferricytochrome}{Ferrocytochrome}$$

The difference in the mid-point potential between cytochromes cand d of cd complex of nitrite reductase was determined from the following equation, according to the procedure of Shimada and Orii (1976):-

$$E_{m}(cyt. d) - E_{m}(cyt. c) = 0.059 \log x K.$$

where K = equilibrium constant and is calculated from the following relationship:

$$K = \frac{(c^{3+})(d^{2+})}{(c^{2+})(d^{3+})}$$

The equilibrium constant (K) was determined experimentally by measuring the % of cytochromes c and d in their oxidized and reduced forms at various time intervals after reducing the purified nitrite reductase with sulphide (see Section 2.6.3). The % of the reduced cytochromes c and d were calculated on the basis of the maximum reduction of these cytochromes obtained after the complete reduction of the enzyme with sulphide.

#### 2.6.5 Paper chromatography of flavins

Aliquots (50  $\mu$ l) of the flavin isolated from the solubilized nitrate reductase (Section 2.5.7) as well as of the authentic samples of FAD (2 mg/ml) and FMN (2 mg/ml) were spotted 7.5 cm apart on a

Whatman 3MM paper (45 x 57 cm). The chromatograms were developed for 30 hr in a descending sequence using t-butanol : water (60 : 40) as the solvent system. After drying, the flavins were located on the chromatograms by U.V. light.

#### 2.6.6 High voltage electrophoresis

The flavin, isolated from the solubilized nitrate reductase, was pre-incubated at  $30^{\circ}$  with either 20 mg snake venom pyrophosphatase for 30 min or with 20 mg of inorganic pyrophosphatase for 30 and 60 min respectively. Then 50 µl of each of the separated flavin as well as of those treated with the hydrolytic enzymes were spotted 2.5 cm apart onto a 3MM Whatman chromatographic paper (15 x 55 cm). Authentic standards of FAD and FMN were also spotted onto the paper which was then subjected to a high-voltage electrophoresis, using 0.1M sodium citrate buffer (pH 5.0), at 1500 V for 1 hr in an apparatus designed by Tate (1968). The electrophoretic mobilities of the flavins were determined under U.V. light, after drying the paper.

#### 2.6.7 Mass spectrometry

The products of nitrite reduction were identified in a mass spectrometer. The reaction was carried out in Warburg flasks, fitted with glass stopcocks and with their side-arms stoppered with subaseals. The reaction mixture in a final volume of 1 ml contained: 0.6 ml of 0.05M Tris-HCl buffer (pH 7.5), 2 µmoles of  ${}^{15}NO_2^-$  (27 atom % excess) and 0.2 ml of the enzyme preparation. The flasks were rigorously evacuated with a two-stage oil pump and filled with pure helium gas. This procedure was repeated three times. After pre-incubating for 5 min at 30<sup>°</sup> in a reciprocating water bath (100 oscillations/min), 5 µmoles of a freshly prepared sulphide solution was injected from a gas-tight syringe *via* the subaseals into the side-arms of the flasks. The reaction was started by mixing the contents. After various time intervals, the reaction was terminated by injecting 250 µl of 10M KOH solution into the flasks through the subaseals. The gases formed were then identified in a MS-2 Mass Spectrometer (Associated Electrical Industries, Manchester, U.K.).

#### 2.6.8 Preparation of columns

#### 2.6.8.1 DEAE-cellulose

Whatman DE-32 was prepared according to the instructions given in the Whatman Technical Bulletin IE2. The cellulose was treated with 0.5N HCl and washed with double-distilled water until the pH of the filtrate was 4.0. It was then treated with 0.5N NaOH and washed several times with double-distilled water until the pH of the filtrate The cellulose was finally equilibrated with 0.05M Tris-HC1 was 8.0. buffer (pH 7.5). The column was packed at  $2^{\circ}$  by pouring the homogenous slurry of the ion exchanger into a glass column. After the cellulose had settled, 0.05M Tris-HCl buffer (pH 7.5) was passed through the column until the pH of the effluent was the same as that of the equilibrating buffer. The sample was loaded onto the column and eluted with an appropriate buffer. For every experiment, DE-32 was regenerated afresh as described above. When not in use, the cellulose was stored at 2 $^{\circ}$ in 0.05M Tris-HCl buffer (pH 7.5) containing 0.1% (w/v) sodium azide.

#### 2.6.8.2 Sephadex and Sepharose 6B

The columns for gel chromatography on Sephadex G-25, G-100, G-150 and Sepharose 6B were prepared according to the instructions given by the manufacturers (Pharmacia Fine Chemicals, Uppsala, Sweden). The gel beads were suspended in 0.05M Tris-HCl buffer (pH 7.5) and stirred gently with a glass rod. After standing for 30 min, the fine particles were removed by decantation. This procedure was repeated a few times. The gel was then allowed to swell at room temperature in the same buffer for an appropriate time depending on the type of Sephadex used. After degassing under reduced pressure, the homogenous slurry of the gel was poured in one operation into a glass column until the desired column height was attained. An appropriate pressure was maintained during packing and running of the column. The column was equilibrated with 0.05M Tris-HCl buffer (pH 7.5). When not in use, the column was stored at  $2^{\circ}$  in the presence of 0.1% (w/v) sodium azide.

The void volume of the column was determined from the elution volume of blue dextran.

#### 2.6.9 Membrane ultrafiltration

Enzyme preparations were concentrated using a Diaflo PM-10 membrane, in an Amicon ultrafiltration unit (Amicon Corp, U.S.A.). The ultrafiltration unit, which was kept in ice during filtration, was operated under  $N_2$  gas at 20 lb/in<sup>2</sup>. The PM-10 membrane retains proteins of molecular weight greater than 10,000.

#### 2.7 Chemical determinations

#### 2.7.1 Protein

Protein was determined by the micro-biuret method of Itzhaki and Gill (1964), using bovine serum albumin as a standard. The optical density was recorded at 310 nm. Absorbance of the fractions, eluted from the columns during enzyme purification, was monitored at 280 and 260 nm in order to assess their protein contents. The protein concentration was calculated from the following equation as described

by Kalckar (1947):-

Protein (mg/ml) =  $1.45[E_{280}] - 0.74[E_{260}]$ .

2.7.2 Nitrite

Nitrite was determined by the procedure of Hewitt and Nicholas (1964). An aliquot containing 20-100 nmoles of nitrite was diluted to 2 ml with double-distilled water. The red azo dye was developed by adding 1 ml of 1% (w/v) sulphanilamide in N HCl, followed by 1 ml of 0.01% (w/v) N-l-naphthylethylenediamine dihydrochloride. After 15 min, these were read at 540 nm in a Shimadzu (QV-50) spectrophotometer. The concentration of nitrite was determined from a standard curve.

#### 2.7.3 Copper and iron

Copper was estimated according to the procedure described by Nambiar (1976). Purified nitrite reductase (containing 12.5 mg protein) was dried in acid-washed pyrex test tubes and digested for 20 min with 5 ml of an acid mixture of  $HNO_3$ ,  $HClO_4$  and  $H_2SO_4$  in the proportion of 40 : 4 : 1. After cooling, the volume was made to 20 ml with deionized water. Copper in the digested sample was precipitated with 4 ml of 0.25% (w/v) ammonium pyrrolidine dithiocarbamate and then extracted by phase separation into 4 ml of methyl isobutylketone. The upper organic phase was injected directly into Varian Atomic Absorption Spectrophotometer (Varian Techtron Model AA4) and read at  $3250A^{O}$ .

For determining the iron content, the samples were first digested with the acid mixture as described for copper. The digested samples were diluted to 25 ml with deionized water and their iron contents were determined at 2485A<sup>O</sup> in a Varian Atomic Absorption

#### Spectrophotometer.

#### 2.7.4 Haems c and d

A spectrophotometric procedure of Newton (1969), based on the absorption at 618 nm of the pyridine haemochromogen of green haem, was used to estimate the haem *d* content of nitrite reductase. The sample was made 25% (v/v) with respect to pyridine. This was followed by the addition of NaOH to a final concentration of 0.1M. The mixture was divided equally and dispensed into two cuvettes (1 cm light path). A crystal of  $K_3Fe(CN)_6$  was then added to the reference cuvette to oxidize the green haem which is otherwise autoreducible in the presence of pyridine and alkali. The difference in the A (618-650 nm) between sample and reference cuvettes was recorded and the haem *d* content was calculated from the extinction coefficient value of 19.6 mM<sup>-1</sup> cm<sup>-1</sup>.

The haem c content of nitrite reductase was determined from the absorption spectra of the alkaline pyridine haemochromogen, prepared as described above for haem d. After adding pyridine and alkali, a few crystals of Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub> were added to the sample cuvette. The difference in the absorbance at 550 nm between the sample and the reference cuvettes was measured. An extinction coefficient value of 30.2 mM<sup>-1</sup> cm<sup>-1</sup> (Horio *et al.*, 1961A) was used to calculate the amount of haem c in the sample.

#### 3. RESULTS

#### 3.1 Nitrate reductase

# 3.1.1 Distribution of sulphite-linked nitrate reductase in various cell fractions

The results in Table 4 show that most of the enzyme activity was located in the pellet fraction (Pl00), obtained after centrifuging the crude extract (Sl0) at 100,000x g for 2 hr. The specific activity of the enzyme in this fraction was 2.5-fold greater than that of the crude extract. Sulphite-linked nitrate reductase activity was low in the high speed supernatant fraction (Sl00).

#### 3.1.2 Enzyme activity with various electron donors

The effectiveness of various reductants and organic acids for the reduction of nitrate in the Pl00 fraction are presented in Table 5. The results show that besides sulphite, which is the physiological reductant for the enzyme, NADH was also equally effective for nitrate reduction. A relatively low activity was obtained with ascorbate plus TMPD whereas other metabolites including NADPH,  $\alpha$ ketoglutarate, lactate, pyruvate and succinate were ineffective.

## 3.1.3 <u>Comparison of the properties of the sulphite- and NADH-linked</u> nitrate reductase

## 3.1.3.1 <u>Nitrate reductase activity with either sulphite or sulphite</u> plus NADH as the electron donor

Since both sulphite and NADH were effective electron donors ' for nitrate reduction, the effect of various concentrations of sulphite

## Table 4.Distribution of sulphite-linked nitrate reductase in<br/>various cell fractions.

The various cell-fractions (S10, S100 and P100) were prepared in 0.05M Tris-HCl buffer (pH 7.5) as described in Sections 2.5.2.1 and 2.5.2.3. Nitrate reductase activity with sulphite as an electron donor was determined as described in Section 2.5.4.1.

Fraction	Procedure	Specific activity (nmoles NO2 produced/15 min/mg protein)
Ι	Homogenate prepared by passing the	152
	washed cells through a French pressure	
	cell, twice at 20,000 p.s.i.	
II	Supernatant (S10) left after	151
	centrifuging I at 10,000x $\underline{g}$ for 30 min.	
III	Supernatant fraction (S100) obtained	15
	by centrifuging II at 100,000x g for	
	90 min.	
IV	Pellet fraction (Pl00) from III	385
	dispensed in 0.05M Tris-HCl buffer	
	(pH 7.5)	

#### Table 5. Nitrate reductase activity with various electron donors.

The reaction mixture in a total volume of 2 ml, contained the following in µmoles: Tris-HCl buffer (pH 7.5), 85; NaNO<sub>3</sub>, 1; electron donor, 1; and 0.3 ml of cell-free extract (S10 containing 42 mg protein/ml). The reaction was carried out at 30<sup>°</sup> as described in Section 2.5.4.1.

Electron donor	Specific activity (nmoles NO2 produced/15 min/mg protein)
•	
Sodium sulphite	90
NADH	95
NADPH	4
α-Ketoglutarate	9
Sodium ascorbate	7
Ascorbate + N,N,N',N'- tetramethylparaphenylenediamine (TMPD)	22
Sodium pyruvate	8
Sodium succinate	12
Sodium lactate	9

alone as well as in the presence of 0.5 and 1.0 mM NADH was studied. The results in Table 6 indicate that the amount of nitrite produced with sulphite plus NADH was almost equivalent to the sum of nitrite formed by these reductants individually. This effect was observed even when both sulphite and NADH were used at optimal concentrations in the reaction mixture.

#### 3.1.3.2 Distribution

The results in Table 7 show that both the sulphite- and NADH-linked nitrate reductase activities had a similar pattern of distribution in the various cellular fractions. Enzyme activity, with either of these reductants, was almost exclusively associated with the membrane fraction (P100). Only negligible sulphite- and NADH-dependent nitrate reductase activities were found in the high speed supernatant fraction (S100). Specific activities of the enzyme, with either of these reductants, were about 2.5-fold higher in the pellet fraction (P100) than in the crude extract (S10).

#### 3.1.3.3 Incubation time, stability and pH optima

Incubation time. Nitrate reductase activity, with either sulphite or NADH as the reductant, was linear up to 15 min (Figure 1). Thus, an incubation time of 15 min was routinely used for assaying the enzyme activity.

Stability during storage. The NADH-linked nitrate reductase in the Pl00 fraction was more unstable during storage than the sulphitelinked activity. Thus, the enzyme activity with sulphite as a donor was not affected after storing the Pl00 fraction at  $0^{\circ}$  for 5 days whereas more than a half of the NADH-dependent activity was lost. However,

## Table 6.Nitrate reductase activity with either sulphite, NADHor sulphite plus NADH as the electron donors.

The details of the assay procedures as described in Section 2.5.4.1 except that the concentrations of sulphite or NADH were varied as indicated in the Table. The enzyme activity in the Pl00 fraction (0.5 mg protein) was determined with either sulphite, NADH or sulphite plus NADH as electron donors.

Concentration of sulphite (mM)	s produc without NADH	Specific activity (nmoles NO2 ced/15 min/mg pro with 0.5mM NADH	tein) with 1.0mM NADH
······································			
0	0	317 -	330
0.125	121	417	434
0.250	152	434	445
0.500	178	478	460
0.750	171	439	456
1.000	169	434	456
1.250	165	417	445
1.500	160	417	408

## Table 7.Distribution of sulphite - and NADH-linked nitratereductase in various cell fractions.

The various cell fractions (Sl0, Sl00 and Pl00) were prepared in 0.05M Tris-HCl buffer (pH 7.5) as described in Sections 2.5.2.1 and 2.5.2.3. Enzyme activity was determined with either sulphite or NADH as the electron donor as described in Section 2.5.4.1.

and the second					
Fraction		Specific activity (nmoles $NO_2^-$ produced/15min/mg protein)			
			Electron donor		
		Sulphite		NADH	
25 - 11					
Cell homogenate		128		126	
<b>S10</b>		117		117	
S100		23		9	
P100		300		340	
	322				

### Figure 1. Effect of incubation time on nitrate reduction.

Nitrate reductase activity was assayed in Pl00 fraction with either  $SO_3^{2-}$  or NADH as the electron donor as described in Section 2.5.4.1, except that the period of incubation has varied as shown in the Figure.



Figure 1

the activity with NADH as a donor was completely restored on preincubating the stored pellet (Pl00) with 5-10 mM GSH at 4<sup>°</sup> for 15 min (Table 8). The sulphite-linked nitrate reductase was not affected by GSH.

Stability during heat treatment. Incubation of the Pl00 fraction at  $40^{\circ}$  for 30 min had no effect on the sulphite-linked nitrate reductase activity. However, the heat treatment at  $45^{\circ}$  for 30 min or at  $50^{\circ}$  for 10 min lowered its activity by 38 and 88%, respectively (Figure 2A).

The NADH-linked nitrate reductase activity was diminished by about 85% after 10 min at  $50^{\circ}$ . Incubation at  $45^{\circ}$  or  $40^{\circ}$  for 30 min decreased the enzyme activity by 55 and 40% respectively (Figure 2B). Thus both the sulphite- and NADH-linked activities were equally sensitive to heat treatment at  $50^{\circ}$  but the sulphite-linked activity was relatively more stable during incubations at  $40^{\circ}$  and  $45^{\circ}$ .

*pH optima*. The pH optima of the sulphite- and NADH-linked nitrate reductase activities were 8.0 and 7.0 respectively (Figure 3A and B). A significantly lower sulphite-linked activity was recorded in phosphate buffer. Thus at pH 8.0, in the presence of phosphate buffer, only 30% of the activity with either Tris-HCl or Tris-maleate buffers was obtained. Nitrate reductase activity with NADH as an electron donor was, however, almost the same in either of these buffers.

#### 3.1.3.4 K values

Effect of nitrate concentration. The enzyme activity at varying concentrations of nitrate, with either sulphite or NADH as an

### Table 8. Stability of sulphite - and NADH-linked nitrate reductase during storage and the restoration of NADH-dependent activity with GSH.

Plo0 fraction, prepared as described in Section 2.5.2.3, was stored at  $0^{\circ}$ . After 5 days, nitrate reductase activity with either sulphite or NADH as an electron donor was assayed as described in Section 2.5.4.1 except that the enzyme preparation, in 0.05M Tris-HCl buffer (pH 7.5), was pre-incubated at  $0^{\circ}$  for 15 min with the specified concentrations of GSH prior to adding the other components of the reaction mixture. The enzyme activity before storage was 300 and 340 nmoles NO<sub>2</sub> produced/15 min/mg protein with sulphite and NADH, respectively, as the electron donors.

Final concentration of GSH (mM)	% of activity before storage Electron donor			
	Sulphite	NADH		
0	100	48		
0.25	100	79		
0.50	100	81		
1.00	100	85		
2.50	100	93		
5.00	98	100		
10.00	84	98		
15.00	82	88		

#### Figure 2. Thermal stability of nitrate reductase.

A :  $SO_3^{2-}$  as an electron donor. B : NADH as an electron donor.

Plo0 fraction (containing 0.5 mg protein), taken in 0.05 M Tris-HCl buffer (pH 7.5), was pre-incubated at either 40°, 45° or 50° for the prescribed times. The enzyme activity, with either  $SO_3^{2-}$  or NADH as the reductant, was determined as described in Section 2.5.4.1, after having equilibrated the enzyme preparation at 30° for 5 min. The results are expressed as % activity of the control samples (without heat treatment) which were 200 and 240 nmoles  $NO_2^{-}$ produced/15 min/mg protein with  $SO_3^{2-}$  and NADH respectively, as the electron donors.




Figure 3. Effect of pH on nitrate reductase activity.

A :  $SO_3^{2-}$  as an electron donor. B : NADH as an electron donor.

Nitrate reductase activity, with either  $SO_3^{2-}$  or NADH as the electron donor, was determined according to the procedure described in Section 2.5.4.1 except that 0.05 M of either phosphate, Tris-HCl or Tris-maleate buffer of indicated pH were used. The pH values given in the Figure correspond to the final pH of the reaction mixture.











electron donor, exhibited a normal Michaelis-Menten kinetics. The K<sub>m</sub> values for nitrate with either sulphite or NADH as the reductant, were 41.6 and 83.3  $\mu$ M respectively (Figure 4A and B).

Effect of sulphite or NADH concentrations. Nitrate reductase activity was determined at varying concentrations of either sulphite or NADH. The results in Figure 5A and B show that the rate of nitrate reduction increased proportionally with increasing concentrations of the reductant. The  $K_m$  values of 0.1 and 0.33 mM were obtained for sulphite and NADH respectively.

#### 3.1.4 <u>Characterization of the electron transport chain involved in</u> nitrate reduction

#### 3.1.4.1 Effects of inhibitors

Since sulphite and NADH were effective electron donors for nitrate reduction, the effects of various inhibitors on the enzyme activity in P100 fraction with both reductants were compared.

Sulphydryl reagents. The data in Table 9 show that the thiol-binding agents were potent inhibitors of nitrate reductase activity when either of the electron donors was used. The sulphitelinked nitrate reductase was strongly inhibited by N-ethylmaleimide (NEM) and to a lesser extent by 0.1 M p-CMB and by 5 mM iodoacetamide. The inhibitory effect of 1 mM NEM was almost completely reversed by 2 mM GSH. The NADH-linked nitrate reductase activity was much more sensitive than the sulphite-linked activity to p-CMB and iodoacetamide. The inhibition of the NADH-linked enzyme activity by 0.01 mM p-CMB and by 1 mM NEM was reversed by GSH.

## Figure 4. Double-reciprocal plot of the effect of nitrate concentration on enzyme activity.

A :  $SO_3^{2-}$  as the reductant. B : NADH as the reductant.

Nitrate reductase activity in Pl00 fraction was determined as described in Section 2.5.4.1 except that the concentration of nitrate in the reaction mixture was varied as indicated in the Figure. 1/V represents the reciprocal of nmoles  $NO_2^{-}$ produced/15 min.



2

\*



1

×



Figure 5. <u>A Lineweaver - Burk's plot of the effect of various</u> concentrations of electron donor on nitrate reductase activity.

A : SO<sub>3</sub><sup>2-</sup> as an electron donor.
B : NADH as an electron donor.

Nitrate reductase activity, in the P100 fraction containing 0.5 mg protein, was determined as described in Section 2.5.4.1 except that the reaction mixture contained various concentrations of either  $\text{SO}_3^{2-}$  (Figure 5A) or NADH (Figure 5B).  $\frac{1}{V}$  represents nmoles  $\text{NO}_2^-$  produced/15 min.



Figure 5B



### Table 9.Effect of inhibitors of -SH groups on nitrate reductaseactivity with either sulphite or NADH as the electron donor.

Plo0 fraction (0.6 mg protein) in 0.05M Tris-HCl buffer (pH 7.5) was incubated at  $30^{\circ}$  for 15 min with the inhibitor (final concentrations as indicated) before addition of other components of the reaction mixture. GSH was added, where indicated, to the enzyme preparation 5 min prior to starting the reaction with the appropriate electron donor.

Results are expressed as % inhibition with respect to the control sample (without inhibitor). The specific activity of the control sample with sulphite and NADH as the electron donors was 390 and 330 nmoles  $NO_2$  produced/15 min/mg protein respectively.

Addition to the reaction mixture	Final concentration (mM)	% inhil Electron Sulphite	oition n donor NADH
en en en anne en altimus anne en anne e			
p-Chloromercuribenzoate	0.01	10	100
p-Chloromercuribenzoate + GSH	0.01 + 2.0	10	37
p-Chloromercuribenzoate	0.1	58	100
p-Chloromercuribenzoate + GSH	0.1 + 2.0	68	84
Iodoacetamide	5.0	49	84
Iodoacetamide + GSH	5.0 + 5.0	49	84
N-Ethylmaleimide	1.0	88	62
N-Ethylmaleimide + GSH	1.0 + 2.0	10	20
N-Ethylmaleimide	5.0	95	75
N-Ethylmaleimide + GSH	5.0 + 2.0	93	69

Flavoprotein inhibitors. Amytal (5 mM) and rotenone (1 mM) inhibited the sulphite-linked nitrate reductase activity by about 70% whereas atabrine was not inhibitory (Table 10). The inhibition by either amytal or rotenone was not relieved by FMN or menadione.

The NADH-linked nitrate reductase on the other hand, was not affected by amytal but atabrine (2 mM) inhibited its activity by 76%. Nitrate reductase activity with either of the reductants, was inhibited by rotenone to almost the same extent. The inhibition of the NADH-dependent enzyme activity by atabrine was reversed completely by menadione whereas the effect of rotenone was relieved only partially.

Inhibitors of electron transport chain. HQNO, antimycin A and dicumarol were potent inhibitors of the sulphite-linked enzyme activity (Table 11). Menadione or coenzyme  $Q_{10}$  were not effective in reversing the inhibitory effects of any of these compounds.

The NADH-linked nitrate reductase was less sensitive to HQNO and antimycin A than was the sulphite-linked enzyme. The enzyme activity with NADH as an electron donor was markedly inhibited by 1 mM dicumarol. Menadione, but not coenzyme Q<sub>10</sub>, reversed the inhibitory effects of HQNO and antimycin A on the NADH-linked nitrate reductase activity.

Metal chelators. Sodium diethyldithiocarbamate (5 mM), bathocuproin (5 mM), azide (0.05 mM), cyanide (0.5 mM), KSCN (0.5 mM) and dithiol (10 mM) inhibited both the sulphite- and NADH-linked nitrate reductase activities to approximately the same extent (Table 12). Other chelating agents like 2,2'-bipyridyl, 8-hydroxyquinoline and urethane

# Table 10.Effects of flavin inhibitors on nitrate reductaseactivity with either sulphite or NADH as the electrondonor.

The composition of the reaction mixture was as described in Section 2.5.4.1. The enzyme preparation (Pl00 fraction), taken in 0.05M Tris-HCl buffer (pH 7.5), was pre-incubated at  $30^{\circ}$  for 10 min with various concentrations of the inhibitor prior to the addition of substrates. Menadione was added just before starting the reaction with the electron donor.

Results are expressed as % inhibition with respect to the control (without inhibitor). The specific activity of the control was 380 and 342 nmoles  $NO_2^-$  produced/15 min/mg protein with sulphite and NADH, respectively, as the electron donors.

Final concentration (MM)	% inhib Electron Sulphite	ition donor NADH	
	ic III		112.03
2.0	38	11	
5.0	69	22	
5.0 + 5.0	69	22	
0.5	Ō	59	
2.0	23	76	
2.0 + 5.0	23	2	
0.5	45	60	
1.0	76	84	
0.5 + 1.0	45	37	
	Final concentration (mM) 2.0 5.0 5.0 + 5.0 0.5 2.0 2.0 + 5.0 0.5 1.0 0.5 + 1.0	Final concentration (mM)% inhib Electron Sulphite $2.0$ 38 $5.0$ 69 $5.0 + 5.0$ 69 $0.5$ 0 $2.0$ 23 $2.0 + 5.0$ 23 $0.5$ 45 $1.0$ 76 $0.5 + 1.0$ 45	Final concentration (mM) $\%$ inhibition Electron donor Sulphite2.038115.069225.0 + 5.069220.50592.023762.0 + 5.02320.545601.076840.5 + 1.04537

#### Table 11. Effects of inhibitors of the electron transport chain on nitrate reductase activity with either sulphite or NADH as the electron donor.

Plo0 fraction (0.5 mg protein), in 0.05M Tris-HCl buffer (pH 7.5), was pre-incubated with the inhibitor for 10 min before adding the substrates. Menadione or coenzyme  $Q_{10}$  were included in the reaction mixture as indicated. The specific activity of the control sample (without inhibitor) was 260 and 230 nmoles  $NO_2$  produced/15 min/mg protein with sulphite and NADH respectively, as the electron donors.

	Final	% inhibition		
Inhibitor	concentration	Electron Sulphite	donor NADH	
51				
2-n-Heptyl-4-hydroxyquinoline- N-oxide (HQNO)	0.l µM	42	0	
ноло	0.5 μΜ	69	0	
HQNO + coenzyme Q10	0.5 µM + 0.1 mM	52	- -	
HQNO	5.0 µM	97	29	
HQNO + menadione	5.0 µM + 1 mM	97	3	
HQNO + coenzyme Q <sub>10</sub>	5.0 µM + 0.1 mM	97	17	
Antimycin A	0.05 mM	60	33	
	0.1 mM	70	45	
Antimycin A + menadione	0.1 mM + 1.0 mM	70	2	
+ coenzyme Q <sub>10</sub>	0.1 mM + 0.25 mM	60	36	
Dicumarol	1.0 mM	72	86	

# Table 12. Effects of metal chelators and other inhibitors on nitrate reductase activity with either sulphite or NADH as the electron donor.

Nitrate reductase activity was determined as described in Section 2.5.4.1. Plo0 fraction (0.5 mg protein) in 0.05 M Tris-HCl buffer (pH 7.5) was pre-incubated with the inhibitor for 10 min prior to adding the substrates. The specific activity of the control sample (without inhibitor) was 260 and 230 nmoles  $NO_2^-$  produced/15 min/mg protein with sulphite and NADH respectively, as the electron donors.

Inhibitor	Final concentration (mM)		% inhib Electron Sulphite	ition donor NADH
2,2'-Bipyridyl	5.0		17	3
Sodium diethyldithiocarbamate	5.0		51	49
8-Hydroxyquinoline	5.0		0	4
Bathocuproin	5.0		38	59
Urethane	5.0	•	16	11
Sodium azide	0.01 0.05		56 85	41 71
Potassium cyanide	0.5		58	34
Potassium thiocyanate	0.5		64	46
Dithiol	10.0	L.	98	70

did not affect either of these activities.

Chlorate. Both the sulphite- and NADH-linked nitrate reductase activities were inhibited by chlorate. Double-reciprocal plots of the effect of chlorate on the enzyme activity at various concentrations of nitrate are presented in Figure 6A and B. Chlorate inhibited the NADH-dependent activity competitively with respect to nitrate while it showed a mixed type inhibition on the sulphite-linked nitrate reductase.

#### 3.1.4.2 Effect of U.V. irradiation

Irradiation of the S10 and P100 fractions with ultra-violet light resulted in a profound inhibition of the sulphite- and the NADH-linked nitrate reductase activities (Figure 7A and B). An exposure of an enzyme preparation to U.V. light for 90 min, as described in Section 2.6.2, resulted in a loss of 45 and 68% of the sulphitelinked activity in the S10 and P100 fractions respectively. The NADH-linked nitrate reductase activity in the S10 and P100 fractions were also affected to a similar extent.

Inclusion of coenzyme  $Q_{10}$  (0.1 mM) in the assay mixture failed to reverse the effect of U.V. irradiation for 60 min, on nitrate reductase activity in Pl00 fraction with either sulphite or NADH as the reductant.

Menadione had no effect on the sulphite-linked enzyme activity in the Pl00 fraction which had been irradiated with U.V. light for 60 min. The effects of menadione on the NADH-linked nitrate reductase activity of an untreated and U.V.-irradiated Pl00 fractions are shown in Figure 8A and B. At lower concentrations, menadione stimulated

Figure 6. Double-reciprocal plot of the effect of potassium chlorate at varying concentrations of nitrate on nitrate reductase activity.

A :  $SO_3^{2-}$  as the reductant. B : NADH as the reductant.

Nitrate reductase activity was determined with either  $SO_3^{2-}$ or NADH as the reductant as described in Section 2.5.4.1. Ploo fraction (0.6 mg protein), in 0.05 M Tris-HCl buffer (pH 7.5), was pre-incubated for 10 min with either 0.5 mM or 1.0 mM KClO<sub>3</sub> prior to the addition of the specified concentrations of nitrate. The reaction was started by adding the respective electron donor.

• Without 
$$Clo_3^-$$
  
 $\Delta$  0.5 mM  $Clo_3^-$   
0 1.0 mM  $Clo_3^-$ 





Figure **6**B



### Figure 7. Effect of U.V. irradiation on nitrate reductase activity in Sl0 and Pl00 fractions.

A :  $SO_3^{2-}$  as an electron donor. B : NADH as an electron donor.

Sl0 and Pl00 fractions were prepared as described in Sections 2.5.2.1 and 2.5.2.3 respectively. 1 ml aliquot of Sl0 fraction (34 mg protein/ml) or Pl00 fraction (25 mg protein/ml), was dispensed in 2 ml of 0.05 M Tris-HCl buffer (pH 7.5) and exposed to U.V. light as described in Section 2.6.2. At the times indicated, 100  $\mu$ l aliquots were withdrawn and assayed for nitrate reductase activity with either SO<sub>3</sub><sup>2-</sup> or NADH as the electron donor as described in Section 2.5.4.1. The results are presented as % of the initial activity (prior to irradiation), which for SO<sub>3</sub><sup>2-</sup> and NADH dependent activities were 123.7 and 123 nmoles NO<sub>2</sub> produced/15 min/mg protein (Sl0 fraction) and 182.4 and 182 nmoles NO<sub>2</sub> produced/15 min/mg protein (Pl00 fraction) respectively.





Figure 7A

Figure 8. Effects of various concentrations of menadione on nitrate reductase activity in the U.V.-irradiated Pl00 fraction.

> A :  $SO_3^{2-}$  as the reductant. B : NADH as the reductant.

Plo0 fraction was irradiated with U.V. light for 60 min as described in Section 2.6.2. Nitrate reductase activity in the presence of menadione, both in the control and the U.V.irradiated samples, was determined with either  $SO_3^{2-}$  or NADH as the reductant as described in Section 2.5.4.1.

0-----0

P100 fraction

Pl00 fraction, irradiated with
U.V. light for 60 min



the enzyme activity of the untreated as well as of the U.V.-irradiated P100 fractions. However, 0.5 mM menadione reversed the inactivation caused by U.V. irradiation by about 25%.

#### 3.1.5 <u>Spectral properties of sulphite- and NADH-linked nitrate</u> reductase

The difference spectra of the Pl00 fraction, reduced with either sulphite or NADH, showed absorption maxima at 424, 524 and 554 nm. These absorption bands, which corresponded to those of reduced cytochrome c, disappeared on adding nitrate to either sulphiteor NADH-reduced enzyme (Figure 9). Nitrite had no effect on the spectra of the reduced enzyme. These results indicate that cytochrome c is involved in the reduction of nitrate.

# 3.1.6 Characterization of the electron transport chain linked to oxygen

#### 3.1.6.1 Various electron donors

The suitability of various compounds as hydrogen donors for  $O_2$  utilization was investigated in crude extract (S10). The results in Table 13 show that sulphite, NADH and, to a lesser extent, NADPH and ascorbate supported the uptake of  $O_2$  whereas succinate, pyruvate, lactate and  $\alpha$ -ketoglutarate were ineffective.

#### 3.1.6.2 <u>Comparison of oxygen uptake in various cell fractions with</u> either sulphite or NADH as reductant

The maximum rate of O<sub>2</sub> uptake was observed in the pellet fraction (PlOO) with either of the reductants (Table 14). The supernatant fraction (SlOO) had low activity. As shown earlier

#### Figure 9. Difference spectra of P100 fraction.

Plo0 fraction (6 mg protein) and 0.05 M Tris-HCl buffer (pH 7.5), in a final volume of 3 ml, were dispensed into each of the sample and the reference cuvettes. A crystal of  $K_3$ Fe(CN)<sub>6</sub> was added to the reference cuvette whereas to the sample cuvette, 2 µmoles of either SO<sub>3</sub><sup>2-</sup> or NADH was added and the difference spectra were recorded (Section 2.6.3).

Α.	 reduced versus oxidized spectra
в.	 spectra recorded after adding
3	$4  \mu$ moles of NaNO $_3$ to A



nm

#### Table 13. Oxygen uptake with various electron donors.

Oxygen uptake in S10 fraction was measured as described in Section 2.6.1. The reaction mixture in a total volume of 2 ml, contained the following in µmoles: Tris-HCl buffer (pH 7.5), 85; electron donor, 2; and 0.1 ml of S10 fraction (34 mg protein/ml).

Electron donor	-	Speci ( uti]	lfic act nmoles ( lized/mi protein)	ivity <sup>)</sup> 2 n/mg )	¥.
Sodium sulphite			32.3		
NADH			29.6		
NADPH			10.2		
Sodium succinate			3.6		
Sodium pyruvate			4.4		
Sodium lactate			2.1		
Sodium ascorbate			12.5		
α-Ketoglutarate			1.5		
				7	

### Table 14.Oxygen uptake by various cell fractions with eithersulphite or NADH as the electron donor.

The various cell fractions (Sl0, Sl00 and Pl00) were prepared in 0.05M Tris-HCl buffer (pH 7.5) as described in Sections 2.5.2.1 and 2.5.2.3. The rate of oxygen uptake was determined with an electrode system as described in Section 2.6.1.

i.

Fraction	Specific activi utilized/min/ Electron Sulphite	ty (nmoles O <sub>2</sub> mg protein) donor NADH	
	- *		
Disrupted cells	27.7	19.6	
S10	21.2	16.0	
S100	4.8	3.7	
P100	22.3	19.1	

(Table 7) nitrate reductase activity was also associated with this particulate fraction.

3.1.6.3 Effects of various inhibitors

Thiol-binding reagents. The effects of thiol-binding reagents on the sulphite-dependent  $O_2$  uptake are given in Table 15. All the three inhibitors diminished the uptake of  $O_2$ . The inhibition by either p-CMB or NEM was reversed by GSH. Comparison of these results with those in Table 9 indicate that the sulphite-linked nitrate reduction in the Pl00 fraction was also inhibited by these compounds.

Inhibitors of the electron transport chain. The effects of various inhibitors of electron transport chain on  $O_2$  uptake with sulphite as the reductant were also examined (Table 16). At 2 mM each, amytal, atabrine and rotenone diminished  $O_2$  uptake by 47, 63 and 71% respectively and 0.5 mM of either antimycin A or dicumarol inhibited the utilization of  $O_2$  by 70 and 59% respectively. At 5  $\mu$ M HQNO,  $O_2$  uptake was suppressed by 85%.

Metal chelators. Azide, cyanide and arsenite were potent inhibitors of the sulphite-linked O<sub>2</sub> uptake (Table 17). This activity was more sensitive to cyanide than the sulphite-linked nitrate reductase (Table 12). Bathocuproin, 8-hydroxyquinoline and 2,2'-bipyridyl, at 5 mM each, inhibited O<sub>2</sub> uptake by about 50% whereas urethane at the same concentration had no effect.

#### 3.1.7 Solubilization of sulphite-linked nitrate reductase

In order to examine the effectiveness of various detergents in solubilizing the membrane-bound nitrate reductase, the Pl00 fraction

### Table 15.Effects of sulphydryl group inhibitors on oxygenuptake with sulphite as an electron donor.

Oxygen uptake was determined as described in Section 2.6.1. Pl00 fraction (1.3 mg protein) in 0.05 M Tris-HCl buffer (pH 7.5) was pre-incubated with the inhibitor for 10 min at  $30^{\circ}$ . GSH was added, as indicated, just before starting the reaction with sulphite. The enzyme activity is expressed as % inhibition as compared to the activity of control sample (without inhibitor) which was 18 nmoles  $O_2$  utilized/min/mg protein.

Inhibitor	Final concentration (mM)	% inhibition
2		
p-Chloromercuribenzoate	0.1	26
	1.0	43
p-Chloromercuribenzoate + GSH	1.0 + 1.0	18
N-Ethylmaleimide	1.0	40
	2.0 .	53
N-Ethylmaleimide + GSH	1.0 + 2.0	0
Iodoácetamide	1.0	20
	2.0	24
	5.0	30
Iodoacetamide + GSH	5.0 + 5.0	31

### Table 16.Effects of inhibitors of the electron transport chainon oxygen uptake with sulphite as an electron donor.

Oxygen uptake was measured as described in Section 2.6.1. Plo0 fraction was pre-incubated for 10 min at  $30^{\circ}$  with inhibitor prior to adding the electron donor. The specific activity of the control sample (without inhibitor) was 22 nmoles O<sub>2</sub> utilized/min/mg protein.

Inhibitor co	Final ncentration	% inhibition
Amytal	2 mM	47
	5 "	68
Atabrine	0.5 mM	30
-	1.0 "	50
	2.0 "	63
Rotenone	0.5 "	31
	1.0 "	44
	2.0 "	71
Antimycin A	0.05 "	54
	0.5 "	67
2-n-Heptyl-4-hydroxyquinoline-N-oxide	0.1 µM	52
	0.5 "	81
	5.0 "	85
Dicumarol	0.5 mM	59
	1.0 "	64
	2.0 "	75
s a		

### Table 17.Effects of metal chelators and other inhibitors on<br/>oxygen uptake with sulphite as an electron donor.

Experimental details as in Table 16. The specific activity of the control sample (without inhibitor) was 27 nmoles  $O_2$  utilized/min/mg protein.

Inhibitor	Final concentration (mM)	% inhibition
Sodium azide	0.01	. 53
	0.1	70
	*	
Potassium cyanide	0.01	86
	0.1	92
5 X		
Sodium arsenite	0.2	52
4	0.1	77
Bathocuproin	5.0	53
-		
Urethane	5.0	14
8-Hydroxyquinoline	5.0	55
<u>م</u> م م		
2 2'-Binyridyl	5.0	40
ςις - ατήλεταλε	J • U	

(30 mg protein/ml) was incubated at 30° for 30 min with either of the following detergents: Tween 80 (2%, v/v), digitonin (0.5 mg/mg protein), Triton X-100 (10%, v/v), sodium deoxycholate (0.05 mg/mg protein) and sodium dedocyl sulphate (0.05 mg/mg protein). After treatment with the detergents, the Pl00 fraction was centrifuged at 100,000x <u>g</u> for 60 min and the enzyme activity was determined in the supernatant fraction. Among the various detergents used, only sodium deoxycholate and sodium dedocyl sulphate were effective in dissociating the enzyme from the membranes. Less than 30% of the enzyme was solubilized on incubating the Pl00 fraction for 30 min with lipase II (0.3 mg/mg protein).

The effects of various concentrations of sodium deoxycholate on the solubilization of the sulphite-linked nitrate reductase are illustrated in Figure 10. Maximum solubilization of the enzyme was obtained on incubating the Pl00 fraction with 0.06 mg sodium deoxycholate per mg protein.

The effects of incubation of Pl00 fraction, with the optimum concentration of sodium deoxycholate at 30<sup>°</sup>, for various time intervals are illustrated in Figure 11. Maximum solubilization was achieved after 30 min incubation.

#### 3.1.8 Inactivation of NADH-linked nitrate reductase on solubilization and restoration of its activity by PMS and menadione

Incubation of the Pl00 fraction with sodium deoxycholate resulted in a marked loss in the specific activity of the NADH-linked nitrate reductase and only 10% of its original activity remained in

Figure 10. Eff

Effect of various concentrations of sodium deoxycholate on the solubilization of the sulphite-linked nitrate reductase in P100 fraction.

Plo0 fraction, prepared as described in Section 2.5.2.3, was diluted with 0.05 M Tris-HCl buffer (pH 7.5) so that its protein concentration was 30 mg/ml. 2 ml aliquots of this Plo0 fraction were then incubated with the specified concentrations of sodium deoxycholate at  $30^{\circ}$  for 30 min. After centrifuging at 100,000x g for 60 min, nitrate reductase activity in the supernatant fractions was assayed by the procedure described in Section 2.5.4.1.





#### Figure 11. Effect of incubation time on the solubilization of the sulphite-linked nitrate reductase from P100 fraction by sodium deoxycholate treatment.

Experimental details are as in Figure 10 except that 2 ml aliquots of P100 fraction (30 mg protein/ml) were incubated with 0.06 mg sodium deoxycholate/mg protein at  $30^{\circ}$  for the indicated periods. Nitrate reductase activity in the supernatant fractions, after centrifuging at 100,000x g for 60 min, was then determined according to the procedure described in Section 2.5.4.1.





the deoxycholate-treated fraction. This activity was not recovered even after desalting, either the treated Pl00 fraction or the Sl00 fraction obtained from it, through a Sephadex G-25 column (Table 18). On the other hand, the specific activity of the sulphite-linked nitrate reductase increased after incubation of the Pl00 fraction with sodium deoxycholate.

An attempt was made to restore the NADH-linked nitrate reductase activity in the deoxycholate-treated Pl00 fraction by supplementing the reaction mixture with various compounds, namely GSH, FMN, PMS, PMS plus mammalian cytochrome c or menadione (Table 19). FMN or GSH did not have any effect whereas PMS and menadione increased the NADH-linked enzyme activity of the deoxycholate treated Pl00 fraction by 40 and 70% respectively. The inclusion of mammalian cytochrome c along with PMS did not enhance the enzyme activity further.

Since menadione was the most effective in restoring the enzyme activity, the effects of graded amounts of menadione on the NADH-linked nitrate reductase activity of the deoxycholate-treated Pl00 fraction were examined. The results in Figure 12 show that the enzyme activity in this fraction was fully restored by 2.5 mM menadione. However, the enzyme activity of the untreated Pl00 fraction was progressively inhibited with increasing concentrations of menadione in the reaction mixture.

## 3.1.9 Purification of the sulphite-linked nitrate reductase and properties of a partially purified enzyme

#### 3.1.9.1 Purification

All operations during the preparation of cell-free extracts
### Table 18. Effect of sodium deoxycholate treatment on nitrate reductase activity with either sulphite or NADH as the electron donor.

Plo0 fraction, prepared as in Section 2.5.2.3, was incubated with 0.06 mg sodium deoxycholate/mg protein for 30 min at  $30^{\circ}$ . A portion of this preparation was passed through a Sephadex G-25 column (1.25 x 20 cm). The remainder was centrifuged at 100,000x <u>g</u> for 60 min and the supernatant fraction thus obtained was also passed through a Sephadex G-25 column as described above. Nitrate reductase activity in these preparations was determined with either sulphite or NADH as the electron donor as described in Section 2.5.4.1.

Fraction	Procedure	Specific activity	
		Electron d Sulphite	lonor NADH
		2	
I	P100 fraction.	300	340
II	Pl00 fraction treated with sodium deoxycholate for 30 min at 30 <sup>0</sup> .	368	31.5
III	Fraction II passed through a Sephadex G-25 column.	318	22
IV	Supernatant fraction obtained after centrifuging fraction II at $100,000 \times g$ for 60 min.	453	24
V	Fraction IV passed through a Sephadex G-25 column and eluted with 0.05 M	428	16
	Tris-HCl buffer (pH 7.5).		

Specific activity : nmoles  $NO_2$  produced/15 min/mg protein.

## Table 19.Restoration of NADH-linked nitrate reductase in<br/>deoxycholate-treated P100 fraction.

Plo0 fraction was incubated with sodium deoxycholate and then passed through a Sephadex G-25 column (1.25 x 20 cm) as described in Table 18. Nitrate reductase activity in the pooled fractions from the Sephadex column was determined with NADH as the reductant. The reaction mixture was the same as described in Section 2.5.4.1 except that in addition it contained either GSH, FMN, PMS, mammalian cytochrome c or menadione at the concentrations given in the Table.

12	Addition to the basic reaction mixture	conc	Final entration (mM)	Specific activity (nmoles NO2 produced/15 min/mg protein)
	None		-	22.7
	GSH		2.0	25.0
	FMN		0.1	23.0
	Phenazine methosulphate	(PMS)	0.5	127
	PMS + cytochrome c		0.5 + 0.025	139
	Menadione		1.0	235
			а	

Figure 12. Restoration by menadione of NADH-linked nitrate reductase activity in the deoxycholate-treated Plo0 fraction.

Plo0 fraction was treated with sodium deoxycholate at 30<sup>°</sup> for 30 min as described in Figure 10. Nitrate reductase activity in the untreated and the deoxycholate-treated Plo0 fractions was assayed, with NADH as the reductant, according to the procedure given in Section 2.5.4.1 except that specified concentrations of menadione were also included in the reaction mixture.

• P

P100 fraction

0-----0

deoxycholate-treated P100 fraction





and enzyme purification were carried out at 4<sup>0</sup> unless otherwise stated.

Washed cells, suspended (25%, w/v) in 0.05 M Tris-HCl buffer (pH 7.5), were disrupted in a French pressure cell as described in Section 2.5.2.1. The resultant homogenate was centrifuged at 10,000x g for 30 min and the supernatant fraction (Fraction I, Table 20) thus obtained, was further centrifuged at 100,000x g for 2 hr. The pellet fraction (Pl00) was suspended in 0.05 M Tris-HCl buffer (pH 7.5) using a glass homogenizer. This suspension was diluted with the same buffer until its protein concentration was 30 mg/ml. This fraction (Fraction II) was treated with an ultrasonic probe (20 Kc/sec) for 30 sec, and then incubated with sodium deoxycholate as described in Section 2.5.3. The deoxycholate-treated Pl00 fraction was centrifuged at 100,000x g for 60 min. The reddish-brown supernatant fraction (Fraction III, Table 20) was loaded onto a Sepharose 6B column (2.5 x 40 cm; void volume, 60 ml) which had been equilibrated with 0.05 M Tris-HCl buffer (pH 7.5). The sample was eluted with the equilibrating buffer. The enzyme was recovered in the fractions immediately after the void volume of the column (Figure 13). The active fractions (11-18) were pooled (Fraction IV, Table 20).

The summary of this purification procedure given in Table 20 shows that a 6-fold purified enzyme, with an overall recovery of 84%, was obtained.

#### 3.1.9.2 Properties

Effect of enzyme concentration. The enzyme activity was linear up to a protein concentration of 0.48 mg in the assay mixture (Figure 14).

## Figure 13. Elution profile of nitrate reductase from a Sepharose 6B column.

Fraction III (Table 20) was loaded on to a Sepharose 6B column (2.5 x 40 cm) which had previously been equilibrated with 0.05 M Tris-HCl buffer (pH 7.5). The sample was eluted with the same buffer at a flow rate of 20 ml/hr and 3 ml fractions were collected.

Nitrate reductase activity with  $SO_3^{2-}$  as an electron donor was determined in 1.0 ml aliquots of each fraction as described in Section 2.5.4.1. The protein concentration was determined from the absorbance at 280 nm.

O----O Nitrate reductase activity



Figure 13

#### Table 20. Purification of sulphite-linked nitrate reductase.

Enzyme activity was determined by measuring the rate of nitrite production as described in Section 2.5.4.1. Washed cells, suspended in 0.05M Tris-HCl buffer (pH 7.5), were disrupted in French pressure cell (Section 2.5.2.1). The supernatant fraction (Sl0), obtained after centrifuging the broken cells at 10,000x g for 30 min, was used as the starting material for the purification of nitrate reductase.

Total	activity	
-------	----------	--

:

:

 $\mu$ moles NO<sub>2</sub> produced/15 min.

Specific activity

 $\mu$ moles NO<sub>2</sub> produced/15 min/mg protein.

#### Table 20. Purification of sulphite-linked nitrate reductase.

Fraction	Procedure	Total activity	Total protein (mg)	Specific activity	Fold purification	% recovery
I	Supermatant fraction (S10) after centrifuging the disrupted cells at 10,000x g for 30 min	330.6	1827	0.18	1.0	
II	Particulate fraction (P100) left after centrifuging I at 100,000x g for 2 hr and suspended in 0.05 M Tris-HCl buffer (pH 7.5)	273.4	690	0.40	2.2	83
III	P100 fraction ultra- sonicated for 30 sec and treated with sodium deoxycholate (0.06 mg/mg protein) for 30 min at 30°C; supernatant fraction after centrifuging at 100,000x g for 60 min	254.0	370	0.69	3.8	77
IV	Fraction III eluted from a Sepharose-6B column with 0.05 M Tris-HC1 buffer (pH 7.5)	276	260	1.1	5.9	84

Figure 14. Effect of protein concentration on the sulphitelinked nitrate reductase activity.

Nitrate reductase activity, with  $SO_3^{2-}$  as the reductant, was determined as described in Section 2.5.4.1 except that the amount of the partially purified enzyme preparation (Fraction IV, Table 20) in the reaction mixture was varied over the range of 0-650 µg protein. The assay was done at the following concentrations of  $SO_3^{2-}$  and  $NO_3^{-}$ .

 $\sim$  0 2 µmoles each of SO<sub>3</sub><sup>2-</sup> and NO<sub>3</sub><sup>-</sup> 0 -  $5 \mu$  moles each of  $SO_3^{2-}$  and  $NO_3^{-}$ 





Stability of the enzyme during storage. The partially purified enzyme (Fraction IV, Table 20) was more stable at  $0^{\circ}$  than at either  $5^{\circ}$  or  $-15^{\circ}$ . Storing the enzyme preparation at  $0^{\circ}$  and  $-15^{\circ}$  for 36 hr resulted in a loss of 59 and 92% of the enzyme activity respectively (Table 21). As compared to the partially purified enzyme, nitrate reductase activity in the Pl00 fraction was more stable since 80% of the initial activity was retained even after storing it at  $0^{\circ}$  for one month.

Heat treatment. The partially purified enzyme was also more sensitive to heat treatment than the enzyme in Pl00 fraction. Incubation of the partially purified enzyme preparation at  $50^{\circ}$  for 5 min resulted in complete loss of nitrate reductase activity and its activity was lowered by 80 and 30% after 15 min at 45 and  $40^{\circ}$  respectively (Figure 15).

# 3.1.10 Characterization of cytochrome c and flavin isolated from nitrate reductase

#### 3.1.10.1 Cytochrome c

The red-coloured fraction isolated from the solubilized nitrate reductase (Section 2.5.7) was identified as cytochrome c from its spectral and physical properties.

On reduction with sodium dithionite, it showed absorption bands at 417, 524 and 554 nm (Figure 16). This spectra corresponded to that of cytochrome c in the PlOO fraction. The evidence for the participation of cytochrome c in nitrate reductase, as deduced from the spectral investigations, has already been presented (Figure 9). This isolated cytochrome c, in addition, was reduced enzymically by sulphite in the presence of nitrate reductase. The redox-potential of the isolated cytochrome c, determined as described in Section 2.6.4,

### Table 21. Stability of the partially purified nitrate reductase on storage for 36 hr at $5^{\circ}$ , $0^{\circ}$ and $-15^{\circ}$ .

Aliquots of the partially purified enzyme (Fraction IV, Table 20) were stored at the temperatures given in the Table. After 36 hr, the enzyme activity was assayed as described in Section 2.5.4.1. The results are expressed as % of the initial activity (before storage) which was 1060 nmoles  $NO_2$  produced/15 min/mg protein.

•	Storage temperature	% Initial activity
	5 <sup>0</sup>	34
	0 <sup>0</sup>	41
	-15 <sup>0</sup>	8

## Figure 15. Thermal stability of partially purified sulphite-linked nitrate reductase.

Aliquots of a partially purified nitrate reductase (Fraction IV, Table 20) were kept at either 40, 45 or  $50^{\circ}$ . Samples were withdrawn at the indicated times and after equilibrating at  $30^{\circ}$  for 5 min, were assayed for nitrate reductase activity according to the procedure given in Section 2.5.4.1.

The results are presented as activity of the control sample (prior to heat treatment). Activity of the control was 1000 nmoles  $NO_{2}^{-}$  produced/15 min/mg protein.

•	Pre-incubated at	50 <sup>0</sup>
00	Pre-incubated at	45 <sup>0</sup>
00	Pre-incubated at	40 <sup>0</sup>



# Figure 16. Spectra of sodium dithionite-reduced cytochrome $c_{554}$ , separated from solubilized nitrate reductase.

Cytochrome  $c_{554}$  was separated from the solubilized nitrate reductase (Fraction III, Table 20) as described in Section 2.5.7. Its spectrum was recorded, against 0.05 M Tris-HCl buffer (pH 7.5), after reducing it with a crystal of Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub>.



Figure 16

was +0.223V. Its molecular weight, estimated by gel-filtration through Sephadex G-75, was 13,000. Thus, both its redox-potential and molecular weight were similar to those of mammalian cytochrome c.

This bacterial cytochrome c, in its reduced state, served as an effective electron donor for the reduction of nitrate by the solubilized enzyme (Fraction IV, Table 20). A plot of the enzyme activity against various concentrations of the reduced cytochrome cexhibited a typical Michaelis - Menten kinetics (Figure 17). The  $K_m$  value for the reduced cytochrome c was 1.53  $\mu$ M which reflects a very high affinity of the enzyme for the cytochrome. However, the ability to reduce nitrate enzymically was not specific to the bacterial cytochrome c since, as shown in Figure 18, reduced mammalian cytochrome c was equally effective. The apparent  $K_m$  value for the mammalian cytochrome c was 1.33  $\mu$ M.

#### 3.1.10.2 Flavin

The yellow-coloured fraction isolated from nitrate reductase (Section 2.5.7) showed an absorption band at 450 nm which disappeared on adding sodium dithionite. These spectral properties indicated that it was a flavin compound.

The isolated flavin, along with authentic samples of FMN and FAD, were chromatographed on 3MM Whatman paper using t-butanol : water (60 : 40) as the solvent as described in Section 2.6.5. On examining the dried chromatograms under U.V. light, the flavin isolated from nitrate reductase was found to have co-migrated with standard FAD (Figure 19).

# Figure 17. Effect of various concentrations of cytochrome $\underline{c}_{554}$ on nitrate reductase activity.

Cytochrome  $c_{554}$ , separated from solubilized nitrate reductase (Fraction III, Table 20) as described in Section 2.5.7, was reduced with Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub> and the excess reductant was removed by passing it through a Sephadex G-25 column (1.2 x 15 cm). The reduced cytochrome  $c_{554}$  was used, in place of SO<sub>3</sub><sup>2-</sup>, as an electron donor for the activity of solubilized nitrate reductase (Fraction III, Table 20) as described in Section 2.5.4.1.

A Lineweaver - Burk's plot of the effect of various concentrations of cytochrome  $c_{554}$  on the rate of nitrate reduction is presented in the inset.



# Figure 18. Effect of different concentrations of mammalian cytochrome c on nitrate reductase activity.

Experimental details as given in Figure 17 except that reduced mammalian cytochrome c was used in place of bacterial cytochrome  $c_{\rm 554}.$ 



Figure 18

 $\mu M$  mammalian cytochrome <u>c</u>

### Figure 19. Paper chromatography of flavin isolated from

### nitrate reductase.

Flavin, from solubilized nitrate reductase (Fraction III, Table 20), was separated by DEAE-column chromatography as described in Section 2.5.7. The isolated flavin along with standards of FAD and FMN was subjected to paper chromatography according to the procedure given in Section 2.6.5.

A	:	Standard FAD
В	:	Standard FMN
С	:	Flavin from nitrate reductase.

## Figure 20. <u>High-voltage electrophoresis of the isolated</u> flavin.

Flavin, separated from solubilized nitrate reductase (Section 2.5.7), was incubated at 30° with either snake venom pyrophosphatase for 30 min or with yeast pyrophosphatase for 30 and 60 min. The treated samples of the flavin, as well as an untreated one along with the standards of FMN and FAD, were spotted on to 3 MM Whatman paper. Their relative mobilities, after a high-voltage electrophoresis at 1500 volts for 1 hr, were determined (Section 2.6.6). The flavins were detected under U.V. light.

А	:	Standard FMN
В	:	Standard FAD
С	:	Flavin isolated from nitrate reductase
D		Flavin, from nitrate reductase,
		incubated with snake venom pyrophosphatase for 30 min
Е	:	Flavin, from nitrate reductase, incubated
		with yeast pyrophosphatase for 30 min
$\mathbf{F}$		As for E except the incubation time was
		60 min.





Figure 20



The identity of this flavin as FAD was further confirmed by incubating it with either snake venom pyrophosphatase or yeast pyrophosphatase. These treated samples of the flavin, along with the standards of FAD and FMN, were then subjected to a high voltage electrophoresis as described in Section 2.6.6. The results in Figure 20 show that the isolated flavin, treated with snake venom pyrophosphatase had co-migrated with FMN whereas the untreated sample or that incubated up to 60 min with yeast pyrophosphatase migrated the same distance as the standard FAD. Since snake venom pyrophosphatase hydrolyses FAD to FMN, these results provide additional evidence that the flavin isolated from nitrate reductase was FAD.

#### 3.1.11 Regulation of NADH-linked nitrate reductase

## 3.1.11.1 Inhibition by NAD<sup>+</sup> and its reversal by NADH and menadione

NADH-linked nitrate reductase activity was progressively inhibited by increasing concentrations of NAD<sup>+</sup> in the reaction mixture as shown in Figure 21. The effect of NAD<sup>+</sup> was influenced by the concentration of NADH in the reaction mixture. Thus a 50% inhibition by 0.3mM NAD<sup>+</sup> in the presence of 0.5mM NADH was reduced to 20% when the concentration of NADH was raised to 1mM.

The inhibitory effect of NAD<sup>+</sup> on NADH-linked nitrate reductase was quite specific since NADP<sup>+</sup> did not affect the activity over the range 0.1 - 1.0mM.

The data in Figure 22 indicate that the enzyme activity was dependent on the ratio of NAD<sup>+</sup>/NADH rather than on the absolute concentrations of either of the nucleotides. The enzyme activity was not affected when the ratio of NAD<sup>+</sup>/NADH was less than 0.1. Above this

#### Effect of various concentrations of NAD<sup>+</sup> on Figure 21. NADH-linked nitrate reductase activity.

Cell-free extract (S18), prepared as described in Section 2.5.2.1, was dialysed overnight against 0.05 M Tris-HCl buffer (pH 7.5) containing 1 mM Na<sub>2</sub>-EDTA. The assay procedure for determining the NADH-linked nitrate reductase activity was as described in Section 2.5.4.1. The reaction mixture in a final Tris-HCl buffer (pH volume of 2 ml, contained the following: 7.5), 80 µmoles; KNO3, 1 µmole; 0.03 ml of dialysed cell-free extract (S18 containing 0.96 mg protein); NADH, either 1 or 2 µmoles as specified and the various concentrations of NAD+ (0-2.4 µmoles).

The results are expressed as % inhibition with respect to control (without NAD+) which was 125 nmoles NO2 produced/15 min/mg protein.

0.5 mM NADH

 $\circ$ 

1.0 mM NADH

Effect of NAD<sup>+</sup>/NADH ratio on NADH-linked nitrate Figure 22. reductase activity.

The data from Figure 21 are plotted as % activity of the control (without NAD<sup>+</sup>) against NAD<sup>+</sup>/NADH ratio.

> $\cap$ 0.5 mM NADH  $\cap$

1.0 mM NADH



Figure 21

mM NAD<sup>+</sup>

Figure 22



value, the enzyme activity was progressively inhibited and it was virtually inactive when the ratio was 1.2.

As shown in Figure 23, the enzyme activity was not a linear function of the concentrations of NADH. The production of nitrite was markedly stimulated by NADH over the range of 0.10 - 0.75mM. In the presence of NAD<sup>+</sup>, this allosteric effect of NADH was, however, significantly suppressed and the maximum velocity also decreased with increasing concentrations of NAD<sup>+</sup>.

The data of Figure 23 are presented as a double-reciprocal plot in Figure 24. Non-linear curves which were concave upwards were obtained in the presence of varying concentrations of NADH. However, the curves tended to approach linearity with increasing concentrations of NAD<sup>+</sup> in the reaction mixture. These data also show that the inhibition of the enzyme activity by NAD<sup>+</sup> was reversed by increasing concentrations of NADH. This inhibition by NAD<sup>+</sup> was also reversed by increasing concentrations of menadione (0.05 - 1.0mM) as shown in Figure 25.

The inhibition by NAD<sup>+</sup> was specific for NADH-linked nitrate reductase because even at lmM, NAD<sup>+</sup> did not affect the sulphite-linked enzyme activity.

#### 3.1.11.2 Effect of various adenine nucleotides

The effects of 5'-AMP, either alone or in the presence of 0.13mM NAD<sup>+</sup>, on the NADH-linked nitrate reductase activity are illustrated in Figure 26. The enzyme activity was inhibited by 5'-AMP and at 10mM, it diminished the enzyme activity by more than 60%.

Similar concentrations of 5'-AMP did not affect the sulphitelinked nitrate reductase activity. In addition, neither 5'-ADP nor

### Figure 23. Effect of NAD<sup>+</sup> on the NADH-linked nitrate reductase activity at various concentrations of NADH.

Nitrate reductase activity, in the presence of three fixed concentrations of NAD<sup>+</sup>, was determined as described in Figure 21 except that the concentration of NADH in the reaction mixture was varied as indicated in the Figure.

00	Without NAD+
•0	$0.25 \text{ mM NAD}^+$
۵۵	0.5 mM NAD <sup>+</sup>
	1.0 mM NAD <sup>+</sup>



Figure 23

mM NADH

### Figure 24. Double-reciprocal plot of the effect of NAD<sup>+</sup> at varying concentrations of NADH on nitrate reductase activity.

The data of Figure 23 are presented in the form of a double-reciprocal plot.  $\frac{1}{v}$  represents the reciprocal of nmoles NO<sub>2</sub> produced/15 min.





# Figure 25. Reversal by menadione of the inhibition of the NADH-linked nitrate reductase by NAD<sup>+</sup>.

The enzyme activity in the presence of 0.6 µmole NAD<sup>+</sup> was determined as described in Figure 21 except that specified concentrations of menadione were also included in the assay mixture.



 $\phi^{\mu}$ 

Figure 26. Effect of 5'-AMP on nitrate reductase activity.

Nitrate reductase activity in the cell-free extract (S18), dialysed against 0.05 M Tris-HCl buffer (pH 7.5) containing 1 mM Na<sub>2</sub>-EDTA, was determined in presence of specified concentrations of 5'-AMP with either  $SO_3^{2-}$  or NADH as the electron donor as described in Section 2.5.4.1. Wherever indicated, 0.6 µmoles NAD<sup>+</sup> was also included in the reaction mixture.

Α.	$\Delta \Delta$	$SO_3^{2-}$ -linked nitrate reductase
в.	00	NADH-linked nitrate reductase
c.	ØØ	Same as B except that 0.3 mM NAD
		was included in the reaction mixture.




5'-ATP had any effect on either the sulphite- or NADH-linked nitrate reductase activities. Other nucleotide monophosphates viz., 3'-AMP, 5'-UMP, 5'-GMP, 5'-CMP, 5'-TMP and nucleotide triphosphates such as 5'-CTP, 5'-GTP and 5'-UTP, at 5mM each, were also without effect on nitrate reductase activity with either NADH or sulphite as the electron donor. These results thus indicate that NADH-linked nitrate reductase was inhibited specifically by 5'-AMP. The inhibition of NADH-linked nitrate reductase by 5'-AMP was not reversed by either ATP or NADH. However, 0.5mM menadione completely reversed the inhibition by 5'-AMP.

# 3.1.11.3 Effect of adenylate charge on NADH-linked and sulphite-linked nitrate reductase activities

On varying the adenylate charge from 0.2 - 1.0 in the reaction mixture, the enzyme activity with either NADH or sulphite as an electron donor did not change significantly.

## 3.1.11.4 Effects of cations on the reversal of inhibition of NADH-linked nitrate reductase by NAD<sup>+</sup> and 5'-AMP

The effects of  $Mg^{2+}$  and some other cations on the inhibition of NADH-dependent nitrate reductase by NAD<sup>+</sup> and 5'-AMP are presented in Table 22. In presence of 5mM  $Mg^{2+}$ , a 66% inhibition of the enzyme activity by 0.5mM NAD<sup>+</sup> was reduced to 10% and that by 10mM 5'-AMP was lowered from 71 to 57%. Other divalent cations like Ca<sup>2+</sup> and Ba<sup>2+</sup> also reversed the inhibitory effect of NAD<sup>+</sup> by about 30%. The inhibition by 5'-AMP was reversed by 50% with Ba<sup>2+</sup> but not with Ca<sup>2+</sup>. Monovalent cations like Na<sup>+</sup> and K<sup>+</sup> were not effective in reversing the inhibition of NADH-linked nitrate reductase by either NAD<sup>+</sup> or 5'-AMP.

The inhibition of NADH-linked nitrate reductase by either NAD<sup>+</sup> or 5'-AMP was not influenced by metabolites of the tricarboxylic acid

# Table 22. Effects of monovalent and divalent cations on the inhibition by NAD<sup>+</sup> and 5'-AMP of the NADH-linked nitrate reductase activity.

NADH-linked nitrate reductase activity in the dialysed cell-free extract (S18), in the presence of either 0.5 mM NAD<sup>+</sup> or 10 mM 5'-AMP was determined as described in Figure 21. In addition, various cations, as shown in the Table, were included in the reaction mixture. The results are presented as % activity with respect to control (without NAD<sup>+</sup> or 5'-AMP) which was 120 nmoles  $NO_2^-$  produced/15 min/mg protein.

Cation	Concentration (mM)	Without NAD <sup>+</sup> or 5'-AMP	<pre>% Activity With 0.5 mM NAD<sup>+</sup></pre>	With 10 mM 5'-AMP
None	_	100	34	29
MgCl <sub>2</sub>	5	121	101	53
MgS04	5	112	93	51
BaCl 2	5	110	73	89
CaCl <sub>2</sub>	5	123	86	36
NaCl	5	103	34	31
	10	109	36	28
KCl	5	109	31	30
	10	110	33	28
	8			

cycle, e.g.  $\alpha$ -ketoglutarate, oxaloacetate, malate, fumarate and succinate.

#### 3.2 Nitrite reductase

#### 3.2.1 Distribution of sulphide-linked nitrite reductase

The distribution of sulphide-linked nitrite reductase in various cell fractions is shown in Table 23. The supernatant fraction (S100), obtained after centrifuging the crude extract (S10) at 100,000x g for 90 min, had very low enzyme activity. The bulk of the enzyme activity was recovered in the pellet fraction (P100).

#### 3.2.2 Various electron donors

#### 3.2.2.1 Sulphur compounds

Since in Pl00 fraction nitrite was reduced with sulphide, the effectiveness of some other sulphur compounds for nitrite reduction was also examined. Results in Table 24 show that nitrite was not reduced in the presence of either tetrathionate, thiosulphate or glutathione. Sulphite and thiosulphate plus GSH showed less than 5% of the activity observed with sulphide.

#### 3.2.2.2 Other reductants

The results in Table 25 show that NADH was not an effective electron donor for nitrite reduction in the pellet fraction (P100). However, a significant enzyme activity with NADH was observed when either menadione or PMS was included in the reaction mixture. Similarly ascorbate alone was ineffective but in the presence of either 2,6-dichlorophenol indophenol (DCIP), N,N,N',N'-tetramethylparaphenylenediamine (TMPD) or PMS, nitrite reductase activity was markedly increased.

#### Table 23. Distribution of sulphide-linked nitrite reductase.

The various cell fractions (S10, S100 and P100) were prepared as described in Sections 2.5.2.1 and 2.5.2.3. Nitrite reductase activity, with sulphide as an electron donor, was determined in test-tubes covered with subaseals according to the procedure given in Section 2.5.4.2.

Fraction Procedure (nmoles N utilized/15 proteir	O2 min/mg
I Homogenate obtained after 155.6	5
disrupting the washed cells in	
French pressure cell at 20,000	
p.s.i.	
II Supernatant fraction (S10) obtained 147.6	ò
by centrifuging the broken cells at	
10,000x g for 30 min.	
III Supernatant fraction (S100) after 26.0	)
centrifuging II at 100,000x g for	
90 min.	
IV Pellet fraction (Ploc) from III 420 2	
discorrect in 0.05M main MGL huffer	1
(W. R. C)	
(pH /.5).	

### Table 24.Nitrite reductase activity with various sulphur<br/>compounds as electron donors.

Plo0 fraction was used for determining the enzyme activity. The reaction mixture in a final volume of 1.0 ml contained the following, in µmoles: Tris-HCl buffer (pH 7.5), 35; NaNO<sub>2</sub>, 1; sulphur compounds (as electron donors), 2; and enzyme preparation (Pl00 fraction containing 3 mg protein). After a 15 min incubation at  $30^{\circ}$ , the reaction was stopped with 1 ml of 10% (w/v) zinc acetate in 10% (v/v) ethanol except for the incubation mixture containing thiosulphate to which 10% (w/v) mercuric chloride was also added to precipitate thiosulphate.

	Sulphur compound	Specific activity (nmoles NO2 utilized/15 min/mg protein)
×	$s_4 o_6^{2-}$ $s_2 o_3^{2-}$ GSH $s_2 o_3^{2-}$ + GSH $s o_3^{2-}$ $s^{2-}$ $s^{2-}$	0 8 0 22 22 466
¥2		

#### Table 25. Nitrite reductase activity with various electron donors.

Details of the reaction mixture and the procedure for determining the enzyme activity in Pl00 fraction were as described in Section 2.5.4.2. The reaction was started by injecting the electron donor through the subaseal caps fitted on to the test-tubes. MV, BV, FMN, FAD and mammalian cytochrome c were reduced with 7.5 µmoles of freshly prepared solution of Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub> in 1.0% (w/v) sodium bicarbonate.

$\mathbf{D}^{2} = 1$	Creating patients
Final Electron donor concentration (mM)	specific activity (nmoles NO <sub>2</sub> utilized/15 min/mg protein)
NADH 1	88
. 2	88
NADH + PMS 2 + 2	444
NADH + menadione 2 + 2	1074
NADPH 2	0
Na <sub>2</sub> S <sub>2</sub> O <sub>4</sub> 2	0
MVH 0.4	33
BVH 0.4	44
FMNH <sub>2</sub> 0.2	0
FADH <sub>2</sub> 0.2	0
Sodium ascorbate 2	94
Ascorbate + 2,6-dichloro- phenol indophenol (DCIP) 2 + 0.15	311
Ascorbate + TMPD 2 + 2	570
Ascorbate + PMS 2 + 2	583
Reduced cytochrome c 0.15	155

A relatively small activity was observed with reduced mammalian cytochrome c. Other reductants like sodium dithionite, reduced viologen dyes (BVH, MVH) and reduced flavins (FMNH<sub>2</sub>, FADH<sub>2</sub>) were ineffective. The maximum enzyme activity was recorded with NADH plus menadione which was two-fold higher than with sulphide. The rate of nitrite reduction with other electron donating systems such as NADH plus PMS or ascorbate plus either DCIP, TMPD or PMS were almost equivalent to that obtained with sulphide.

Since, menadione promoted nitrite reductase activity with NADH as an electron donor, the effect of various concentrations of this carrier on the enzyme activity was examined. As shown in Figure 27, the enzyme activity increased proportionally up to lmM menadione in the reaction mixture.

#### 3.2.3 Optimum conditions for enzyme activity

#### 3.2.3.1 Effect of incubation time

The time course for nitrite reductase activity in the P100 fraction is illustrated in Figure 28. Utilization of nitrite was linear up to 15 min. In all the subsequent experiments an incubation time of 15 min was used.

#### 3.2.3.2 Effect of enzyme concentration

The effect of varying enzyme concentrations (P100 fraction) on the rate of nitrite reduction at two concentrations of nitrite and sulphide is illustrated in Figure 29. In the presence of sulphide and nitrite, each at 2mM, the enzyme activity was linear up to 3.5 mg protein. However, with 5mM each of sulphide and nitrite, the enzyme activity was linear up to 10 mg protein.

# Figure 27. Effect of various concentrations of menadione on nitrite reductase activity with NADH as an electron donor.

The reaction mixture was as described in Section 2.5.4.2 except that in addition it contained menadione at the concentrations given in the Figure. The reaction was started by injecting l µmole NADH. After 15 min incubation, the reaction was terminated by adding l ml of cold 10% (w/v) zinc acetate in 10% (v/v) ethanol and the residual  $NO_2^-$  was determined, in a suitable aliquot of the supernatant, according to the method given in Section 2.7.2.



Figure 27

### Figure 28. Effect of incubation time on the sulphide-linked nitrite reductase activity.

The assay was carried out in a Warburg flask fitted with a stopcock as described in Section 2.5.4.2. Reaction mixture in a final volume of 2.5 ml contained the following, in µmoles: Tris-HCl buffer (pH 7.5), 87.5; NaNO<sub>2</sub>, 2.5 and 0.25 ml enzyme preparation (Pl00 fraction containing 8.3 mg protein). The flask was evacuated and flushed with N<sub>2</sub> gas. After equilibrating at  $30^{\circ}$  for 5 min, 5.0 µmoles of freshly prepared solution of sulphide was injected through the subaseal fitted on to the side arm of the Warburg flask. At the times indicated, 100 µl aliquots were withdrawn with an airtight microsyringe and mixed immediately with 100 µl of cold 10% (w/v) zinc acetate solution in 10% (v/v) ethanol. After centrifuging, the residual NO<sub>2</sub> was determined in 100 µl of the supernatant as described in Section 2.7.2.





### Figure 29. Effect of varying protein concentrations on the sulphide-linked nitrite reductase activity.

The enzyme activity, with  $S^{2-}$  as an electron donor, was determined as described in Section 2.5.4.2 except that various amounts of Pl00 fraction (containing 30 mg protein/ml) were used and the concentrations of  $NO_2^-$  and  $S^{2-}$  in the reaction mixtures were:

0	0	2	µmoles	each	of	s <sup>2-</sup>	and	N02
•	0	5	µmoles	each	of	s <sup>2-</sup>	and	NO2



Figure 29

#### 3.2.3.3 Effect of pH

Nitrite reductase activity showed a broad pH optimum over the range of 6.0 - 7.0 (Figure 30).

#### 3.2.3.4 Heat stability

The effect of heat treatment on the sulphide-linked nitrite reductase is shown in Figure 31. Incubation of the enzyme (Pl00 fraction) at  $70^{\circ}$  for 30 min resulted in a 70% loss of activity while at  $50^{\circ}$  and  $40^{\circ}$ , the enzyme activity was diminished by 55 and 20% respectively.

### 3.2.4 <u> $K_m$ values for sulphide and nitrite</u>

At a fixed concentration of lmM sodium nitrite, the enzyme activity increased proportionally with increasing concentrations of sulphide up to 2mM. However, sulphide at concentrations exceeding 3mM inhibited the enzyme activity (Figure 32). At lMm nitrite, the K<sub>m</sub> value for sulphide was 1.25mM.

In the presence of 2mM sulphide, nitrite reductase activity was linear up to 1mM nitrite. The apparent  $K_{\rm m}$  value for nitrite, at a fixed concentration of 2mM sulphide, was 1.6mM (Figure 33).

#### 3.2.5 Characterization of the electron-transport chain involved in nitrite reduction

#### 3.2.5.1 Effects of inhibitors

The effects of various inhibitors on nitrite reductase activity in the pellet fraction (P100) were studied to identify the components of the respiratory chain involved in the reduction of nitrite. In addition to sulphide as a donor, the effects of various

### Figure 30. Effect of pH on the sulphide-linked nitrite reductase.

The enzyme activity, in PloO fraction, was determined as described in Section 2.5.4.2 except that 0.05 M of either phosphate, Tris-HCl or Tris-maleate buffer of indicated pH was used. The pH values given in the Figure correspond to the final pH of the reaction mixture.

00	Phosphate
••	Tris-maleate
ΔΔ	Tris-HCl



e.

Figure 30

### Figure 31. Thermal stability of the sulphide-linked nitrite reductase.

The enzyme preparation (P100 fraction containing 1.75 mg protein) and 0.7 ml of 0.05 M Tris-HCl buffer (pH 7.5) was pre-incubated at the specified temperatures. At the times indicated, nitrite reductase activity was determined as described in Section 2.5.4.2.

Enzyme activity is expressed as % of the original activity (without heat treatment) which was 503 nmoles  $NO_2^-$  utilized/15 min/mg protein.







Nitrite reductase activity was assayed, in the Pl00 fraction, as described in Section 2.5.4.2 except that the concentration of sulphide was varied over the range 0.2-4  $\mu$ moles.  $\frac{1}{V}$  represents reciprocal of nmoles NO<sub>2</sub> utilized/ 15 min.

### Figure 32. A Lineweaver - Burk's plot of the effect of varying sulphide concentrations on nitrite reductase activity.



Figure 32

### Figure 33. Double-reciprocal plot of the effect of varying nitrite concentrations on enzyme activity.

Nitrite reductase activity was determined as described in Section 2.5.4.2. Plo0 fraction (1.75 mg protein) in 0.05 M Tris-HCl buffer (pH 7.5) was pre-incubated with the indicated concentrations of  $NO_2^-$  at 30° for 5 min before starting the reaction by injecting 2 µmoles sulphide solution.  $\frac{1}{v}$ represents reciprocal of nmoles  $NO_2^-$  utilized/15 min.



inhibitors on the enzyme activity with NADH plus menadione and ascorbate plus PMS were also examined.

Effect of thiol binding reagents. As shown in Table 26, sulphydryl reagents markedly inhibited the sulphide-linked nitrite reductase activity. The inhibition by N-ethylmaleimide (NEM) was reversed by GSH. The enzyme activity with either NADH plus menadione or ascorbate plus PMS as the reductant, was not affected by either iodoacetamide or NEM, but p-CMB at lmM inhibited both the activities by 37 and 48% respectively.

Effects of inhibitors of electron transport chain. Rotenone, atabrine and amytal did not effect nitrite reductase activity with either of the three electron donating systems. Antimycin A at lmM lowered the sulphide-linked nitrite reductase activity by 39%, and HQNO (0.05mM) inhibited by 40% only the enzyme activity with NADH plus menadione (Table 27). The sulphide-linked nitrite reductase activity was markedly inhibited by dicumarol (50% at 0.7mM) and this effect was partially reversed by 0.5mM coenzyme  $Q_{10}$  and to a lesser extent by menadione.

Effects of arsenite, azide and cyanide. Arsenite and cyanide, each at lmM, diminished the sulphide-linked nitrite reductase activity by 100 and 82% respectively (Table 28). Cyanide, at lmM, also inhibited the NADH plus menadione dependent enzyme activity by 60%. The enzyme activity, with ascorbate plus PMS as the donor system, was not inhibited by any of these inhibitors.

Effects of metal chelators. The effects of some metal-binding reagents on the enzyme activity are given in Table 29. Bathocuproin,

### Table 26. Effects of sulphydryl group inhibitors on nitrite reductase activity.

The effects of sulphydryl group inhibitors on nitrite reductase activity in Pl00 fraction was studied with three electron donating systems (sulphide, NADH plus menadione and ascorbate plus PMS) as described in Section 2.5.4.2. Pl00 fraction (1.75 mg protein) in 0.025 M Tris-HCl buffer (pH 7.5) was pre-incubated for 10 min at  $30^{\circ}$ with the inhibitor. After adding NO<sub>2</sub>, the reaction was started with the appropriate electron donor system.

Results are expressed as % inhibition of the enzyme activity. The specific activities of 444, 488 and 527 nmoles  $NO_2^-$  utilized/15 min/mg protein were recorded for the control samples (without inhibitor) with  $s^{2-}$ , NADH plus menadione and ascorbate plus PMS as the electron donors respectively.

Addition to the reaction mixture	Final concentration (MM)	s El Sulphide	inhibition lectron donor NADH + menadione	Ascorbate + PMS
p-Chloromercuribenzoate	0.1	40	0	6
	0.2	57	17	17
	1.0	75	37	48
p-Chloromercuribenzoate + GSH	0.2 + 0.25	73 ·	20	20
N-Ethylmaleimide (NEM)	1.0	31	0	0
¥ 	2.0	55	0	0
	5.0	90	0	0
NEM + GSH	2.0 + 2.0	11	0	0
Iodoacetamide	0.5	75	0	0
	5.0	95	0	0
Iodoacetamide + GSH	0.5 + 2.0	77	0	0

### Table 27. Effects of inhibitors of the electron transport chain on nitrite reductase activity.

The experimental details are as given in Table 26. Plo0 fraction (1.75 mg protein) in 0.025 M Tris-HCl buffer (pH 7.5) was pre-incubated with various inhibitors for 10 min at  $30^{\circ}$  before adding the other components of the reaction mixture.

The specific activities of the controls (without inhibitor) were 472, 427 and 526 nmoles  $NO_2$  utilized/15 min/mg protein with  $s^{2-}$ , NADH plus menadione and ascorbate plus PMS as the electron donors, respectively.

Addition to the reaction mixture	Final concentration	% inhibition Electron donor			
	(ner)	Sulphide	menadione	+ PMS	
Rotenone	0.1	10	3	0	
	1.0	16	17	0	
Atabrine	1.0	21	0	13	
х. Х	5.0	22	20	30	
Amytal	2.0	30	16	0	
	5.0	30	36	0	
Antimycin A	0.5	22	8	0	
ž	1.0	39	11	0	
2-n-Heptyl-4-hydroxy-	0.005	4	30	0	
duruofine-N-oxide (HÖNO)	0.05	8	39	4	
Dicumarol	0.2	35	4	0	
0	0.5	43	8	0	
	0.7	52	20	0	
Dicumarol + menadione	0.7 + 1.0	42	: 🛶 아	_	
Dicumarol + coenzyme $Q_{10}$	0.5 + 0.5	19	-	~	
Thenoyltrifluoroacetone	2.0	12	0	0	

### Table 28. Effects of arsenite, azide and cyanide on nitrite reductase activity.

Details of the experimental procedures used are as described in Table 26. The specific activities of the controls (without inhibitor) were the same as in Table 27.

Addition to the reaction mixture	Final concentration (mM)	E Sulphide	% inhibitio lectron don NADH + menadione	n or Ascorbate + PMS
Sodium arsenite	0.1	67	10	0
	1.0	100	16	0
Sodium azide	0.5	9 27	19 28	0
Potassium cyanide	0.1	66 76	0 28	0 11
1	1.0	82	60	16

### Table 29. Effects of various metal chelators on nitrite reductase activity.

The details of the experimental procedures used are given in Table 26. The specific activities of the control samples (without inhibitor) were 583, 566 and 600 nmoles  $NO_2^-$  utilized/15 min/mg protein for  $s^{2-}$ , NADH plus menadione and ascorbate plus PMS as the electron donors, respectively.

Inhibitor	Final concentration (mM)	% El Sulphide	inhibition ectron donor NADH + menadione	Ascorbate + PMS
Bathocuproin	2 5	34 45	42 52	7
O-Phenanthroline	2 5	44 50	22 32	16 17
2,2'-Bipyridyl	5	43	0	0
8-Hydroxyquinoline	5	36	0	0
Sodium diethyldithio- carbamate	5	0	0	0
Urethane	5	0	0	0

o-phenanthroline, 2,2'-bipyridyl and 8-hydroxyquinoline markedly inhibited the sulphide-linked enzyme while sodium diethyldithiocarbamate (DIECA) and urethane, each at 5mM, had no effect. The enzyme activity with NADH plus menadione was inhibited by bathocuproin and -phen nthroline each at 5mM by 52 and 32% respectively. The enzyme activity with ascorbate plus PMS as an electron donor was not affected by any of these metal chelators.

#### 3.2.5.2 Spectral properties

The sulphide-reduced versus oxidized difference spectra of P100 fraction showed absorption bands at 420, 523, 553 and 615 nm and a shoulder around 470-475 nm (Figure 34). The absorption bands at 420, 523 and 553 nm, which are associated with cytochrome c and those at 615 and 470-475 nm due to cytochrome d, were oxidized on addition of nitrite anaerobically to the sulphide-reduced enzyme. These results thus indicate the involvement of these cytochromes in nitrite reduction. Both cytochromes c and d of the sulphide-reduced pellet fraction (P100) were also oxidized in air.

# 3.2.6 Characterization of the electron-transport chain linked to oxygen

### 3.2.6.1 <u>Comparison of oxygen uptake and nitrite reduction with</u> various electron donors

The data in Table 30 show that either sulphide or NADH plus menadione effectively supported both the  $O_2$  uptake and nitrite reduction in the pellet fraction (PlOO). Although a significant uptake of  $O_2$ was also observed with either sulphite or NADH alone, these reductants

### Figure 34. Sulphide-reduced versus oxidized difference spectra of Pl00 fraction.

The Pl00 fraction (0.4 ml containing 12 mg protein) and 0.05M Tris-HCl buffer (pH 7.5), in a total volume of 2.5 ml, were dispensed into both sample (Thunberg) and reference (ordinary) cuvettes. The side arm of the sample cuvette contained 4 µmoles of sulphide. This cuvette was thoroughly evacuated and then flushed with a high purity  $N_2$  gas as described in Section 2.6.3. Sulphide solution was tipped from the side arm into the cuvette and mixed thoroughly. The spectra of the sample was recorded against the reference cuvette. The sample cuvette was then opened, shaken in air for 1 min and its spectra was again recorded. In another experiment, instead of shaking in air, 40 µmoles of NaNO2 was added anaerobically.

A. Reduced versus oxidized difference spectra

B. ----- After either readmitting air or adding 40 µmoles of NO<sub>2</sub> anaerobically to A.





### Table 30. Comparison of oxygen uptake and nitrite reduction with various electron donors.

Oxygen uptake was monitored using an oxygen electrode as described in Section 2.6.1. Nitrite reductase activity was assayed as described in Section 2.5.4.2. For each assay, 0.1 ml Pl00 fraction containing 2.15 mg protein was used.

Electron donor	Final concentration (mM)	0 <sub>2</sub> uptake (nmoles 0 <sub>2</sub> utilized/min/mg protein)	Nitrite reductase (nmoles NO <sup>-</sup> 2 utilized/min/mg protein)
Sulphide	2	22	31
Sulphite	2	31	1
NADH	1	12	6
	2	17	6
NADH + menadione	2 + 1	17	57
	2 + 2	49	85
Sodium ascorbate	2	6	6
Ascorbate + PMS	2 + 2		39

were not effective in reducing nitrite. Ascorbate alone did not promote either O<sub>2</sub> uptake or nitrite reduction, but in the presence of PMS, it stimulated the latter reaction by about six-fold. The effect of ascorbate plus PMS on oxygen uptake could not, however, be determined because PMS is autooxidizable, resulting in a substantial non-enzymic uptake of oxygen.

# 3.2.6.2 Effects of various inhibitors on oxygen uptake with sulphide as reductant

Sulphydryl group inhibitors. The effects of various thiolbinding agents on  $O_2$  uptake with sulphide as an electron donor are shown in Table 31. NEM at 5mM inhibited  $O_2$  uptake by 90% while 1mM p-CMB and 5mM iodoacetamide reduced the activity by 40 and 55% respectively. The inhibition of  $O_2$  uptake by either NEM or p-CMB was partially relieved by GSH.

Inhibitors of electron transport chain. Amytal, atabrine, rotenone, antimycin A and HQNO did not inhibit O<sub>2</sub> uptake. However, dicumarol, at lmM reduced the activity by about 60% (Table 32).

Metal chelators and other inhibitors. The effects of various metal chelating agents on the sulphide-dependent  $O_2$  uptake are presented in Table 33. 2,2'-bipyridyl and 8-hydroxyquinoline, each at 5mM, inhibited the uptake of  $O_2$  by about 50%. Cyanide was strongly inhibitory and at 0.1mM, it reduced the activity by 80%. Diethyldithiocarbamate (DIECA), bathocuproin and urethane did not have any effect whereas azide at 1mM was only slightly inhibitory.

### Table 31.Effects of inhibitors of -SH groups on oxygen uptakewith sulphide as an electron donor.

The details of the assay procedures are the same as given in Section 2.6.1 except that Pl00 fraction (2 mg protein) and 42.5 µmoles of Tris-HCl buffer (pH 7.5) were pre-incubated with each inhibitor for 10 min prior to adding the electron donor. The results are expressed as % inhibition as compared to the control (without inhibitor) which was 20 nmoles  $O_2$  utilized/min/mg protein.

Addition to the reaction mixture	Final concentration (MM)	% inhibition
p-Chloromercuribenzoate	0.1	32
5	1.0	39
p-Chloromercuribenzoate + GSH	1.0 + 1.0	21
N-Ethylmaleimide (NEM)	1.0	37
	2.0	45
	5.0	90
NEM + GSH	2.0 + 2.0	34
20 41		
Iodoacetamide	1.0	30
*)	2.0	43
	5.0	55
Iodoacetamide + GSH	5.0 + 5.0	52

## Table 32.Effects of inhibitors of electron transport chain onsulphide-linked oxygen uptake.

Aliquots of Pl00 fraction were pre-incubated with the inhibitors for 10 min as described in Table 31. The specific activity of the control (without inhibitor) was 19 nmoles  $O_2$  utilized/min/mg protein.

Inhibitor	Final concentration (mM)	% inhibition
Amytal	2 0	22
	5.0	23
14	5.0	27
Atabrine	2.0	11
	5.0	23
Rotenone	0.5	27
	1.0	34
Antimycin A	0.1	16
	0.5	35
HQNO	0.01	25
	0.05	25
Dicumarol	0.5	45
	1.0	58

### Table 33.Effects of metal chelators and other inhibitors onsulphide-linked oxygen uptake.

The details of the experimental procedure are described in Table 31. The specific activity of the control sample, without inhibitor, was 20 nmoles O<sub>2</sub> utilized/min/mg protein.

Inhibitor	Final concentration (mM)	% inhibition
2,2'-Bipyridyl	5.0	48
8-Hydroxyquinoline	5.0	44
Sodium diethyldithiocarbamate	5.0	0
Urethane	5.0	4
Bathocuproin	5.0	0
o-Phenanthroline	5.0	23
Sodium azide	1.0	30
Potassium cyanide	0.05	59
	0.1	80
Sodium arsenite	1.0	0

#### 3.2.7 Purification of sulphide-linked nitrite reductase

#### 3.2.7.1 Solubilization

As mentioned earlier (Table 23), nitrite reductase is associated with membraneous fraction (P100). Attempts to solubilize the enzyme from P100 fraction by treatment with detergents namely, sodium deoxycholate, SDS, Triton X-100 or Tween 80 or with lipases and proteolytic enzymes were ineffective. However, the enzyme from acetone-dried cells was recovered in the supernatant fraction. Thus an acetone treatment of the cells was effective in dissociating the sulphide-linked nitrite reductase from the membranes. As shown in Figure 35, addition of sulphide to the high speed supernatant fraction (S100) of the acetone-dried cells results in the appearance of absorption bands, corresponding to reduced cytochrome c (420, 523 and 553 nm) and cytochrome d (615 and a shoulder at 470-475 nm). In contrast, sulphite did not cause any spectral changes thus indicating that the acetone treatment of the cells did not dissociate the sulphitelinked nitrate reductase from the membranes. However, the addition of sulphite to the S100 fraction, prepared after treating the crude extract of the acetone-dried cells with sodium deoxycholate (0.06 mg/mg protein) for 30 min at  $30^{\circ}$ , showed absorption bands at 424, 524 and 554 nm. This absorption spectra is identical to that observed in Figure 9 for sulphite-linked nitrate reductase.

#### 3.2.7.2 Purification

Enzyme activity at the various stages of purification was assayed by determining the rate of nitrite reduction with sulphide as the reductant. The buffer used during purification was 0.025M Tris-HCl (pH 7.5) and unless stated otherwise, all the procedures were carried out at 4<sup>°</sup>.
# Figure 35. Effect of sodium deoxycholate treatment on reduced versus oxidized spectra of Sl00 fraction prepared from acetone-dried cells.

The acetone-dried cells, disrupted as described in Section 2.5.1.1, were centrifuged at 18,000x g for 45 min. The supernatant fraction thus obtained was further centrifuged at 100,000x g for 90 min either without any treatment or after incubating it with sodium deoxycholate (0.1%, w/w) at  $30^{\circ}$  for 30 min. Reduced versus oxidized difference spectra of the supernatant fractions (S100) were recorded as described in Section 2.6.3.

> $SO_3^{2-}$ -reduced versus oxidized difference spectra of Sl00 fraction  $SO_3^{2-}$ -reduced versus oxidized difference spectra of Sl00 fraction of the deoxycholate-treated sample  $S^{2-}$ -reduced versus oxidized difference

spectra of S100 fraction.



nm

The cell-free extract (S18), prepared from acetone-dried cells as described in Section 2.5.2.2, was further centrifuged at 100,000x g for 90 min. To the supernatant fraction (S100) thus obtained, solid ammonium sulphate was added gradually with continuous stirring to bring it to 40% saturation. The suspension was centrifuged at 20,000x g for 20 min. The supernatant fraction was then brought to 95% saturation by further addition of solid ammonium sulphate. The pellet, obtained after centrifuging at 20,000x g, was dispensed in a minimum volume of 0.025M Tris-HCl buffer (pH 7.5) and dialyzed for 16 hr against three changes of double-distilled water.

A DEAE-32 cellulose column (2.5 x 26 cm), prepared as described in Section 2.6.8.1, was equilibrated with 0.025M Tris-HCl buffer (pH 7.5) until the pH of the effluent was the same as that of the equilibrating buffer. The enzyme preparation, from the preceding step (Fraction III, Table 34), was then loaded onto the column and eluted at a flow rate of 60 ml/hr with 0.025M Tris-HCl buffer (pH 7.5). Nitrite reductase, which was not adsorbed on to the column, was recovered after one column volume in the greenish brown fractions (Figure 36). A reddish brown band retained at the top of the column was eluted with 0.025M Tris-HCl buffer (pH 7.5) containing 0.5M KCl. The spectral characteristics of this fraction are presented in Figure 37. In its oxidized state, it had absorption maxima at 408 and 525 nm and on reduction with sodium dithionite at 416, 524 and 554 nm. These absorption bands indicated that this reddish brown fraction contained a cytochrome of the c type.

The fractions (15-26) from DEAE-cellulose column, which contained most of the nitrite reductase activity, were pooled and concentrated in an Aminco ultrafiltration unit using PM-10 membrane filter under  $N_2$  gas as described in Section 2.6.9.

### Figure 36. Elution profile of nitrite reductase from a DEAE-cellulose column.

A DEAE-cellulose column (DE32; 2.5 x 26 cm) was equilibrated with 0.025 M Tris-HCl buffer (pH 7.5) at  $4^{\circ}$ . Fraction III (Table 34) was loaded on to the column and eluted with the same buffer at a flow rate of 60 ml/hr. Fractions (3 ml) were collected in an automatic fraction collector and nitrite reductase activity was determined in 0.2 ml aliquots of the each fraction as described in Section 2.5.4.2.  $E_{280}$ of a suitably diluted fraction, in a final volume of 2.5 ml, was recorded. The values given for  $E_{280}$  in the Figure have been calculated for the absorbance/ml of each fraction.

<sup>E</sup>280

 $\Delta - - - - - - \Delta$ 

Nitrite reductase activity





Figure 36

### Figure 37. Dithionite-reduced versus oxidized difference spectra of cytochromes $c_{551}$ and $c_{554}$ .

Cytochromes  $c_{551}$  and  $c_{554}$  were isolated from *T. denitrificans* during purification of nitrite reductase as described in Section 4.2.7.2. 200 µl of either cytochrome  $c_{551}$  or  $c_{554}$  and 0.025 M Tris-HCl (pH 7.5), in a final volume of 2.5 ml, were dispensed in to both sample and reference cuvettes. After adding a small amount of Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub> to the sample cuvette and a crystal of K<sub>3</sub>Fe(CN)<sub>6</sub> to the reference cuvette, the difference spectra were recorded.

 $Na_2S_2O_4$ -reduced versus oxidized spectra of cytochrome  $c_{551}$ 

 $Na_2S_2O_4$ -reduced versus oxidized spectra of cytochrome  $c_{554}$ .



Figure **37** 

A Sephadex G-100 column (2.5 x 45 cm), prepared as described in Section 2.6.8.2, was equilibrated with 0.025M Tris-HCl buffer (pH 7.5). The concentrated enzyme from DEAE-cellulose (Fraction IV, Table 34) was applied carefully onto the column and the sample was eluted at a flow rate of 15 ml/hr with the equilibrating buffer and 3 ml fractions were collected using L.K.B. fraction collector. The elution profile of the sample from this column indicated that Fraction IV was resolved into two components (Figure 38). Nitrite reductase was recovered in the fractions immediately after the void volume. The second fraction contained a cytochrome of c type with absorption maxima at 408 and 525 nm in the oxidized form and at 412, 521 and 551 nm after its reduction with sodium dithionite (Figure 37).

This purification procedure, summarized in Table 34, resulted in a twenty-fold increase in specific activity of nitrite reductase compared to the crude extract (S18) and 36% of the original enzyme activity was recovered. The purified enzyme had an  $A_{280}/A_{405}$  nm ratio of 0.75.

#### 3.2.8 Properties of the purified enzyme

#### 3.2.8.1 Molecular weight

Molecular weight of the purified enzyme (Fraction V, Table 34) was determined by gel filtration through a Sephadex G-150 column and using the equation of Squire (1964) as described in Section 2.5.6.1. Assuming that the enzyme was a globular protein, its estimated molecular weight was 120,000 daltons with a standard error of ±5,000.

### Figure 38. Elution profile of nitrite reductase from Sephadex G-100 column.

Fraction IV (Table 34) was loaded onto a Sephadex G-100 column (2.5 x 45 cm), pre-equilibrated with 0.025 M Tris-HC1 buffer (pH 7.5). The sample was eluted with the same buffer at the flow rate of 15 ml/hr and 3 ml fractions were collected. Nitrite reductase activity was determined in 0.2 ml aliquots of each fraction according to the procedure given in Section 2.5.4.2. Haem c content of each fraction was measured at 405 nm.

©------ <sup>E</sup>280

0----0

Nitrite reductase activity



#### Table 34. Purification of nitrite reductase.

Acetone-dried powder of the washed cells was prepared as described in Section 2.5.1. The acetone-dried cells, suspended in 0.025 M Tris-HCl buffer (pH 7.5), were stirred at  $4^{\circ}$  for 3 hr. The suspension was passed through a French pressure cell (Section 2.5.2.2). The broken cells were centrifuged at 18,000x <u>g</u> for 45 min and the supernatant (S18) fraction was used for the further purification of the enzyme.

Nitrite reductase activity, in various fractions obtained, was determined as described in Section 2.5.4.2.

Total activity :  $\mu$ moles  $NO_2^-$  utilized/15 min Specific activity :  $\mu$ moles  $NO_2^-$  utilized/15 min/mg protein

#### Table 34. Purification of nitrite reductase.

Fraction	Procedure	Total activity	Total protein (mg)	Specific activity	Fold Purification	Recovery (%)
I	S18, supernatant fraction left after centrifuging broken cells at 18,000x <u>g</u> for 45 min	507.5	9207	0.05	l	100
II	S100, supernatant fraction obtained after centrifuging the S18 fraction at 100,000x g for 90 min	384	4800	0.08	1.5	75
III	Fraction II precipitated with ammonium sulphate between 40-95% saturation and dialyzed against double distilled water overnight	184.8	1680	0.11	1.9	36
IV	Fraction III eluted with 0.025 M Tris-HCl buffer (pH 7.5) from a DEAE- cellulose column and concentrated by ultra- filtration	173.8	245	0.71	12.8	34
v	Fraction IV eluted with 0.025 M Tris-HCl buffer (pH 7.5) from Sephadex G-100 column	132.3	118	1.12	20.4	26

#### 3.2.8.2 Copper and iron contents

Copper was determined by atomic absorption spectrometry as described in Section 2.7.3. With this technique up to 0.1 µg of copper can be determined. However, copper was not detected in the 12.5 mg protein used for the assay. Results in Table 35 show that the purified enzyme contained 29 µmoles of iron per mg protein.

### 3.2.8.3 Relative concentrations of haems c and d in purified

#### nitrite reductase

It was shown earlier that the reducing equivalents from sulphide were transferred to nitrite via cytochromes c and d of purified nitrite reductase. The relative concentrations of haems cand d in the purified enzyme was, therefore, determined from the respective molecular extinction coefficients of their pyridine haemochromogen derivatives. The results of three determinations carried out with different enzyme preparations indicate that these two haems were present in an equimolar amount (Table 36).

#### 3.2.8.4 Sequence of reduction and oxidation of cytochromes c and d

As mentioned in the preceding Section (3.2.8.3), both haems c and d were present in the purified nitrite reductase in the ratio of l : l. An attempt was, therefore, made to establish the sequence of electron transfer from sulphide to nitrite *via* nitrite reductase. Although both the cytochromes were reduced after adding sulphide, the rate of reduction of cytochrome c during the first few minutes was much faster than that of cytochrome d. Thus l min after adding sulphide sulphide to the enzyme, 50% of cytochrome c was reduced as compared

## Table 35.Iron content of purified nitrite reductase (cytochrome cdcomplex), dissociated cytochrome c and haem d.

Purified nitrite reductase (1.0 ml containing 12.5 mg protein of Fraction V, Table 34), dissociated cytochrome c (1.5 ml containing 9.2 mg protein) and haem d (1.5 ml containing 3.15 mg protein) were dried separately in acid-washed pyrex test-tubes. The dried samples were digested in an acid mixture of  $HNO_3$ ,  $HCIO_4$  and  $H_2SO_4$  (in ratio of 40 : 4 : 1) and their iron contents were estimated by Varian atomic absorption spectrophotometer as described in Section 2.7.3.

Sample	Concentration (µmoles/gm protein)
Purified nitrite reductase	29.0
Dissociated cytochrome c	16.8
Haem d	16.8

### Table 36.Relative concentrations of haem c and haem d in the<br/>purified nitrite reductase.

The concentrations of haem c and haem d of the purified nitrite reductase (Fraction V, Table 34) were determined spectrophotometrically in an alkaline pyridine solution as described in Section 2.7.4. The extinction coefficient values of 30.2 and 19.6  $\text{mM}^{-1}$  cm<sup>-1</sup> were used for calculating the amounts of haem c and haem drespectively.

Preparation No.	Нает <i>с</i> (µМ)	Haem <i>d</i> (µM)	<u>Haem d</u> ratio Haem c
l	104	110	1.05
2	92	96	1.04
3	100	105	1.05

to 25% of cytochrome d (Figure 39). After 13 min both the cytochromes were fully reduced. The relatively faster rate of cytochrome creduction during early stages indicates that the reducing equivalents were first transferred to cytochrome c and then to cytochrome d.

The rates of oxidation of the sulphide-reduced cytochromes c and d with nitrite were also examined. It is clear from Figure 40 that on adding nitrite, cytochrome d was oxidized at a much faster rate than cytochrome c. Thus, after 15 min more than 75% of cytochrome d was oxidized compared with 34% of cytochrome c. The sequence of the oxidation of cytochromes c and d indicates that cytochrome d functions as the immediate electron donor for nitrite reduction.

#### 3.2.8.5 Difference in redox-potential of cytochromes c and d

The experimental values for the relative concentrations of reduced and oxidized cytochromes c and d given in Table 37, for calculating the values of equilibrium constant, were derived from the data presented in Figure 39. On reduction of the enzyme with sulphide, an equilibrium was established between the haem c and d moieties. The difference in the redox-potential of the two cytochromes was calculated by substituting the values of the equilibrium constant in the equation given in Section 2.6.4. The redox-potential of cytochrome d was 22 mV more positive than that of cytochrome c.

#### 3.2.8.6 Electron donors

Nitrite reductase activity of the purified enzyme was examined in the presence of a variety of electron donor systems. The results in Table 38 indicate that as in PlOO fraction (Table 25), sulphide as well as ascorbate plus PMS were effective electron donors.

### Figure 39. Rate of reduction of cytochromes c and d of cd complex of purified nitrite reductase.

Purified nitrite reductase (200  $\mu$ l of Fraction V, Table 34) and 0.025 M Tris-HCl buffer (pH 7.5), in a total volume of 2.5 ml, were taken in a Thunberg cuvette which contained 5  $\mu$ moles of sulphide solution in the side arm. The cuvette was vigorously evacuated and then filled with pure N<sub>2</sub> gas. This procedure was repeated twice (Section 2.6.3). Sulphide solution, from the side arm, was then mixed with the contents of the cuvette and change in absorbance at 554 nm (for cytochrome c) and at 610 nm (for cytochrome d) was recorded at various time intervals. The % reduction of cytochromes c and d respectively, was calculated on the basis of maximum reduction of the cytochromes obtained after prolonged incubation period.



Figure 40.

#### Rate of oxidation of sulphide-reduced cytochromes

## c and d of cd complex of purified nitrite reductase with nitrite.

Experimental procedure was as described in Figure 39 except that 5 µmoles of sulphide solution was added to the Thunberg cuvette and the side arm contained 10 µmoles of nitrite solution. The Thunberg cuvette was made anaerobic as described in Section 2.6.3. Optical density at 554 and 610 nm for cytochromes c and drespectively, of the reduced enzyme was recorded until maximum absorbance was obtained. Nitrite was then tipped from the side arm and mixed thoroughly with the contents of the cuvette. Changes in absorbance at 554 and 610 nm were recorded at various time intervals. % oxidation of each cytochrome was calculated on the basis of the respective amounts of the fully reduced cytochromes c and d present before adding nitrite.

> ΔA 554 nm

<sup>∆A</sup>610 nm





Figure **40** 



### Table 37.Difference in redox-potentials of cytochromes c and d of the cdcomplex of nitrite reductase.

The difference in the redox-potentials of cytochromes c and d of purified nitrite reductase (Fraction, Table 34) was calculated, as described in Section 2.6.4, from the data given in Figure 39.

Time (min)	Cytoch % Oxidized (c <sup>3+</sup> )	rome <i>c</i> % Reduced (c <sup>2+</sup> )	Cytochr % Oxidized (d <sup>3+</sup> )	come d % Reduced (d <sup>2+</sup> )	Equilibrium constant (K) = $\frac{(c^{3+})(d^{2+})}{(c^{2+})(d^{3+})}$	Em(cyt.d) - Em(cyt.c) = 0.059 logxK (mV)
						3
1.0	48.06	51.94	75.00	25.00	3.24	+30
2.0	32.47	67.53	45.47	54.52	1.72	+14
				ĸ		
3.0	20.80	79.20	29.55	70.45	1.59	+12
5.0	9.10	90.90	17.30	82.70	2.08	+19
9.0	1.30	98.70	4.50	94.45	3.56	+32
15.0	0.00	100.00	0.00	100.00	-	

Mean = +22

### Table 38.Effect of various electron donors on nitrite reductaseactivity in the purified enzyme preparation.

Nitrite reductase activity in the purified preparation (Fraction V, Table 34), with various electron-donating systems, was assayed in Warburg flasks fitted with stopcocks as described in Section 2.5.4.2.

Electron donor	Final concentration (mM)	Specific activity (nmoles NO <sub>2</sub> utilized/15 min/mg protein)
Sulphide	2	660
NADH + menadione	2 + 2	0
Ascorbate + PMS	2 + 2	1040
-		20 C

Nitrite, however, was not reduced with NADH plus menadione as the reductant.

The effectiveness of various redox-dyes, with potentials ranging from -0.08 to +0.217 V, as electron carriers for nitrite reduction in the presence of ascorbate was examined (Table 39). Maximum enzyme activity was obtained with PMS and TMPD, while methylene blue, thionine, or 2,6-dichlorophenol indophenol (DCIP) gave 75% of the activity obtained with TMPD. The inclusion of cytochrome c, isolated from *T. denitrificans*, along with DCIP did not increase the enzyme activity markedly. Thus, nitrite reductase activity was maximal with dyes with redox-potentials within the range of +0.08 to +0.197 V.

## 3.2.8.7 Effects of mammalian and bacterial cytochromes c on nitrite reductase activity

Results in Table 40 indicate that the inclusion of 2.2  $\mu$ M cytochrome  $c_{551}$  (see Section 3.2.7.2) in the reaction mixture enhanced the sulphide-linked nitrite reductase activity by about two-fold. The presence of either cytochrome  $c_{554}$ , also isolated from *T. denitrificans* (Section 3.2.7.2, Figure 37), or mammalian cytochrome *c* had no effect on the enzyme activity. Thus, sulphide-linked nitrite reductase activity was stimulated only by the bacterial cytochrome  $c_{551}$ .

The effects of various concentrations of cytochrome  $c_{551}$  on the enzyme activity were examined. The maximum stimulation was observed with 2  $\mu$ M cytochrome  $c_{551}$  (Figure 41) and higher amounts, up to 10  $\mu$ M, did not enhance the enzyme activity further.

Reduced cytochrome  $c_{551}$  served as an effective electron donor for nitrite reduction by the purified enzyme (Figure 42). The

# Table 39. Effects of various electron mediators (dyes) on the activity of nitrite reductase in the purified enzyme (Fraction V, Table 34).

The details of assay procedure are as described in Section 2.5.4.2 except that instead of sulphide, 1 mM of the specified electron mediators were used. Methylene blue was, however, used at a final concentration of 0.5 mM. These mediators were reduced by injecting 2  $\mu$ moles of neutralized ascorbic acid into the Warburg flasks. After a 15 min incubation at 30°, residual nitrite was determined colorimetrically as described in Section 2.7.2.

Electron mediator	E' of mediator (V)	Specific activity (nmoles NO2 utilized/15 min/mg protein)
Triphonyltotropolium	0.00	
111phenyitetrazorium	-0.08	130
Methylene blue	+0.011	640
Thionine	+0.065	640
Phenazine methosulphate (PMS)	+0.08	820
N,N,N',N'-tetramethylpara- phenylenediamine (TMPD)	+0.197	840
TMPD + cytochrome $c^*$		860
2,6-dichlorophenol indophenol (DC	CIP) +0.217	696
DCIP + cytochrome $c_{551}^*$		756

cytochrome c<sub>551</sub> was separated from *T. denitrificans* (Section 4.2.7.2, Figure 37).

# Table 40. Effect of cytochrome c from various sources on nitrite reductase activity in purified enzyme preparation (Fraction V, Table 34).

The composition of the reaction mixture was as given in Section 2.5.4.2 except that cytochromes  $c_{551}$  and  $c_{554}$ , both isolated from *T. denitrificans* (Section 4.2.7.2, Figure 37), and mammalian cytochrome *c*, at indicated concentrations, were included in the reaction mixtures. The reaction was started by injecting sulphide solution through the subaseals-fitted side arms of the Warburg flasks.

Addition to the reaction mixture	Final concentration (mM)	Specific activity (nmoles NO <sub>2</sub> utilized/15 min/mg proteins)
λ.		
None	-	569
Cytochrome c 551	1.1	1015
	2.2	1123
Cytochrome c 554	1.0	569
	2.0	560
Mammalian cytochrome c	1.0	550
	2.0	573

### Figure 41. Effect of various concentrations of cytochrome $\frac{c_{551}}{(\text{Fraction V, Table 34)}}$ .

The enzyme activity was assayed as described in Section 2.5.4.2 except that cytochrome  $c_{551}$ , at the concentrations specified in the Figure, was included in the reaction mixture.

### Figure 42. Reduced cytochrome c<sub>551</sub> and mammalian cytochrome c as electron donors for the activity of purified reductase.

Concentrated solutions of cytochrome  $c_{551}$  and mammalian cytochrome c respectively were reduced with an excess of Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub>. Each of the reduced cytochromes was then dialysed overnight against 0.025 M Tris-HCl buffer (pH 7.5). The enzyme activity was assayed in Warburg flasks fitted with stopcocks. The reaction mixture in a final volume of 1 ml contained the following, in µmoles: Tris-HCl buffer (pH 7.5), 35; NaNO<sub>2</sub>, 0.2 and 50 µl of purified nitrite reductase (Fraction V, Table 34) containing 0.625 mg protein. The flasks were repeatedly evacuated and filled with N<sub>2</sub> gas. The reaction was started by injecting different amounts of either reduced cytochrome  $c_{551}$  or reduced mammalian cytochrome c, through subaseals fitted on to the side arms of the Warburg flasks.

After 15 min incubation, 1 ml of 10% (w/v) zinc acetate in 10% (v/v) ethanol was added to stop the reaction. 0.5 ml aliquots were used for estimating residual nitrite according to the procedure given in Section 2.7.2.

0	0	Reduced	l

Reduced cytochrome  $c_{551}$  as the reductant

Reduced mammalian cytochrome c as the reductant.



Figure 42



stoichiometry of approximately 1 : 1 was obtained for the amount of nitrite reduced to the amount of cytochrome  $c_{551}$  added to the reaction mixture. In contrast, reduced mammalian cytochrome c was an ineffective electron donor. Only 25 nmoles of nitrite were reduced in the presence of 130 nmoles of reduced mammalian cytochrome c.

#### 3.2.9 Cytochrome oxidase activity of the purified nitrite reductase

Cytochrome oxidase activity was determined in the various fractions obtained during the purification of nitrite reductase (Table 34). Cytochrome oxidase activity was measured by following the rate of  $O_2$  uptake with sulphide as an electron donor as described in Section 2.5.4.3. Results in Table 41 show that the specific activity of cytochrome oxidase increased progressively during the successive stages of purification of nitrite reductase.

# 3.2.9.1 Comparison of nitrite reductase and cytochrome oxidase activities

Some of the properties of nitrite reductase and cytochrome oxidase activities in the purified enzyme were compared. Both the activities, with sulphide as an electron donor, were stimulated twoto three-fold by adding 2  $\mu$ M cytochrome  $c_{551}$  to the reaction mixture. Reduced cytochrome  $c_{551}$  was also an effective electron donor for  $O_2$ uptake and nitrite reduction. For each mole of reduced cytochrome  $c_{551}$  added, one mole of either nitrite or  $O_2$  was reduced. Glutathione, at 2mM, inhibited the cytochrome oxidase activity by about 40% while nitrite reductase was not affected (Table 42).

## Table 41.Cytochrome oxidase activity in the various fractionsobtained during the purification of nitrite reductase.

The fractions were prepared as described in Table 34 (Section 4.2.7.2) and the cytochrome oxidase activity was monitored in an  $O_2$  electrode cell using sulphide as the electron donor (Section 2.5.4.3).

Fraction	Procedure	Total activity (µmoles O <sub>2</sub> utilized/min)	Specific activity (µmoles O <sub>2</sub> utilized/min/mg protein)
I	Crude extract (S18) left after centrifuging the disrupted acetone-dried cells at 18,000x <u>g</u> for 45 min.	74.1	8.0
II	Supernatant fraction (S100) obtained after centrifuging S18 at 100,000x g for 90 min.	37.6	7.89
III	Fraction II precipitated with ammonium sulphate (40-95% saturation) and dialyzed against 0.025 M Tris-HCl buffer (pH 7.5) overnight.	19.2	11.4
IV	Fraction III loaded on to a DEAE-cellulose column and eluted with 0.025 M Tris-HCl buffer (pH 7.5). The active fractions concentrated by ultrafiltration.	6.5	26.4

### Table 42.Comparison of nitrite reductase and cytochrome oxidase activitiesin the purified nitrite reductase.

A. Nitrite reductase and cytochrome oxidase activities in the purified nitrite reductase enzyme (Fraction V, Table 34) were determined, with sulphide as a donor, as described in Section 2.5.4.2 and 2.5.4.3 respectively. Wherever indicated either cytochrome  $c_{551}$  (Section 4.2.7.2, Figure 37) or GSH were included in the reaction mixture.

B. Reduced cytochrome  $c_{551}$ , in place of sulphide, was used as the electron donor.

	Electron donor	Addition to the basal reaction mixture	Nitrite reductase (nmoles NO <sub>2</sub> utilized/min/mg protein)	Cytochrome oxidase (nmoles O <sub>2</sub> utilized/min/mg protein)
		-		
Α.	Sulphide (2 µmoles)	-	35	18
•		Cytochrome c <sub>551</sub> (2 µM)	107	48
		GSH (2 mM)	35	11
				5
В.	Reduced cytochrome			
	c <sub>551</sub>			
	20 nmoles	-	21	20
	53 nmoles	÷	59	53
				-

#### 3.2.10 · Spectral properties of purified nitrite reductase

#### 3.2.10.1 Absorption spectra

The absorption spectra of purified nitrite reductase (Fraction V, Table 34) showed absorption bands at 405, 525 and 645 nm and on reduction with sodium dithionite at 418, 468, 523, 549-554, 615 nm and a broad plateau around 665-700 nm (Figure 43). The absorption bands at 418, 523, 549-554 nm are associated with cytochrome c while those at 468, 615 and a plateau between 665-700 nm were due to cytochrome d. The ratio of  $\alpha/\beta$  peaks of reduced cytochrome cwas 1.0. The ratio of the absorbance bands at 468 and 615 nm of reduced cytochrome d was 0.7. On reduction with sodium dithionite, the absorption of the purified enzyme at 645 nm disappeared.

Addition of sulphide to the purified nitrite reductase, resulted in the following changes in the absorption spectra: absorption bands of reduced cytochrome c appeared within a minute of adding the reductant and their peak heights continued to increase with time. The reduction of cytochrome d was relatively slower than that of cytochrome c. Both the cytochromes were fully reduced after 15 min. The sulphide-reduced enzyme showed absorption maxima at 418, 462, 523, 549-554, 610 and 664-667 nm (Figure 44). The plateau at 665-700 nm, observed in the sodium dithionite-reduced enzyme, was replaced by a broad band between 664-667 nm. Moreover, instead of a peak at 468 nm obtained with sodium dithionite, there was a sharp peak at 462 nm in the sulphidereduced enzyme. The absorption ratio of 462 nm and 610 nm was 2.1 which was three times greater than in the dithionite-reduced enzyme. The absorption characteristics of cytochrome c with either of these reductants were, however, the same. On reduction of the enzyme with

### Figure 43. Absolute and difference spectra of purified nitrite reductase.

Absorption spectra of purified nitrite reductase (Fraction V, Table 34) were recorded as described in Section 2.6.3.

Absolute spectra

the second second second second

ملو الند الليد البد

Difference spectra, Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub> reduced versus oxidized.



### Figure 44. <u>Time course of reduction of the cytochrome *cd*</u> complex of purified nitrite reductase with sulphide.

The experiment was done in Thunberg cuvettes. 150  $\mu$ l of purified enzyme (Fraction V, Table 34) and 0.025 M Tris-HCl buffer (pH 7.5), in a final volume of 2.5 ml, were dispensed into each of the reference and sample cuvettes. Freshly prepared solution of sulphide (15  $\mu$ moles) was placed in the side arm of the sample cuvette which was vigorously evacuated and then filled with N<sub>2</sub> gas as described in Section 2.6.3. The reaction was started by adding sulphide solution to the cuvette from the side arm and the spectra was recorded against the reference cuvette after following time intervals:

	l min
	5 min
2 ••••••••	15 min

### Figure 45. Time course of reduction of cytochrome *cd* complex of purified nitrite reductase with ascorbate *plus* PMS.

Experimental details as described in Figure 44 except that instead of sulphide, 5 µmoles of neutralized solution of ascorbate and 0.5 µmoles of PMS were used. The same amount of PMS was also included in the reference cuvette. The spectra were recorded after the following time intervals:

 1 min	
 5 min	
 15 min 🤷	





 $\langle \hat{\mathbf{x}} \rangle$ 

nm

Figure 44

Figure 45

sulphide as well as with dithionite, the enzyme turned dark green from its original reddish brown colour. Spectral changes similar to those with sulphide were recorded when ascorbate plus PMS was used as the reductant (Figure 45).

The results in Figure 46 show that on adding nitrite anaerobically to the sulphide-reduced enzyme, both cytochromes c and dwere oxidized; cytochrome c, however, remained in a partially reduced state even after 15 min of adding nitrite. The absorption bands at 462 and 664-667 nm disappeared immediately on adding nitrite, followed by the one at 610 nm. The peaks associated with cytochrome c were, however, oxidized at a relatively slower rate. The oxidation of cytochrome components of the enzyme was accompanied by an appearance of a peak at 573 nm. Oxidation of either the sulphide- or the dithionite-reduced enzyme with nitrite, resulted in restoration of the original reddish brown colour of the purified enzyme. The cytochromes c and d were not oxidized when nitrate, instead of nitrite, was added to the sulphide-reduced enzyme.

Absorption spectrum of cytochrome d of the sulphide-reduced enzyme was markedly influenced by the pH of the reaction mixture whereas that of cytochrome c remained unaffected. The absorption peak of cytochrome d at 645 nm of the purified nitrite reductase was also not influenced over the pH range of 5.5 - 8.0. As shown in Table 43, the ratio of the absorbance at 610/645 nm increased from 0.93 to 1.8 on lowering the pH of the reaction mixture from 8.0 to 5.5. The  $\gamma$ -band of cytochrome d at 462 nm was less prominent at the acid pH. Thus, the ratio of 462/610 nm declined from 1.7 at pH 8.0 to 0.42 at pH 5.5.

#### Figure 46. Oxidation of the sulphide-reduced cytochrome cd complex of purified nitrite reductase with nitrite.

Purified nitrite reductase (150 µl of Fraction V, Table 34), 0.025 M Tris-HCl (pH 7.5) and 10 µmoles of sulphide, in final volume of 2.5 ml, were dispensed into a sample cuvette (Thunberg) and 20  $\mu\text{moles}$  of  $\text{NaNO}_2$  was placed in the side arm of the cuvette. The cuvette was vigorously evacuated and filled with pure  $N_2^{}$  gas as described in Section 2.6.3. After recording the spectra of the fully reduced enzyme, nitrite was tipped from the side arm and mixed thoroughly with the contents of the cuvette. The spectra was recorded at the times indicated below:

1 min after adding NO2 5 min after adding NO2

Reduced with S2-

15 min after adding NO<sub>2</sub>



Figure 46
### Table 43. Effect of pH on the absorbance characteristics of cytochrome d component of sulphide reduced cytochrome cd complex of the purified nitrite reductase (Fraction V, Table 34).

Nitrite reductase enzyme (200  $\mu$ l containing 2.5 mg protein) and 2.5 ml of 0.05M phosphate buffer of various pH values were dispensed in Thunberg cuvettes. The enzyme was then reduced with 10  $\mu$ moles of sulphide solution. The absorbance of 645 nm before adding sulphide and at 462 nm (Y - peak) and at 610 nm ( $\alpha$  - peak) of the sulphide-reduced cytochrome *d* component of the purified nitrite reductase were noted.

рН	Absorbanc <u>Reduced Υ (462 nm)</u> Reduced α (610 nm)	ce ratios <u>Reduced α (610 nm)</u> Oxidized (645 nm)
8.0	1.7	0.93
7.6	1.1	1.15
7.0	1.0	1.20
5.5	0.42	1.80

# 3.2.10.2 Effects of cyanide and carbon monoxide on the absorption spectra of purified nitrite reductase

The results in Figure 47 show that cyanide had a profound effect on spectra of the sulphide-reduced enzyme. The  $\gamma$ -band of cytochrome d at 462 nm was abolished when the enzyme was reduced with sulphide in the presence of lmM KCN and instead, two bands at 443 and 482 nm appeared. In addition, the  $\alpha$ -peak of cytochrome dshifted to 627 nm and the absorbance peak at 523 nm diminished significantly. The 554 nm component of the bifurcated  $\alpha$ -band of the reduced cytochrome c became more prominent as compared to the one at 549 nm. Cyanide had no effect on the absorption band at 418 nm. The spectra of the oxidized enzyme was not altered in the presence of cyanide.

On reducing the enzyme with sulphide, in the presence of CO, the  $\gamma$ -peak of cytochrome d was suppressed and the absorption band at 610 nm was replaced by a broad plateau around 650 nm (Figure 48). The bands of reduced cytochrome c remained unaffected by CO.

#### 3.2.10.3 Pyridine haemochromogen

Cytochrome d was reduced by akaline pyridine even in the absence of a reductant; a sharp absorption band appeared at 620 nm in place of a broad band at 610 nm which was present in the sulphidereduced enzyme. The 664-667 nm band was not detected in the alkaline pyridine preparation (Figure 49). In contrast, the cytochrome ccomponent of the enzyme was not reduced by alkaline pyridine unless sodium dithionite was added. Under these conditions, the bifurcated  $\alpha$ -peak of cytochrome c at 549-554 nm was replaced by a sharp peak at 549 nm. In addition, the ratio of  $\alpha/\beta$  bands of reduced cytochrome cincreased to 2.4 as compared to 1.0, recorded in the absence of

# Figure 47. Effect of cyanide on the spectra of purified nitrite reductase (Fraction V, Table 34).

Details of the procedure are the same as in Figure 44 except that 2.5 µmoles of KCN was included in the sample cuvette (Thunberg). After evacuating and flushing the cuvette with pure  $N_2$  gas, 10 µmoles of sulphide was added from the side arm of the Thunberg cuvette and difference spectra recorded after 15 min.

Figure 48. Effect of CO on the spectra of purified nitrite reductase (Fraction V, Table 34).

Experimental details as described in Figure 44 except that CO was bubbled through the sample cuvette for 5 min and the following spectra were recorded:

Absolute spectrum of CO-treated enzyme

CO-sulphide-reduced versus oxidized, difference spectra

CO-sulphide-reduced *versus* sulphide reduced, difference spectra.







n m

# Figure 49. Spectra of purified nitrite reductase in the presence of alkaline pyridine and sodium dithionite.

The method of Newton (1969) described in Section 2.7.4 was used for examining the formation of pyridine haemochromogen. 250 µl of purified enzyme (Fraction V, Table 34) in 0.025 M Tris-HCl buffer (pH 7.5) was made 25% (v/v) and 0.1 M with respect to pyridine and NaOH respectively. This mixture (5 ml) was then dispensed into two 1 cm cuvettes. To the reference cuvette, a crystal of  $K_3Fe(CN)_6$  was added and the difference spectra was recorded (----). Difference spectra was recorded again after adding Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub> to the sample cuvette (-----).

# Figure 50. Absolute and sulphide-reduced versus oxidized spectra of SDS-treated nitrite reductase.

Purified nitrite reductase (12.5 mg protein/ml) was incubated with 1.5% (w/v) SDS at  $30^{\circ}$  for 30 min. The absolute spectrum of SDS-treated enzyme (----) and difference spectra, after reducing with sulphide (-----), were recorded as described in Section 2.6.3.



Figure **50** 



### 3.2.11 Dissociation of purified nitrite reductase into cytochrome c and haem d

Cytochrome c and haem d were dissociated from the purified nitrite reductase on incubating it with SDS. The optimum concentration of SDS required to dissociate these components from the enzyme ranged between 1.0 to 1.5 mg SDS per mg protein.

The absorption spectra of the SDS-treated enzyme after its reduction with sulphide is illustrated in Figure 50. The appearance of absorption bands at 418, 523 and 553 nm indicated that only the cytochrome c component of the SDS-treated enzyme was reduced by sulphide. Thus the absorption bands of reduced cytochrome d were not detected on reducing the SDS-treated enzyme with sulphide.

Gel-filtration, through a Sephadex G-100 column, resolved the SDS-treated enzyme into two distinct components, one pink and the other green coloured fraction as shown in Figure 51.

The pink fraction, containing the cytochrome c moiety, was eluted first from the column followed by dark green cytochrome dcontaining fractions. The absorbance of these two fractions at 280 nm indicated that bulk of the protein was associated with cytochrome ccontaining fractions (Figure 52).

#### 3.2.11.1 Spectral properties of dissociated cytochrome c

Absorption spectra of the cytochrome c containing component, recovered after passage of the SDS-treated enzyme through a Sephadex G-100 column, is illustrated in Figure 53. It had absorption bands

### Figure 51. Separation of cytochromes c and d on Sephadex G-100 column.

Purified nitrite reductase (Fraction V, Table 34), after incubation with 1.5% (w/v) SDS at  $30^{\circ}$  for 30 min as described in Section 2.5.8 was loaded on to a Sephadex G-100 column (2.5 x 40 cm). Separation of SDS-treated nitrite reductase into red (containing cytochrome c) and green (cytochrome d) fractions is illustrated in the Figure.

4





### Figure 52. Elution profile of cytochromes c and d from Sephadex G-100 column.

Purified nitrite reductase (Fraction V, Table 34) was incubated with 1.5% (w/v) SDS at  $30^{\circ}$  for 30 min and loaded on to a Sephadex G-100 column (2.5 x 40 cm) which had been equilibrated, at room temperature, with 0.025 M Tris-HCl buffer (pH 7.5). The sample was eluted with the same buffer at a flow rate of 15 ml/hr. 3 ml fractions were collected and the absorbance at 405 nm and 280 nm were measured.

> □----□ E<sub>280</sub> ■-----■ E<sub>405</sub>



### Figure 53. Absolute and dithionite-reduced versus oxidized, difference spectra of dissociated cytochrome c.

The sample and the reference cuvettes each, in a final volume of 2.5 ml, contained dissociated cytochrome c (1.6 mg protein) and 0.025 M Tris-HCl buffer (pH 7.5). To the sample cuvette, a few crystals of Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub> were added and absorption spectra were recorded as described in Section 2.6.3.

Absolute spectra of dissociated cytochrome c

Difference spectra, Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub>-reduced *versus* oxidized.



Figure 53

nm

at 405 and 530 nm and on reduction with sodium dithionite at 416-418, 523 and 549-554 nm. This spectra was almost identical with that obtained for cytochrome c of the undissociated reduced enzyme (Figure 42) except that bifurcation of  $\alpha$ -band was not very marked in the dissociated cytochrome c. The ratio of  $\alpha/\beta$  bands of the separated cytochrome c was 0.81 which was lower than the value of 1.0 for the undissociated enzyme.

On adding sulphide to the dissociated cytochrome c, absorption spectra similar to that after reduction with sodium dithionite was observed except that the rate of reduction was much slower. Although only a slight bifurcation of the  $\alpha$ -band of the either dithionite- or sulphide-reduced cytochrome c was observed, it became more pronounced on adding nitrite (Figure 54). In the presence of nitrite, the  $\alpha/\beta$ ratio of the cytochrome c was lowered to 0.42. Reduction of dissociated cytochrome c with sulphide, in the presence of lmM KCN, showed a sharp peak at 551 nm and the addition of nitrite did not result in the bifurcation of the  $\alpha$ -peak as was observed in the absence of KCN. This bifurcation of  $\alpha$ -peak of the reduced cytochrome c was caused specifically by nitrite since nitrate did not produce this effect. The addition of nitrite to the dithionite-reduced cytochrome  $c_{551}$ , which was isolated from T. denitrificans (Section 3.2.7.2, Figure 37) did not affect its The dissociated cytochrome c was not reduced by sulphite. α-peak.

The spectrum for the pyridine haemochromogen of dissociated cytochrome c (Figure 55) was identical to that of the cytochrome c of the undissociated enzyme (Figure 49).

#### 3.2.11.2 Spectral properties of haem d

The haem d component separated from the SDS-treated purified

# Figure 54. Absorption spectra of sulphide-reduced dissociated cytochrome c.

Experimental details are as in Figure 53 except that 10 µmoles of sulphide, instead of  $Na_2S_2O_4$ , was added and the reaction mixture was incubated for 20 min. After recording the spectra, 20 µmoles of  $NaNO_2$  was added to the sample cuvette.

Α.	s <sup>2-</sup> -red	uced	versus	oxidized,
	differe	nce s	spectra	

в. \_\_\_\_

After adding NaNO<sub>2</sub> to A.



Figure 54

nm

Figure 55. Difference spectra of alkaline pyridine haemochromogen of cytochrome c, dissociated from cytochrome cd complex.

The method of Newton (1969), given in Section 2.7.4, was used for the preparation of pyridine haemochromogen. The cytochrome c (3.2 mg protein) in 0.025 M Tris-HCl buffer (pH 7.5) was made 25% (v/v) and 0.1 M with respect to pyridine and NaOH respectively. This mixture was dispensed equally into two cuvettes. After adding a small amount of Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub> to the sample cuvette and a crystal of  $K_3$ Fe(CN)<sub>6</sub> to the reference cuvette, the difference spectra was recorded.

161.



nitrite reductase, showed absorption bands at 405 and 680 nm with a shoulder around 480 nm (Figure 56). On reduction with sodium dithionite, two broad peaks at 460 and 610-620 nm were observed. Under anaerobic conditions, the haem *d* component was reduced slowly by sulphide and it showed absorption band at 457 nm and a broad peak at 610-640 nm. The ratio of absorbance at 457 and 610-640 nm was 2.2. Haem *d* was readily autooxidizable in air (Figure 57).

#### 3.2.12 Products of nitrite reduction by purified enzyme

The reduction products of  ${}^{15}$ N-labelled nitrite by the purified nitrite reductase were examined in a mass spectrometer. The data in Figure 58 show that both NO (mass 31) and N<sub>2</sub>O (mass 45) were produced. During the first 30 min, N<sub>2</sub>O was formed in relatively larger amounts than NO, however beyond this period, N<sub>2</sub>O content remained constant whereas NO continued to increase for a further 120 min incubation period. Only traces of N<sub>2</sub> gas (mass 29) were detected during the reduction of nitrite by the purified enzyme.

#### 3.2.13 Effect of NO gas on purified nitrite reductase

#### 3.2.13.1 Formation of NO- enzyme complex

On bubbling NO gas for 20 sec through the purified nitrite reductase, which had been previously reduced anaerobically with sulphide, a small but distinct peak appeared at 573 nm (Figure 59). At the same time, the height of the absorption bands of the reduced cytochromes cand d decreased. As demonstrated earlier (Figure 46), this band at 573 nm was also observed during the oxidation of the sulphide-reduced nitrite reductase enzyme with nitrite.

# Figure 56. Absolute and dithionite-reduced versus oxidized spectra of haem <u>d</u>.

Details of the procedure are as in Figure 53 except that haem d, isolated from purified nitrite reductase (Fraction V, Table 34) as described in Section 2.5.8, was used in place of dissociated cytochrome c.

Absolute spectra

Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub>-reduced versus oxidized, difference spectra.



# Figure 57. Sulphide-reduced versus oxidized difference spectra of haem d.

Haem d, 250  $\mu$ l (3.25  $\mu$ M) and 0.025 M Tris-HCl buffer (pH 7.5), in a total volume of 2.5 ml, were dispensed into both sample (Thunberg) and reference cuvettes. Side arm of the sample cuvette contained 10  $\mu$ moles of sulphide solution. The sample cuvette was thoroughly evacuated and flushed with high purity N<sub>2</sub> gas. This procedure was repeated three times. Sulphide solution was then tipped into the cuvette from the side arm and after 30 min incubation at room temperature, difference spectra was recorded.

Sulphide reduced *versus* oxidized difference spectra

в.

Α.

After bubbling air for 30 sec through A.

164.



Figure 57

nm

#### Figure 58.

### Effect of incubation time on the production of various nitrogenous gases from $\frac{15}{NO_2}$ by purified nitrite reductase (Fraction V, Table 34).

The experimental details are given in Section 2.6.7. The reaction mixture in Warburg flasks, in a final volume of l ml, contained the following: Tris-HCl (pH 7.5), 35 µmoles;  $Na^{15}NO_2$  (27 atom % excess), 2 µmoles and 0.2 ml of purified nitrite reductase enzyme (2.4 mg protein). The reaction was started by injecting 5 µmoles of sulphide solution into the side arm via subaseal. At specified time, the reaction was stopped by injecting 250 µl of 10 M KOH into the flasks. The gaseous products were identified in an A.E.I. mass spectrometer.

The results are expressed as relative peak heights (proportional to concentrations) and have been corrected for control values (without the enzyme).

• NO (mass 31) •  $N_2^{O}$  (mass 45) •  $N_2^{O}$  (mass 29).

165.





# Figure 59. Effect of NO gas on the spectra of purified nitrite reductase reduced with sulphide.

Purified enzyme (200  $\mu$ l of Fraction V, Table 34) and 0.05 M Tris-HCl buffer (pH 7.5), in a final volume of 2.5 ml, were dispensed into both sample (Thunberg) and reference cuvettes. The side arm of the sample cuvette contained 10  $\mu$ moles of freshly prepared sulphide solution. The cuvette was made anaerobic as described in Section 2.6.3. Sulphide was mixed with the contents of the cuvette and incubated for 10 min. Then NO gas was bubbled through the reaction mixture for 20 sec. The cuvette was then immediately closed under the stream of N<sub>2</sub> gas and the difference spectra recorded after 2 min.

# Figure 60. Effect of sulphide and dithionite on the spectra of purified nitrite reductase treated with NO gas.

Purified nitrite reductase (200  $\mu$ l) and 0.05 M Tris-HCl buffer (pH 7.5) were dispensed into both sample and reference cuvettes. After bubbling NO gas through the sample cuvette for 15 sec, the enzyme was reduced with either 10  $\mu$ moles S<sup>2-</sup> or 2 mg Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub> and the following spectra were recorded:

Α.	 NO-treated enzyme
в.	 $s^{2-}$ added to A
c.	 $Na_2S_2O_4$ added to A.



Figure 60



In subsequent experiments, the purified nitrite reductase was treated with NO gas for 15 sec in the absence of the reductant, viz. sulphide, to ascertain whether prior reduction of the enzyme was essential for the formation of the NO- enzyme complex. Under these conditions the absorption peak at 573 nm was not detected (Figure 60). The absorption bands at 525 and 645 nm, associated with the purified nitrite reductase were, however, replaced by peaks at 530 and 565 nm after treatment with NO gas. The addition of sulphide to this NO-treated enzyme did not change the pattern of the spectra except that the peak heights were lowered. On adding sodium dithionite to the NO-treated enzyme, the peaks at 530 and 565 nm disappeared and a small shoulder appeared at 573 nm.

#### 3.2.13.2 Dissociation of the NO-enzyme complex

The spectra in Figure 61 indicate that the absorbance at 573 nm, due to the formation of a complex between NO and the sulphide-reduced enzyme, decreased with time. It was accompanied by the oxidation of both cytochromes c and d. After a 30 min incubation, the absorbance bands of reduced cytochrome c and d and that at 573 nm disappeared and two prominent peaks at 530 and 565 nm appeared. As mentioned earlier similar peaks were observed on treating the enzyme with NO gas in the absence of sulphide. The colour of the enzyme preparation at this stage, had turned yellow and the pH of the reaction mixture was very acid (pH 3.5).

# 3.2.13.3 Effects of pH, CO, CN on the formation and dissociation of the NO-enzyme complex

As noted in the preceding experiment, the pH of the reaction mixture was markedly lowered after the treatment with NO gas. The

167.

### Figure 61. Effect of incubation time on the formation of complex between NO and sulphide-reduced purified nitrite reductase (Fraction V, Table 34).

Experimental details as in Figure 59 except that the difference spectra were recorded at the following times after bubbling NO gas through the reaction mixture:

And the second second second second second		After 3	min
	ŝ		
		After l	0 min
		After 3	0 min

### Figure 62. Formation of complex between NO and sulphidereduced purified nitrite reductase at pH 10.2.

Experimental details are as in Figure 59 except that 0.1 M Tris-HCl buffer (pH 10.2) was used and difference spectra were recorded at the following times after bubbling NO gas through the reaction mixture:

10.00

After 3 min

After 1 hr.

Figure 61



Figure **62** 



nm

effect of NO gas was, therefore, investigated in the presence of a buffer of a higher ionic strength (0.1M Tris-HCl instead of 0.05M) at pH 10.2. On bubbling NO gas through the sulphide-reduced enzymes, the peak at 573 nm did not appear immediately (Figure 62). After 1 hr, the absorbance bands of cytochromes c and d were markedly diminished and this was accompanied by the appearance of a peak at 573 nm. The pH of the reaction mixture at this stage was 7.6.

From these results, it is apparent that the formation of the NO- enzyme complex, at 573 nm, occurred at the neutral pH, while the peaks at 530 and 565 nm appeared at acid pH (below 6.5). As previously mentioned, when NO gas was bubbled through the enzyme (without reductant) absorption bands appeared at 530 and 565 nm and the pH of the reaction mixture was lowered to 2.3. However on raising the pH to 6.5 with IN NaOH, the peak heights at 530 and 565 nm were markedly reduced and at pH 7.8, they disappeared altogether (Figure 63). These bands reappeared on lowering the pH of the reaction mixture to 3.3 with N HC1.

Pre-treatment of the sulphide-reduced enzyme with CO for 5 min, did not affect the formation of the peak at 573 nm when NO gas was subsequently bubbled through this preparation. CO, however, delayed the disappearance of the 573 nm band (Figure 64) by about 1 hr.

Cyanide had no effect on either the appearance or disappearance of the absorption band at 573 nm (Figure 65).

#### 3.2.13.4 Effect of NO on dissociated cytochrome c and haem d

The cytochrome c dissociated from purified nitrite reductase was reduced with sulphide and then treated with NO gas. Instead of the

# Figure 63. Effect of pH on the formation of complex between NO and the purified nitrite reductase.

Purified enzyme (200 µl of Fraction V, Table 34) and 0.1 M Tris-HCl buffer (pH 7.5) were dispensed into both sample (Thunberg) and reference cuvettes. No gas was bubbled through the sample cuvette for 20 sec. pH of the NO-treated sample was 3.4. After recording the difference spectra, pH of the reaction mixture was progressively adjusted to 4.0, 6.5 and 7.8 with 2N NaOH. The absorption spectra at each pH value was recorded. pH of the reaction mixture was then lowered with 2N HCl from 7.8 to 3.3 and the absorption spectra recorded.

Α.		pH 3.4 or 4.0
в.		рН 6.5
C.		рН 7.8
D.	$- \Delta - \Delta - \Delta - \Delta - \Delta -$	pH of C lowered to 3.3.

Figure 64. Effect of CO on the formation and dissociation of complex between NO and sulphide-reduced purified enzyme (Fraction V, Table 34).

Experimental details are as in Figure 59 except that CO was bubbled through the reaction mixture for 5 min prior to treating it with NO gas for 20 sec. After the latter treatment, the difference spectra was recorded at the following time intervals:

 After bubbling CO gas through t reaction mixture	he
 l hr after bubbling NO gas	12
 2 hr after bubbling NO gas.	



Figure 64



### Figure 65. Effect of cyanide on formation and dissociation of the complex between NO and sulphide-reduced purified nitrite reductase.

Experimental details are as in Figure 59 except that 1 mM KCN was also included in the reaction mixture. Difference spectra at following times, after bubbling NO through reaction mixture for 20 sec, were recorded:

After 2 min

After 30 min.

Figure 66.

# Effect of NO on spectra of sulphide-reduced cytochrome c, dissociated from the cd complex

#### of purified nitrite reductase.

0.2 ml (3.6 mg protein) of cytochrome c component, isolated from purified nitrite reductase by the procedure given in Section 2.5.8, was dispensed into a cuvette containing 0.05 M Tris-HCl buffer (pH 7.5). After reducing it with  $S^{2-}$  as described in Figure 54, NO gas was passed through the reaction mixture for 15 sec.

Α.	 $s^{2-}$ -reduced spectra of dissociated cytochrome $c$ .
в.	 2 min after treatment of A with NO gas.









normal absorption bands at 523 and 549-554 nm, two peaks appeared at 530 and 565 nm (Figure 66). A similar experiment with haem d showed that NO gas abolished the spectral bands at 405 and 680 nm.
### 4. DISCUSSION

## 4.1 Nitrate reductase

Nitrate reductase is the first enzyme involved in both assimilatory and dissimilatory nitrate reduction in microorganisms. Since in the latter process nitrate serves as an alternate terminal oxidant to O2, dissimilatory nitrate reductase is generally integrated with the respiratory chain of the organism. The association of nitrate reductase in T. denitrificans with cell-membranes, shown in the present investigation, confirms the earlier work of Adams et al. (1971) and is in accord with its role in cell respiration under anaerobic conditions. With the exception of Spirillum itersonii (Gauthier et al., 1970) and Clostridium perfringens (Chiba and Ishimoto, 1973, 1977), respiratory nitrate reductases from various bacteria, Escherichia coli (Iida and Taniguchi, 1959), Pseudomonas aeruginosa (Fewson and Nicholas, 1961B), Micrococcus denitrificans (Lam and Nicholas, 1969A, B), Aerobacter aerogenes (Van't Riet and Planta, 1969), Pseudomonas denitrificans (Radcliffe and Nicholas, 1970) and Bacillus stearothermophilus (Kiszkiss and Downey, 1972) have been shown to be associated with the cell membranes.

Dissimilatory nitrate reductase in *T. denitrificans* differs from nitrate reductase in other organisms because the reduction of nitrate in this chemolithotrophic bacterium is linked to the oxidation of sulphur compounds (Baalsrud and Baalsrud, 1954; Peeters and Aleem, 1970; Adams *et al.*, 1971; Aminuddin and Nicholas, 1974B). The work presented here confirms that nitrate was reduced, both in the Pl00 fraction and in the partially purified enzyme, when sulphite was the electron donor. In addition, as reported by Adams *et al.* (1971), NADH was also an effective reductant for nitrate reduction in the Pl00 fraction.

In the present investigation, the properties of sulphite- and NADH-linked nitrate reductase activities were compared. The enzyme activity with sulphite and NADH as electron donors had different pH optima at 8.0 and 7.0 respectively. These values are similar to the pH optima reported by Adams *et al.* (1971). The sulphite-linked nitrate reductase but not the NADH-linked enzyme activity was markedly suppressed in phosphate buffer. It is likely that this anion affected the sulphitelinked enzyme by acting as a pseudo end product because of its chemical similarity with sulphate. This mechanism was proposed for the inhibition by phosphate of the sulphite : cytochrome *c* oxido-reductase from *Thiobacillus thioparus* by Lyric and Suzuki (1970A).

The NADH-linked nitrate reductase activity was relatively more labile than the sulphite-linked enzyme during storage and also at elevated temperatures. Restoration of the NADH-dependent activity of the stored enzyme by GSH indicates the involvement of -SH groups and its greater sensitivity to thiol binding reagents, viz. p-CMB and iodoacetamide, than the sulphite-dependent activity further supports this conclusion. The very marked inhibition of the sulphite-linked activity by NEM was probably due to the formation of an NEM-sulphite complex (Ellis, 1964) rather than a direct effect on the enzyme. The requirement of -SH groups for the activity of dissimilatory nitrate reductases has been demonstrated in *P. aeruginosa* (Fewson and Nicholas, 1961B), *M. denitrificans* (Lam and Nicholas, 1969B) and *P. denitrificans* (Radcliffe and Nicholas, 1970). Highly purified nitrate reductases from *M. halodenitrificans* (Rosso *et al.*, 1973), *E. coli* (Forget, 1974) and *M. denitrificans* (Forget, 1971), all contained labile sulphide. The results presented herein indicate that both the sulphite- and NADH-linked nitrate reductase activities in *T. denitrificans* were affected by the inhibitors of electron transport chain. The differential sensitivity of the nitrate reductase activity with these reductants towards flavoprotein inhibitors suggests the involvement of two different flavoproteins or dehydrogenases for the transfer of electrons from sulphite and NADH to nitrate. The isolated FAD from the solubilized sulphite-linked nitrate reductase, as reported herein, suggests that it functions as an electron carrier for the enzyme. A requirement for flavins for dissimilatory nitrate reductases has been demonstrated in *E. coli* (Taniguchi *et al.*, 1956), *Achromobacter fisheri* (Sadana and McElroy, 1957), *P. aeruginosa* (Fewson and Nicholas, 1961B) and *M. denitrificans* (Lam and Nicholas, 1969B). Assimilatory nitrate reductases from various organisms are also flavoprotein (Nason, 1962).

Downey (1962) has shown that U.V. light destroys vitamin K type electron carriers. The diminished activities of both the sulphiteand NADH-linked nitrate reductase after exposure to U.V. light thus indicates that a quinone is required for nitrate reduction in *T. denitrificans*. This was confirmed by the inhibition of both these activities by a quinone antagonist, dicumarol. The failure of exogenously supplied ubiquinone to restore nitrate reductase activity in the U.V.-irradiated Pl00 fraction of *T. denitrificans* may well be due to the damaged quinone remaining tightly bound to the membranes (Knook and Planta, 1971A). Ubiquinone has been implicated in the reduction of nitrate in other gram-negative bacteria including *E. coli* (Itagaki, 1964; Enoch and Lester, 1974), *A. aerogenes* (Knook and Planta, 1971B).

As reported herein, nitrate reduction in P100 fraction with either sulphite or NADH as reductant was inhibited by antimycin A which blocks the transfer of electrons between cytochromes b and c. A role for cytochromes b and c in sulphite-linked nitrate reduction has been inferred earlier by Peteers and Aleem (1970), Adams et al. (1971), and Aminuddin and Nicholas (1974B). As shown in the current investigations, spectral changes in the P100 fraction, on its reduction with either sulphite or NADH and its subsequent oxidation with nitrate provides further evidence for cytochrome c as an electron carrier in the reduction of nitrate. Several other observations confirm this Thus a cytochrome  $c_{554}$ , isolated from the solubilized nitrate view. reductase, was reduced by sulphite in the presence of catalytic amounts of nitrate reductase. This reduced cytochrome  $c_{554}$  served as effective electron donor for nitrate reduction. The low  $K_m$  value of purified nitrate reductase for the reduced cytochrome  $c_{554}$  indicates a high affinity of the enzyme for this electron carrier. Because the redoxpotential of nitrate/nitrite couple at pH 7.0 is +421 mV (Burton, 1957), the reduction of nitrate via cytochrome  $c_{554}$  (E' = +223 mV) is thermodynamically possible.

The molecular weight and the redox-potential of the cytochrome  $c_{554}$  isolated from the solubilized nitrate reductase from *T. denitrificans* was similar to that of mammalian cytochrome *c* (Dutton *et al.*, 1970). A cytochrome of the *c* type, with a redox-potential of +270 mV, prepared from *T. denitrificans* also mediated electron flow between thiosulphate or sulphite and nitrate (Aubert *et al.*, 1958; Milhaud *et al.*, 1958).

The participation of cytochrome c in nitrate reduction in T. denitrificans, as reported in the present work, contrasts with many of the other facultative anaerobic bacteria where electrons are diverted to nitrate from cytochrome b of the respiratory chain. Thus cytochrome b has been shown to be the penultimate electron donor for nitrate reduction in E. coli (Itagaki et al., 1962; Ruiz-Herrera and DeMoss, 1969; Enoch and Lester, 1974; Garland et al., 1975), M. denitrificans (Scholes and Smith, 1968; Lam and Nicholas, 1969B), P. denitrificans (Radcliffe and Nicholas, 1970), B. stearothermophilus (Kiszkiss and Downey, 1972A) and A. aerogenes (Van't Riet et al., 1972; Knook et al., 1973). Only in P. aeruginosa (Fewson and Nicholas, 1961B) and in E. coli (Gray et al., 1963; Fujita and Sato, 1966) does cytochrome c function in nitrate reduction. The physiological significance of the oxidation by nitrate of reduced cytochrome  $c_{552}$  in E. coli has however been questioned since this cytochrome is believed to be involved in nitrite but not in nitrate reduction. A purified nitrate reductase from A. fisheri showed cytochrome c type absorption spectra (Sadana and McElroy, 1957).

The requirement of molybdenum for nitrate reduction with either sulphite or NADH as the reductant was indicated in the present studies by the inhibitory effects of dithiol and thiocyanate which are known to chelate this micronutrient (Meriwether *et al.*, 1966). Molybdenum is a constituent of purified nitrate reductase in *M. denitrificans* (Lam and Nicholas, 1969B) and its participation in nitrate reduction was confirmed by ESR studies with the enzyme from *P. aeruginosa* (Fewson and Nicholas, 1961B) and *M. denitrificans* (Forget and Dervartanian, 1972).

Inhibition of nitrate reductase from *T. denitrificans* by azide and cyanide with either sulphite or NADH as the reductant reported in this thesis is in agreement with the results for the enzyme in *E. coli*  (Iida and Taniguchi, 1959; Forget, 1974) and *M. denitrificans* (Forget, 1971). However the assimilatory nitrate reductases are less sensitive to azide (Payne, 1973). The competitive inhibition of nitrate reductase by azide and the non-competitive effect of cyanide with respect to nitrate, are among some of the properties which distinguish this enzyme from the assimilatory type (Pichinoty, 1964).

The competitive inhibition by chlorate with respect to nitrate, observed in this work, suggests that it acts as an alternate substrate for the enzyme. This is in accord with the effect of chlorate on respiratory nitrate reductases from several other microorganisms (Fewson and Nicholas, 1961B; Hackenthal and Hackenthal, 1965; Payne, 1973).

The particulate nitrate reductase from T. denitrificans was solubilized by sodium deoxycholate treatment. After treatment of the P100 fraction with the detergent, the specific activity of the sulphitelinked nitrate reductase increased, whereas the NADH-dependent enzyme was inactivated. A similar loss in NADH-linked nitrate reductase after its dissociation from cell-membranes has been reported in M. denitrificans (Lam and Nicholas, 1969B), P. denitrificans (Radcliffe and Nicholas, 1970), B. stearothermophilus (Kiszkiss and Downey, 1972A). This loss in enzyme activity has been attributed to the disruption of electron flow from NADH to nitrate (Kiszkiss and Downey, 1972B). In B. stearothermophilus, the activity of the solubilized enzyme was restored on adding divalent cations especially Mg<sup>2+</sup> (Kiszkiss and Downey, 1972B). In the present studies, the NADH-linked nitrate reductase of the deoxycholate treated Plo0 fraction was partially restored on adding PMS and almost completely with menadione. In addition, menadione alleviated the effects of various inhibitors, viz. atabrine, rotenone, HQNO and antimycin A on the NADH-

linked but not the sulphite-linked nitrate reductase activity. This effect is probably associated with an active NADH : menadione reductase which in the presence of menadione transfers electrons from NADH directly to cytochrome c. In fact in E. coli, Medina and Heredia (1958) have described an NADH-dependent system coupled to menadione reductase which utilized vitamin  $K_3$  as an electron carrier for the reductases has been demonstrated in E. coli (Bragg, 1965) and in Mycobacterium phlei (Asano et al., 1965).

From the results presented in this thesis, it may be concluded that the reducing equivalents from either sulphite or NADH are transferred to nitrate via the electron transport chain composed of flavins, quinone, cytochromes b, c and finally to the Mo-containing Nitrate reductase activities with sulphite and NADH nitrate reductase. respectively as the reductants, however, showed some important differences: the greater dependence of NADH-linked activity on -SH groups, inactivation by deoxycholate treatment of the NADH system only, differential effects of flavin inhibitors and difference in their pH optima, indicate that the electrons from these two reductants are donated at separate sites in the electron transport chain. Similar inhibitory effects of dicumarol, U.V. irradiation, antimycin A, metal chelators and the observed involvement of cytochrome  $c_{554}$  in both the sulphite- and NADH-linked nitrate reductase activities indicate that the components of the electron transfer chain beyond the flavins are common to both systems.

During the present investigation, a comparison was made between the sulphite oxidizing systems with nitrate and  $O_2$  as terminal electron acceptors. Sulphite-linked  $O_2$  uptake and nitrate reduction was maximal in the Pl00 fraction, indicating that both systems were associated with

similar cell membranes. Kodama and Mori (1968) found that the sulphite oxidizing system linked to O<sub>2</sub> was associated with cell membranes in *Thiobacillus thiooxidans*.

In the present work it is shown that as with nitrate reductase, sulphite and NADH were effective reductants for O<sub>2</sub> uptake. The inhibition of sulphite-linked O<sub>2</sub> uptake by thiol-binding agents, reported herein, has been observed for sulphite oxidizing systems in *Thiobacillus novellus* (Charles and Suzuki, 1966) and *Thiobacillus thioparus* (Lyric and Suzuki, 1970A). However, a particulate sulphite oxidase from *T. thiooxidans* was insensitive to -SH group inhibitors (Kodama and Mori, 1968).

Sulphite-linked  $O_2$  uptake and nitrate reduction are shown herein to be affected to about the same extent by various inhibitors of the electron transport chain, viz. rotenone, atabrine, dicumarol, HQNO and antimycin A. Peeters and Aleem (1970) however observed aerobic oxidation of sulphite in *T. denitrificans* was markedly less sensitive to rotenone than was the anaerobic oxidation of sulphite.

In the work presented here, the sulphite-dependent 0<sub>2</sub> uptake is more susceptible to cyanide than the sulphite-linked nitrate reductase whereas the reverse was true for azide.

The association of O<sub>2</sub> uptake and nitrate reduction with the same cellular fraction, their identical requirements for the reductants and comparable sensitivities to the various inhibitors of the respiratory chain indicate that as in the other nitrate dissimilatory bacteria, the reducing equivalents were transferred from sulphite and NADH to either O<sub>2</sub> or nitrate via a common respiratory chain.

On the basis of these observations, the following scheme is proposed for the transfer of electrons from sulphite and NADH respectively



to nitrate in T. denitrificans.

Menadione-by-pass

The inhibitory effects of NAD<sup>+</sup> and 5'-AMP on the NADH-linked nitrate reductase reported herein, are probably important in conserving NADH for the anabolic reactions of the cell. In chemolithotrophic bacteria including T. denitrificans, the incorporation of CO2 into carbon compounds requires a supply of ATP and NADH. In these bacteria, ATP is produced from the oxidation of inorganic compounds whereas, NADH is generated through an ATP-driven reverse electron flow (Aleem et al., The present investigation shows that in T. denitrificans NADH 1963). is also oxidized via the respiratory chain by either nitrate or 0,. Therefore the observed effects of NAD<sup>+</sup> and 5'-AMP on the NADH-linked nitrate reductase activity are probably important in regulating the The positive cooperative effect of NADH on rate of NADH oxidation. the NADH-linked nitrate reductase in the S18 fraction indicates that this nucleotide would be utilized at an accelerated rate only when present at relatively high concentrations. Whenever its concentration

in the cell declines, the accumulated NAD<sup>+</sup> besides inhibiting the NADH-linked enzyme would also suppress the positive cooperative effect of NADH on nitrate reductase. Thus under conditions of a limited supply of NADH, the accumulated NAD<sup>+</sup> would restrict the utilization of NADH for nitrate reduction and conserve it for anabolic reactions.

The inhibition of the NADH-linked nitrate reductase by 5'-AMP suggests that its activity is also controlled by the energy status of the bacterium. Whenever the concentration of ATP in the cell diminishes, elevated concentrations of 5'-AMP, produced from the action of adenylate kinase on ADP, would decelerate the utilization of NADH by nitrate. Such a regulation in chemolithotrophic bacteria, in which NADH is generated via an ATP-driven reverse electron flow, is essential particularly when the concentration of ATP in the cell is low. These findings are pertinent in view of the proposal by Aleem (1977) that the functioning of electron transport chain with respect to the forward and reverse electron flow could be dependent on the phosphorylation state of the adenine nucleotides. Thus the control of NADH-linked nitrate reductase by NAD<sup>+</sup> as well as energy charge of the bacterium may well be of physiological importance. Respiratory NADH-oxidase in Mycobacterium tuberculosis (Worcel et al., 1965) and Mycobacterium phlei (Bogin et al., 1969) and NADH-dehydrogenase from E. coli (Dancey and Shapiro, 1976) are also known to be regulated by NAD<sup>+</sup>, AMP and Pi.

# 4.2 Nitrite reductase

The association of nitrite reductase with the cell membranes in T. *denitrificans* reported in the present investigation contrasts with

the soluble dissimilatory nitrite reductases in the other denitrifying bacteria (Radcliffe and Nicholas, 1968; Lam and Nicholas, 1969C; Newton, 1969; Iwasaki and Matsubara, 1972; Cox and Payne, 1973). The present work also shows that the Pl00 fraction from *T. denitrificans* contained an active sulphide-linked  $O_2$  uptake system in addition to the sulphidelinked nitrite reductase. Thus the oxidation of sulphide in this fraction was linked to either nitrite or  $O_2$  as a terminal oxidant.

The activity of nitrite reductase in this bacterium was also linked to the oxidation of sulphide. None of the other sulphur compounds examined was effective in reducing nitrite. Although the intact cells reduce nitrite with thiosulphate as a reductant (Baalsrud and Baalsrud, 1954; Ishaque and Aleem, 1973), in the present investigations, thiosulphate was not effective for nitrite reduction in the Pl00 fraction. As pointed out by Aminuddin (1974), it is likely that an enzyme system, probably the cleaving enzyme, required for the oxidation of thiosulphate *via* nitrite reductase, was either inactivated or removed during the preparation of the Pl00 fraction.

Besides sulphide, nitrite reductase in PlOO fraction functioned with either NADH plus menadione or with ascorbate in conjunction with various electron carriers, viz. PMS, DCIP, TMPD as donor systems. The purified nitrite reductase although active with sulphide and ascorbate plus PMS did not utilize NADH plus menadione as a reductant. This effect was probably due to either the inactivation or removal of the NADHdependent menadione reductase which mediated the transfer of electrons from NADH via menadione to nitrite reductase.

During these investigations, the effects of various inhibitors on the sulphide-linked nitrite reduction and  $O_2$  uptake were examined.

Both activities were inhibited by p-CMB, iodoacetamide and NEM. The more pronounced inhibition by NEM was probably due to the formation of a complex between sulphide and NEM (Ellis, 1964). The effect of these inhibitors indicates a requirement for -SH groups for the two activities and is in accord with the properties of dissimilatory nitrite reductases from *P. aeruginosa* (Walker and Nicholas, 1961) and *P. denitrificans* (Radcliffe and Nicholas, 1968).

The results presented in this thesis show that the inhibitors of the respiratory chain did not affect the sulphide-linked nitrite reductase or  $O_2$  uptake. The insensitivity of these reactions towards inhibitors of the electron transport chain (except dicumarol) indicates the non-involvement of some electron carriers of the respiratory chain for the transfer of electrons from sulphide to either nitrite or  $O_2$ . Ishaque and Aleem (1973) have also observed that some inhibitors of the electron transport chain did not affect nitrite reductase activity in the cells of *T. denitrificans*. They attributed this to the modification in the path of electrons were transferred directly from a flavoprotein system to nitrite. However the work reported herein precludes the involvement of either flavin or cytochrome *b* in the sulphidedependent nitrite reduction in the P100 fraction or in the purified enzyme.

The inhibition of the sulphide-dependent nitrite reductase and O<sub>2</sub> uptake in the Pl00 fraction by cyanide, 2,2'-bipyridyl, O-phenanthroline and 8-hydroxyquinoline indicates participation of transition metal(s). During these studies, the purified nitrite reductase has been shown to contain iron but not copper. These effects of the metal chelating agents on the nitrite reductase are similar to those reported for the respiratory nitrite reductases from other bacteria (Walker and Nicholas, 1961;

Radcliffe and Nicholas, 1968; Lam and Nicholas, 1969C; Iwasaki and Matsubara, 1972).

The combined results on the nature of the reductants required and effects of the inhibitors of the electron transport chain on the sulphide-linked nitrite reductase and  $O_2$  uptake activities indicate that the reducing equivalents are transferred from sulphide to either nitrite or  $O_2$  via the same electron transfer chain. Lam and Nicholas (1969C) also concluded that in *M. denitrificans*, the oxidation of NADH and succinate with either  $O_2$  or nitrite as the terminal electron acceptor was mediated via the same electron transport system.

During the investigation reported herein, the membrane-bound, sulphide-linked nitrite reductase was dissociated from the membranes of cells treated with cold acetone. The solubilized enzyme was purified twenty-fold and had a molecular weight of 120,000 daltons. Although this value is slightly higher than that of 90,000 daltons of the crystalline enzyme from *Alcaligenes faecalis* (Iwasaki and Matsubara, 1971) it is in agreement with the molecular weight of 120,000 daltons for nitrite reductase from *M. denitrificans* (Lam and Nicholas, 1969C; Newton, 1969) and of 121,000 daltons of *P. aeruginosa* (Gudat *et al.*, 1973).

As with the enzyme from *M. denitrificans* (Lam and Nicholas, 1969C) and *P. aeruginosa* (Horio *et al.*, 1961A; Gudat *et al.*, 1973) the purified nitrite reductase from *T. denitrificans* did not contain copper. However, as for *P. aeruginosa* (Gudat *et al.*, 1973) iron was present in the purified enzyme from *T. denitrificans*.

The results presented in this thesis show that on adding sulphide anaerobically to either PlOO fraction or the purified enzyme, a composite spectra for cytochromes c and d were obtained. The spectra of the

cytochrome c component of the cd complex of the purified enzyme, however, differed from that of mammalian cytochrome c. A bifurcated  $\alpha$ -band of the bacterial cytochrome c was produced on adding reductant and it had a lower  $\alpha/\beta$  absorbance ratio than mammalian cytochrome c. The spectra of cytochrome d of the reduced enzyme showed that the absorbance of its  $\gamma$ -band was much smaller. The absorbance ratio of  $\alpha/\gamma$  bands of cytochrome d was markedly influenced by pH of the reaction mixture. The cytochrome d component of nitrite reductase was reduced by alkaline pyridine even in the absence of a reductant. Thus cytochrome d of the cd complex showed characteristic features of a chlorin haem. Cytochrome d of the enzyme from P. *aeruginosa* (Yamanaka and Okunuki, 1963C) and M. *denitrificans* (Newton, 1969) have also been identified as a chlorin derivative of haem d.

186.

As reported herein, the sulphide-reduced cytochromes c and dof either the P100 fraction or the purified nitrite reductase were oxidized with nitrite but not with nitrate. These results signify that the sulphide-reduced cytochromes c and d participate in the reduction of nitrite. Nitrite reductases from P. *aeruginosa* (Horio *et al.*, 1961A), which was initially studied as a cytochrome oxidase but referred to as nitrite reductase in a later communication (Yamanaka and Okunuki, 1974) and from M. *denitrificans* (Lam and Nicholas, 1969C; Newton, 1969) and A. *faecalis* (Iwasaki and Matsubara, 1971) also contain cytochromes c and d. As with the enzyme from P. *aeruginosa* (Horio *et al.*, 1961A) and A. *faecalis* (Iwasaki and Matsubara, 1971), it is shown in this thesis that the purified nitrite reductase from T. *denitrificans* contained cytochromes c and d in 1 : 1 molar ratio. The results for the reduction and reoxidation of cytochromes c and d reducing equivalents from sulphide are transferred first to cytochrome c and then to cytochrome d which acted as an immediate electron donor for the reduction of nitrite. A similar sequence of reduction of cytochromes c and d was observed in P. aeruginosa (Wharton et al., 1973; Shimada and Orii, 1976). The proposed sequence of electron transfer for cd complex from T. denitrificans was further confirmed by a more positive redox-potential of cytochrome d than for cytochrome c. However, the redox-potential of cytochrome d of the enzyme from P. aeruginosa was more negative than that of cytochrome c (Horio et al., 1961B; Wharton et al., 1973; Shimada and Orii, 1976).

During purification of nitrite reductase, cytochrome  $c_{551}$ and cytochrome  $c_{554}$  were isolated. Cytochromes  $c_{551}$  and  $c_{554}$  from *P. aeruginosa* (Horio *et al.*, 1958) and cytochrome  $c_{552}$  from *M. denitrificans* (Smith *et al.*, 1966) have also been separated. In addition to these species of cytochrome *c*, a blue protein was also recovered from *P. aeruginosa* (Horio, 1958; Gudat *et al.*, 1973). This blue protein, which in its reduced state functioned as a reductant for cytochrome oxidase and nitrite reductase in *P. aeruginosa* (Yamanaka and Okunuki, 1963A; Gudat *et al.*, 1973), was however not detected during the purification of nitrite reductase from *T. denitrificans*.

Catalytic amounts of cytochrome  $c_{551}$ , isolated from *T. denitrificans* during the present investigation, significantly enhanced the sulphide-linked nitrite reductase activity. This effect was quite specific since neither cytochrome  $c_{554}$  nor mammalian cytochrome cstimulated enzyme activity. Thus cytochrome  $c_{551}$  is involved in transferring electrons from sulphide to nitrite. This was further supported by the observation that reduced cytochrome  $c_{551}$  but not mammalian cytochrome c served as an effective reductant for the purified

nitrite reductase. Reduced forms of cytochrome  $c_{551}$ , isolated from *P. aeruginosa* (Horio *et al.*, 1961A; Yamanaka and Okunuki, 1963A) and cytochrome  $c_{552}$  from *M. denitrificans* (Smith *et al.*, 1966) have also been shown to serve as electron donors for purified nitrite reductases and cytochrome oxidases in these microorganisms. In addition as shown herein for nitrite reductase from *T. denitrificans*, the specificity of the enzyme towards cytochrome  $c_{551}$  as a reductant was also observed in *P. aeruginosa* (Yamanaka and Okunuki, 1961A).

The purified nitrite reductase from *T. denitrificans* also exhibited a sulphide-linked cytochrome oxidase activity. As with nitrite reductase, this cytochrome oxidase activity was also stimulated by cytochrome  $c_{551}$  and in its reduced state it was an effective reductant for cytochrome oxidase activity.

The results presented in this thesis indicate that nitrite reductase and cytochrome oxidase have several common properties. These include their identical requirements for reductants, similar effects of various inhibitors of the respiratory chain in P100 fraction, involvement of cytochromes c and d in these reactions and a functional role for cytochrome  $c_{551}$  for both these activities. From these results it can be concluded that as in *P. aeruginosa* (Yamanaka and Okunuki, 1963A) and *M. denitrificans* (Lam and Nicholas, 1961C; Newton, 1969), these two activities are associated with the same enzyme which functions with either  $O_2$  or nitrite as a terminal electron acceptor. The purified cytochrome cd complex of *A. faecalis* was also reported to possess both nitrite reductase and cytochrome oxidase activities (Iwasaki and Matsubara, 1971). In the present investigation, since the bacterium

were grown anaerobically with nitrate, the physiological function of this enzyme is for nitrite reduction. This is confirmed by a higher specific activity of the purified enzyme for nitrite reduction than for cytochrome oxidase activity. On the basis of the turnover numbers for nitrite reductase and cytochrome oxidase activities of the purified enzyme from anaerobically grown cells of *P. aeruginosa* (Yamanaka and Okunuki, 1963A) and *M. denitrificans* (Lam and Nicholas, 1969C; Newton, 1969), it was concluded that in these bacteria also the physiological function of the enzyme was for the reduction of nitrite.

As reported earlier, the sulphide-reduced, purified nitrite reductase was oxidized by either nitrite or O2. On oxidation of the sulphide-reduced enzyme with nitrite, a prominent shoulder appeared at 573 nm which was not observed when O, was the oxidant. The appearance of this band was probably associated with the formation of a complex between the enzyme and a product of nitrite reduction. A similar peak was obtained on flushing the sulphide-reduced enzyme with NO gas but not by either N20, N2 or nitrate. Thus this spectral effect was produced by an interaction between NO and the reduced enzyme. The decrease in the absorbance at 573 nm with time, noted during the present investigation, suggests that NO was reduced further by the purified Rowe et al. (1977) also recorded an absorption nitrite reductase. peak at 572 nm on passing NO through cell-free extracts of P. aeruginosa and they also demonstrated that its disappearance was associated with the production of  $N_0O$  gas. The formation of similar peak at 572 nm was also reported during the oxidation of a reduced nitrite reductase from P. aeruginosa by nitrite (Yamanaka et al., 1961; Yamanaka and Okunuki, 1963B) or after passing NO gas through a purified preparation of nitrite reductase (Yamanaka and Okunuki, 1963B; Shimada and Orii, 1975).

The results of present investigations with *T. denitrificans* also show that in addition to the NO complex with the sulphide-reduced enzyme, two absorption peaks at 535 and 565 nm were produced when NO was passed through the purified enzyme in the absence of sulphide. These peaks were observed at acid pH only (6.5 to 4). These spectral changes were associated with the formation of a complex between the oxidized enzyme and NO, because in the absence of the gas these effects were not observed. Shimada and Orii (1975) also reported the appearance of twin peaks at 529 and 560 nm at acid pH with the oxidized enzyme from *P. aeruginosa*, treated with NO. Similar bands were obtained during the present studies, on passing NO through the dissociated cytochrome cof the cd complex of *T. denitrificans*. Thus the absorption peaks at 530 and 565 nm result from an interaction of NO with the cytochrome ccomponent of the cd complex.

The products of nitrite reduction by the purified enzyme from T. denitrificans were NO and  $N_2O$ . In contrast, the purified nitrite reductase from P. aeruginosa (Walker and Nicholas, 1961), P. denitrificans (Radcliffe and Nicholas, 1968), M. denitrificans (Lam and Nicholas, 1969C) and P. perfectomarinus (Payne et al., 1971) produced NO only. However, highly purified nitrite reductase from A. faecalis (Matsubara and Iwasaki, 1972) reduced nitrite to NO and  $N_2O$ , in agreement with the results reported herein. The reduction of NO to  $N_2O$  by the purified nitrite reductase from A. faecalis may not be physiologically significant in view of the presence of another particulate nitric oxide reductase with a lower  $K_m$  for NO. Whether a separate nitric oxide reductase occurs in T. denitrificans is not known. In P. perfectomarinus nitrite, NO and  $N_2O$  reducing fractions have been isolated (Payne et al., 1971). As reported herein, cytochrome c and haem d were cleaved from the purified nitrite reductase by treating the enzyme with SDS. The purified enzymes from P. *aeruginosa* (Yamanaka and Okunuki, 1963C) and M. *denitrificans* (Lam and Nicholas, 1969C) were dissociated into haem d and a protein containing cytochrome c components by an acidacetone treatment of the enzyme. In the present investigation neither the separated cytochrome c nor haem d, either singly or when mixed together, showed nitrite reductase or cytochrome oxidase activity. This contrasts with a report by Kijimoto (1968) that a cytochrome dfraction, obtained after sucrose-density gradient centrifugation of the SDS-treated nitrite reductase from P. *aeruginosa*, catalyzed the reduction of nitrite. Since their cytochrome d preparation contained appreciable amounts of cytochrome c, it cannot be concluded that the former component alone functions as a nitrite reductase.

On the basis of the properties of nitrite reductase described in this thesis, the following scheme is proposed for the sulphide-linked nitrite reduction in *T. denitrificans:-*

$$s^{2} \longrightarrow Cyt.c_{551} \longrightarrow \overbrace{Cytochrome \ cd} \longrightarrow \overbrace{No_{2}^{2} \longrightarrow NO \rightarrow N_{2}^{0}}^{0_{2}}$$
  
Nitrite reductase

The present investigation indicates that nitrate and nitrite reductases in *T. denitrificans* have some enzymic properties that are similar to those in other denitrifying bacteria. The main important differences are associated with the reduced sulphur compounds which serve as physiological electron donors for these enzymes. This unusual requirement for sulphite and sulphide as the reductants for nitrate and nitrite reductases respectively correlates well with the

unique physiology of this chemolithotrophic bacteria which couples the oxidation of these reduced inorganic sulphur compounds with denitrification.

From the results presented in this thesis, it is concluded that the reducing equivalents from sulphite and sulphide to nitrate and nitrite respectively are transferred *via* different electron carriers.

An integrated electron transfer scheme coupling the oxidation of sulphite and sulphide to the reduction of nitrate and nitrite respectively, involving different carrier systems, is as follows:-

NADH-SH  $\longrightarrow$  Flavin → Cyt.b t.°554  $SO_3^{2-}$ -SH  $\longrightarrow$  Flavin NO  $NO_2$  $NO \longrightarrow N_2O$  $s^{2-} \longrightarrow Cyt.c_{551}$ Cyt.cd 02 Nitrite reductase

#### 5. BIBLIOGRAPHY

Adair, F.W. (1966). Membrane-associated sulphur oxidation by the autotroph Thiobacillus denitrificans. J. Bact. 92, 899-904.

Adams, C.A., Warnes, G.M., and Nicholas, D.J.D. (1971).
 A sulphite-dependent nitrate reductase from Thiobacillus
 denitrificans.
 Biochim. Biophys. Acta 235, 398-406.

Aleem, M.I.H. (1966A). Generation of reducing power in chemosynthesis. III. Energy-linked reduction of pyridine nucleotides in Thiobacillus novellus. J. Bact. 91, 729-736.

Aleem, M.I.H. (1966B). Generation of reducing power in chemosynthesis. II. Energy-linked reduction of pyridine nucleotides in the chemoautotroph Nitrosomonas europaea. Biochim. Biophys. Acta 113, 216-224.

Aleem, M.I.H. (1966C).

Generation of reducing power in chemosynthesis. IV. Energy-linked reduction of pyridine nucleotides by succinate in Thiobacillus novellus. Biochim. Biophys. Acta 128, 1-12.

Aleem, M.I.H. (1969). Generation of reducing power in chemosynthesis. VI. Energy-linked reactions in chemoautotroph Thiobacillus neapolitanus. Antonie van Leeuwenhoek 35, 379-391.

Aleem, M.I.H. (1970). Oxidation of inorganic nitrogen compounds. Ann. Rev. Pl. Physiol. 21, 67-90.

Aleem, M.I.H. (1975). Biochemical reaction mechanisms in sulphur oxidation by chemosynthetic bacteria. Plant and Soil 43, 587-609.

Aleem, M.I.H. (1977). Coupling of energy with electron transfer reactions in chemolithotrophic bacteria. In Microbial Energetics (Haddock, B.A. and Hamilton, W.A. eds.) pp. 351-382. Cambridge Press, London, New York and Melbourne. Aleem, M.I.H., Lees, H., and Nicholas, D.J.D. (1963). Adenosine triphosphate dependent reduction of nicotinamide adenine dinucleotide by ferro-cytochrome c in chemolithotrophic bacteria. Nature 200, 759-761. Alexander, M. (1971). Microbial ecology. In Microorganisms and Biochemistry (John Wiley and Sons. Inc.) pp. 422, New York, London, Sydney and Toronto. Aminuddin, M. (1974). Inorganic sulphur compound oxidation and denitrification in Thiobacillus denitrificans. Ph.D. Thesis, University of Adelaide. Aminuddin, M., and Nicholas, D.J.D. (1973). Sulphite oxidation linked to the reduction of nitrate and nitrite in Thiobacillus denitrificans. Biochim. Biophys. Acta 325, 81-93. Aminuddin, M., and Nicholas, D.J.D. (1974A). An AMP-independent sulphite oxidase from Thiobacillus denitrificans: purification and properties. J. Gen. Microbiol. 82, 101-113. Aminuddin, M., and Nicholas, D.J.D. (1974B). Electron transfer during sulphide and sulphite oxidation in Thiobacillus denitrificans. J. Gen. Microbiol. 82, 115-123. Andrews, P. (1964). Estimation of molecular weights of proteins by Sephadex gel-filtration. Biochem. J. 91, 222-233. Andrews, P. (1965). The gel-filtration behaviour of proteins related to their molecular weights over a wide range. Biochem. J. 96, 595-606. Asano, A., Kaneshiro, T., and Brodie, A.F. (1965). Malate-vitamin K reductase, phospholipid requiring enzyme. J. Biol. Chem. 240, 895-905. Aubert, J.P., Milhaud, G., Moncel, C., and Millet, J. (1958). Existence, isolement et propriétés physicochimiques d'un cytochrome c de Thiobacillus denitrificans. Compte rend. 246, 1616-1619. Aubert, J.P., Millet, J., and Milhaud, G. (1959). Isolation and properties of cytochrome c from Thiobacillus denitrificans. Ann. Inst. Pasteur 96, 559-576.

Baalsrud, K., and Baalsrud, K.S. (1954). Studies on Thiobacillus denitrificans. Arch. Mikrobiol. 20, 34-62.

Barbaree, J.M., and Payne, W.J. (1967). Products of denitrification by marine bacterium as revealed by gas chromatography. Marine Biol. 1, 136-139.

Beijernick, M.W. (1904). Über die bakterien, welche sich in Dunkeln mit kehlens äure als kohlenstoffquelle ernähren können. Zentr. Bakteriol. Parasitenk, Abt. II, 11 : 593-599.

- Bogin, E., Higashi, T., and Brodie, A.F. (1969). Exogenous NADH oxidation and particulate fumarate reductase in Mycobacterium phlei. Arch. Biochim. Biophys. 129, 211-220.
- Bowen, T.J., Happold, F.C., and Taylor, B.F. (1966). Studies on adenosine 5'-phosphosulphate reductase from Thiobacillus denitrificans. Biochim. Biophys. Acta 118, 566-576.
- Bragg, P.D. (1965).
   Purification and properties of menadione reductase from
   Escherichia coli.
   Biochim. Biophys. Acta 96, 263-271.
- Burke, K.A., and Lascelles, J. (1975). Nitrate reductase system in Staphylococcus aureus wild type and mutants. J. Bact. 123, 308-316.
- Burton, K. (1957). Free energy data of biological interest. Ergeb. Physiol. Biol. Chem. Exp. Pharmakol. 49, 275-298.
- Chang, J.B., and Lascelles, J. (1963).
   Nitrate reductase in cell-free extracts of a haemin-requiring
   strain of Staphylococcus aureus.
   Biochem. J. 89, 503-510.

Charles, A.M., and Suzuki, I. (1966). Purification and properties of sulphite: cytochrome c oxido-reductase from Thiobacillus novellus. Biochim. Biophys. Acta 128, 522-534.

Chatelain, R. (1969).

Réduction des nitrites par Alcaligenes odorans var. viridans. Ann. Inst. Pasteur 116, 498-500.

- Chiba, S., and Ishimoto, M. (1973).
   Ferredoxin-linked nitrate reductase from Clostridium perfringens.
   J. Biochem. (Tokyo) 73, 1315-1318.
- Chiba, S., and Ishimoto, M. (1977).
   Studies on nitrate reductase of Clostridium perfringens.
   I. Purification, some properties and effect of tungstate
   on its formation.
   J. Biochem. (Tokyo) 82, 1663-1671.
- Clegg, R.A. (1976). Purification and some properties of nitrate reductase from Escherichia coli Kl2. Biochem. J. 153, 533-541.
- Cole, J.A., and Wimpenny, J.W.T. (1968). Metabolic pathways for nitrate reduction in *Escherichia coli*. *Biochim. Biophys. Acta 162*, 39-48.
- Cox, C.D., and Payne, W.J. (1973). Separation of soluble denitrifying enzymes and cytochromes from Pseudomonas perfectomarinus. Can. J. Microbiol. 19, 861-872.
- Dancey, G.F., and Shapiro, B.M. (1976). The NADH dehydrogenase of the respiratory chain of Escherichia coli. J. Biol. Chem. 251, 5921-5928.
- Davenport, H.E., and Hill, R. (1952). The preparation and some properties of cytochrome f. Proc. Roy. Soc. B 139, 327-345.
- Decker, K., Jungermann, K., and Thauer, R.K. (1970). Energy production in anaerobic organism. Angew. Chemie 9, 138-158.
- DeGroot, G.N., and Stouthamer, A.H. (1970). Regulation of reductase formation in Proteus mirabilis. II. Influence of growth with azide and of haem deficiency on nitrate reductase formation. Biochim. Biophys. Acta 208, 414-427.
- Delwiche, C.D., and Bryan, A.B. (1976). Denitrification. Ann. Rev. Microbiol. 30, 241-262.
- Dixon, M., and Kleppe, K. (1965). D-amino acid oxidase. I. Dissociation and recombination of the holoenzyme. Biochim. Biophys. Acta 96, 357-367.

- Doelle, H.W. (1975).
   Photosynthesis and photometabolism Terminology in bacterial
   metabolism.
   In Bacterial Metabolism. 2nd edition, pp. 84, Academic Press,
   New York, San Francisco, London.
- Downey, R.J. (1962). Naphthoquinone intermediate in the respiration of Bacillus stearothermophilus. J. Bact. 84, 953-960.
- Downey, R.J. (1966). Nitrate reductase and respiratory adaptation in Bacillus stearothermophilus. J. Bact. 91, 634-641.
- Dutton, P.L., Wilson, D.F., and Lee, C.P. (1970). Oxidation-reduction potentials of cytochromes in mitochondria. *Biochemistry 9*, 5077-5082.
- Enoch, H.G., and Lester, R.L. (1974). The role of novel cytochrome *b*-containing nitrate reductase and quinone on the *in vivo* reconstruction of formate-nitrate reductase activity of *Escherichia coli*. *Biochim. Biophys. Res. Commun. 61*, 1234-1241.
- Ellis, R.J. (1964). A rapid assay for sulphite reductase. Biochim. Biophys. Acta 85, 335-337.
- Fewson, C.A., and Nicholas, D.J.D. (1961A). Respiratory enzyme from Micrococcus denitrificans. Biochim. Biophys. Acta 48, 208-210.
- Fewson, C.A., and Nicholas, D.J.D. (1961B). Nitrate reductase from Pseudomonas aeruginosa. Biochim. Biophys. Acta 49, 335-345.
- Forget, P. (1971). Les nitrate-reductases bactériennes solubilization, purification et propriétés de l'enzyme A de *Micrococcus denitrificans*. *Eur. J. Biochem. 18*, 442-450.
- Forget, P. (1974).
  The bacterial nitrate reductases. Solubilization, purification
  and properties of enzyme A of Escherichia coli K12
  Eur. J. Biochem. 42, 325-332.
- Forget, P., and Pichinoty, F. (1964). Influence de la respiration anaérobie du nitrate et du fumarate sur le métabolism fermentaire d'Aerobacter aerogenes. Biochim. Biophys. Acta 82, 441-444.

- Forget, P., and Dervartanian, D.V. (1972). Bacterial nitrate reductases. EPR studies on nitrate reductase A from *Micrococcus denitrificans*. *Biochim. Biophys. Acta 256*, 600-606.
- Fujita, T., and Sato, R. (1966).
  Studies on soluble cytochrome in Enterobacteriaceae.
  Iv. Possible involvement of cytochrome c-552 in anaerobic
  nitrite reductase.
  J. Biochem. (Tokyo) 60, 691-700.
- Garland, P.B., Clegg, R.A., Boxer, D., Downie, J.A. and Haddock, B.A. (1975).

Proton-translocating nitrate reductase of *Escherichia coli*. In *Electron transfer chain and oxidative phosphorylation*. (Quagliarielo, E., Papa, S., Palmieri, F., Slater, E.C., and Siliprandi, N. eds.) pp. 351-358. North Holland Publishing Co., Amsterdam.

- Gauthier, D.K., Clark-Walker, C.D., Garrand, Jr., W.T., and Lascelles, J. (1970). Nitrate reductase and soluble cytochrome c in Spirillum itersonii. J. Bact. 102, 797-803.
- Gent-Ruijters, M.L.W., DeVries, W., and Stouthamer, A.H. (1975).
  Influence of nitrate on fermentation pattern, molar growth
  yields and synthesis of cytochrome b in Propionibacterium
  pentosaceum.
  J. Gen. Microbiol. 88, 36-48.
- Gomori, G. (1955).
  Preparation of buffers for use in enzyme studies.
  In Methods in Enzymology (Colowick, S.P., and Kaplan, N.O.
  eds.) Vol. 1, pp. 138-146. Academic Press, New York.
- Gool, A. van, and Laudelout, H. (1967). Spectrophotometric and kinetic study of nitrite and formate oxidation in Nitrobacter winogradskyi. J. Bact. 93,215-220.
- Gracia, J.L. (1975). Traite de microbiologie des sols. Bull. Inst. Pasteur 73, 167-193.
- Gray, C.T., Wimpenny, J.W.T., Hughes, D.E., and Ranlett, M. (1963). A soluble c type cytochrome from anaerobically grown Escherichia coli and various enterobacteriaceae. Biochim. Biophys. Acta 67, 157-160.
- Greenberg, E.P., and Becker, G.E. (1972).
   A comparison of denitrification by three strains of
   Pseudomonas fluorescens.
   Bacteriol. Proc. 72, 175.
- Gudat, J.C., Singh, J., and Wharton, D.C. (1973). Cytochrome oxidase from Pseudomonas aeruginosa. I. Purification and some properties. Biochim. Biophys. Acta 292, 376-390.

- Hackenthal, E., and Hackenthal, R. (1965). Die nitratreduktion durch zellefreie extrakte aus Bacillus cereus. Biochim. Biophys. Acta 107, 189-202.
- Haddock, B.A., and Kendall-Tobias, M.W. (1975). Functional anaerobic electron transport linked to the reduction of nitrate and fumarate in membranes from *Escherichia coli* as demonstrated by quenching of atebrin fluorescence. *Biochem. J. 152*, 655-659.
- Haddock, B.A., and Jones, C.W. (1977). Bacterial respiration. Bacteriol. Rev. 41, 47-99.
- Hadjipetrou, L.P., and Stouthamer, A.H. (1965). Energy production during nitrate respiration by Aerobacter aerogenes. J. Gen. Microbiol. 38, 29-34.
- Hart, L.T., Larson, A.D., and McCleskey,C.S. (1965). Denitrification by Corynebacterium nephridii. J. Bact. 89, 1104-1108.
- Hemfling, W.P., Trudinger, P.A., and Vishniac, W. (1967). Purification and some properties of sulphite oxidase from Thiobacillus neapolitanus. Arch. Mikrobiol. 59, 149-157.
- Hewitt, E.J., and Nicholas, D.J.D. (1964). Enzymes in inorganic nitrogen metabolism. In Modern Methods of Plant Analysis (Linskens, Y.F., Sanwal, B.D., and Tracey, M.V. eds.), Vol. 7, pp. 67-172, Springer, Berlin.
- Horio, T. (1958). Terminal oxidation system in bacteria II. Some physical and physiological properties of purified cytochromes of Pseudomonas aeruginosa. J. Biochem. (Tokyo) 45, 267-279.
- Horio, T., Higashi, T., Matsubara, H., Kusai, K., Nakai, M., and Okunuki, K. (1958). High purification and properties of *Pseudomonas* cytochrome oxidase. *Biochim. Biophys. Acta 29*, 297-302.
- Horio, T., Higashi, T., Yamanaka, T., Matsubara, H., and Okunuki, K. (1961A). Purification and properties of cytochrome oxidase from Pseudomonas aeruginosa. J. Biol. Chem. 236, 944-951.

- Horio, T., Kamen, M.D., and Klerk, H. De (1961B). Relative oxidation-reduction potentials of haem groups in two soluble double-haem protein. J. Biol. Chem. 236, 2783-2787.
- Hughes, D.E., Wimpenny, J.W.T., and Lloyd, D. (1971). The disintegration of micro-organisms. In *Methods in Microbiology* (Norris, J.R., and Robbins, D.W. eds.) Vol. 5B, pp. 1-54, Academic Press, London and New York.
- Hutchinson, M., Johnstone, K.I., and White, D. (1969). Taxonomy of the genus Thiobacillus: the outcome of numerical taxonomy applied to the group as a whole. J. Gen. Microbiol. 57, 397-410.
- Iida, K., and Taniguchi, S. (1959).
  Studies on nitrate reductase system of Escherichia coli.
  l. Particulate electron transport system to nitrate and its
  solubilization.
  J. Biochem. (Tokyo) 46, 1041-1055.
- Ishaque, M., and Aleem, M.I.H. (1973).
  Intermediates of denitrification on the chemoautotroph
  Thiobacillus denitrificans.
  Arch. Mikrobiol. 94, 269-282.
- Itagaki, E. (1964).
  The role of lipophilic guinone in the electron transport
  system of Escherichia coli.
  J. Biochem. (Tokyo) 55, 432-445.
- Itagaki, E., Fujita, T., and Sato, R. (1961). Cytochrome b<sub>1</sub>-nitrate reductase interaction in a solubilized system from Escherichia coli. Biochim. Biophys. Acta 51, 390-392.
- Itagaki, E., Fujita, T., and Sato, R. (1962). Solubilization and properties of formate dehydrogenase and cytochrome b<sub>1</sub> from Escherichia coli. J. Biochem. (Tokyo) 52, 131-141.
- Itzhaki, R.F., and Gill, D.M. (1964).
  A micro-biuret method for estimating protein.
  Anal. Biochem. 9, 401-410.
- Iwasaki, H., and Matsubara, T. (1971). Cytochrome c 557 (551) and cytochrome cd of Alcaligenes faecalis. J. Biochem. (Tokyo) 69, 847-857.
- Iwasaki, H., and Matsubara, T. (1972).
  A nitrite reductase from Achromobacter cycloclastes.
  J. Biochem. (Tokyo) 71, 645-652.

Kalckar, H.M. (1947). Differential spectrophotometry of purine compounds by means of specific enzymes. J. Biol. Chem. 167, 461-466. Kelly, D.P. (1967). The incorporation of acetate by the chemoautotroph Thiobacillus neapolitanus strain c. Arch. Mikrobiol. 58, 99-116. Kelly, D.P. (1968). Biochemistry of oxidation of sulphur compounds by microorganisms. Aust. J. Sci. 31, 165-173. Kelly, D.P. (1971). Autotrophy: concepts of lithotrophic bacteria and their organic metabolism. Ann. Rev. Microbiol. 25, 177-210. Kijimoto, S. (1968). Studies on Pseudomonas cytochrome oxidase. II. Split of Pseudomonas cytochrome oxidase into haem c- and haem dcomponents. Annual reports of biological works, Faculty of Science, Osaka University 16, 19-33. Kiszkiss, D.F., and Downey, R.J. (1972A). Localization and solubilization of the respiratory nitrate reductase of Bacillus stearothermophilus. J. Bact. 109, 803, 810. Kiszkiss, D.F., and Downey, R.L. (1972B). Physical aggregation and functional reconstitution of solublized membranes of Bacillus stearothermophilus. J. Bact. 109, 811-819. Kluyver, A.J. (1959). Some aspects of nitrate reduction. In High life and work (Kamp, A.F., LaRiviere, J.W.M., and Verhoeven, W. eds.), pp. 483-502. Interscience, New York. Knook, D.L., and Planta, R.J. (1971A). Function of ubiquinone in electron transport from reduced nicotinamide adenine dinucleotide to nitrate and oxygen in Aerobacter aerogenes. J. Bact. 105, 483-488. Knook, D.L., and Planta, R.J. (1971B).

RNOOK, D.L., and Planta, R.J. (1971B). Restoration of electron transport chain in ultravioletirradiated membranes of Aerobacter aerogenes. FEBS Lett. 14, 54-56.

- Knook, D.L., Van't Riet, J., and Planta, R.J. (1973). The participation of cytochromes in the process of nitrate respiration of Klebsiella (Aerobacter) aerogenes. Biochim. Biophys. Acta 292, 237-245.
- Kodama, A., and Mori, T. (1968). Studies on the metabolism of sulphur oxidizing bacterium. V. Comparative studies on sulphur and sulphite oxidizing systems of *Thiobacillus thiooxidans*. *Pl. Cell Physiol. 9*, 725-734.

Kröger, A. (1977). Phosphorylative electron transport with fumarate and nitrate as terminal hydrogen acceptors. In *Microbial Energetics*. Symposium of the Society for General Microbiology (Haddock, B.A., and Hamilton, W.A. eds.) 27, pp. 61-93. Cambridge University Press, London, New York and Melbourne.

- Kuronen, T. and Ellfolk, N. (1972). A new purification procedure and molecular properties of Pseudomonas cytochrome oxidase. Biochim. Biophys. Acta 275, 308-318.
- Lam, Y., and Nicholas, D.J.D. (1969A). Aerobic and anaerobic respiration in Micrococcus denitrificans. Biochim. Biophys. Acta 172, 450-461.
- Lam, Y., and Nicholas, D.J.D. (1969B).
  A nitrate reductase from Micrococcus denitrificans.
  Biochim. Biophys. Acta 178, 225-234.
- Lam, Y., and Nicholas, D.J.D. (1969C).
  A nitrite reductase with cytochrome oxidase activity from
  Micrococcus denitrificans.
  Biochim. Biophys. Acta 180, 459-472.
- Lees, H. (1960). Energy metabolism in chemolithotrophic bacteria. Ann. Rev. Microbiol. 14, 83-93.

Lees, H. (1962). Symposium on autotrophy. IV. Some thoughts on the energetics of chemosynthesis. Bacteriol. Rev. 26, 165-167.

- Lineweaver, H., and Burk, D. (1934). A determination of enzyme dissociation constants. J. Am. Chem. Soc. 56, 658-663.
- London, J., and Rittenberg, S.C. (1964).
  Path of sulphur in sulphide and thiosulphate oxidation of
  thiobacilli.
  Proc. Natl. Acad. Sci. U.S. 52, 1183-1190.

Lyric, D.M., and Suzuki, I. (1970A). Enzymes involved in the metabolism of thiosulphate by Thiobacillus thioparus. I. Survey of enzymes and properties of sulphite : cytochrome c oxidoreductase. Can. J. Biochem. 48,334-343.

Lyric, D.M., and Suzuki, I. (1970B). Enzymes involved in the metabolism of thiosulphate by *Thiobacillus thioparus*. II. Properties of APS-reductase. Can. J. Biochem. 48, 344-354.

MacGregor, C.H., Schnaitman, C.A., Normansell, D.E., and Hodgins, M.G. (1974). Purification and properties of nitrate reductase from Escherichia coli Kl2. J. Biol. Chem. 949, 5321-5327.

Matsubara, T. (1970).
Studies on denitrification. XII. Gas production from
amines and nitrite.
J. Biochem. (Tokyo) 67, 229-235.

Matsubara, T. (1971).
 Studies on denitrification. XIII. Some properties of the
 N<sub>2</sub>O-anaerobically grown cell.
 J. Biochem. (Tokyo) 69, 991-1001.

Matsubara, T. (1975). The participation of cytochromes in the reduction of  $N_2O$  to  $N_2$  by a denitrifying bacterium. J. Biochem. (Tokyo) 77, 627-632.

Matsubara, T., and Mori, T. (1968).
Studies on denitrification . IX. Nitrous oxide, its
production and reduction to nitrogen.
J. Biochem. (Tokyo) 64, 863-871.

Matsubara, T., and Iwasaki, H. (1971). Enzymatic steps of dissimilatory nitrite reduction in Alcaligenes faecalis. J. Biochem. (Tokyo) 69, 859-868.

Matsubara, T., and Iwasaki, H. (1972). Nitric oxide-reducing activity of Alcaligenes faecalis cytochrome cd. J. Biochem. (Tokyo) 72, 57-64.

McCarty, P.L. (1972). Energetics of organic matter degradation. In Water Pollution Microbiology (Mitchell, R. ed.) pp. 91-118, Wiley-Interscience, New York.

- Medina, A., and Nicholas, D.J.D. (1957). Interference by reduced pyridine nucleotides in the diazotization of nitrite. Biochim. Biophys. Acta 23, 440-442.
- Medina, A., and Heredia, C.F. de (1958). Vitamin K-dependent nitrate reductase in Escherichia coli. Biochim. Biophys. Acta 28, 452-453.
- Meriwether, L.S., Marzluff, W.F., and Hodgson, W.G. (1966). Molybdenum-thiol complexes as models for molybdenum bound enzyme. Nature 212, 465-467.
- Milhaud, G., Aubert, J.P., and Millet, J. (1958). Rôle physiologique du cytochrome c de la bactérie chimioautotrophe *Thiobacillus denitrificans*. *Compte rend*. 246, 1766-1769.
- Miyata, M. (1971).
  Studies on denitrification. XIV. The electron donating
  system in the reduction of nitric oxide and nitrate.
  J. Biochem. (Tokyo) 70, 205-213.
- Miyata, M., Matsubara, T., and Mori, T. (1969). Studies on denitrification. XI. Some properties of nitric oxide reductase. J. Biochem. (Tokyo) 66, 759-765.
- Moriarty, D.J.W. (1969). Inorganic sulphur compound oxidation in *Thiobacillus concretivorus*. Ph.D. Thesis, University of Adelaide.
- Moriarty, D.J.W., and Nicholas, D.J.D. (1969). Enzymic sulphide oxidation by *Thiobacillus concretivorus*. *Biochim. Biophys. Acta 184*, 114-123.
- Murphy, M.J., Siegel, L.M., Tove, S.R., and Kamin, H. (1974). Siroheme: a new prosthetic group participating in six-electron reduction reactions catalyzed by both sulphite and nitrite reductases. Proc. Natl. Acad. Sci. U.S.A. 71, 612-616.
- Nagata, Y., Yamanaka, T., and Okunuki, K. (1970). Amino acid composition and N-terminus of Pseudomonas cytochrome oxidase (= Pseudomonas nitrite reductase). Biochim. Biophys. Acta 221, 668-671.
- Najjar, V.A., and Chung, C.W. (1956). Enzymatic steps in denitrification. In Symposium on Inorganic Nitrogen Metabolism (McElroy, W.D., and Glass, B.H. eds.) pp. 260-291, John Hopkins Press, Baltimore.

Nambiar, E.K.S. (1976). Genetic differences in the copper nutrition of cereals. II. Genotypic differences in response to copper in relation to copper, nitrogen and other mineral contents of plants. Aust. J. Agric. Res. 27, 467-477. Nason, A. (1962). Symposium on metabolism of inorganic compounds. II. Enzymatic pathways of nitrate, nitrite and hydroxylamine metabolism. Bacteriol. Rev. 26, 16-41. Nason, A. (1963). Nitrate reductases. In The Enzymes (Boyer, P.D., Lardy, H., and Myrback, K. eds.) pp. 587-607, Academic Press Inc., New York. Nathansohn, A. (1902). Uber eine neue Gruppe von Schwefelbakterien und ihren Stoffwechsel. Mitt. zool. Sta. Neapel 15, 665-680. Newton, N. (1967). A soluble cytochrome containing c-type and a-type haem groups from Micrococcus denitrificans. Biochem. J. 105, 21c-23c. Newton, N. (1969). The two-haem nitrite reductase from Micrococcus denitrificans. Biochim. Biophys. Acta 185, 316-331. O'Reilly, J.E. (1973). Oxidation-reduction potential of the ferro-ferricyanide systems in buffer solutions. Biochim. Biophys. Acta 292, 509-515. Pankhurst, E.S. (1964). Polarographic evidence of the production of polythionates during the bacterial oxidation of thiosulphate. J. Gen. Microbiol. 34, 427-439. Payne, W.J. (1970). Energy yields and growth of heterotrophs. Ann. Rev. Microbiol. 24, 17-52. Payne, W.J. (1973). Reduction of nitrogenous oxides by microorganisms. Bacteriol. Rev. 37, 409-452. Payne, W.J. (1976). Denitrification. Trends Biochemical Sci. 1, 220-222.

reducing fractions from Pseudomonas perfectomarinus. J. Bact. 106, 356-361. Peck, H.D. (1961). Evidence for the reversibility of the reaction catalyzed by APS-reductase. Biochim. Biophys. Acta 49, 621-624. Peck, H.D. (1962). Symposium on metabolism of inorganic compounds. V. Comparative metabolism of inorganic sulphur compounds in microorganisms. Bacteriol. Rev. 26, 67-94. Peck, H.D. (1968). Energy-coupling mechanisms in chemolithotrophic bacteria. Ann. Rev. Microbiol. 22, 489-518. Peck, H.D., Deacon, T.E., and Davidson, J.T. (1965). Studies on APS-reductase from Desulphovibrio desulphuricans and Thiobacillus thioparus. 1. The assay and purification. Biochim. Biophys. Acta 96, 429-446.

Separate nitrite, nitric oxide and nitrous oxide

Payne, W.J., Riley, P.S., and Cox Jr., C.D. (1971).

Peeters, T., and Aleem, M.I.H. (1970). Oxidation of sulphur compounds and electron transport in Thiobacillus denitrificans. Arch. Mikrobiol. 71, 319-330.

Pichinoty, F. (1964).

A propos des nitrate-reductases d'une bacterie denitrifiante. Biochim. Biophys. Acta 89, 378-381.

Pichinoty, F. (1970). Les nitrate-reductases bacteriennes. IV. Regulation de la biosynthese et de l'activite de l'enzyme B. Arch. Mikrobiol. 71, 116-122.

Pichinoty, F., and D'Ornano, L. (1961). Inhibition by oxygen of biosynthesis and activity of nitrate-reductase in Aerobacter aerogenes. Nature 191, 879-881.

Postgate, J.R. (1965). Recent advances in the study of the sulphate-reducing bacteria. Bacteriol. Rev. 29, 425-441.

Radcliffe, B.C., and Nicholas, D.J.D. (1968). Some properties of nitrite reductase from *Pseudomonas denitrificans*. *Biochim. Biophys. Acta 153*, 545-554. Renner, E.D., and Becker, G.L. (1970). Production of nitric oxide and nitrous oxide during denitrification by Corynebacterium nephridii. J. Bact. 101, 821-826.

Rittenberg, S.C. (1969).
The role of exogenous organic matter in the physiology of
chemolithotrophic bacteria.
In Advances in Microbial Physiology (Rose, A.H., and
Wilkinson, J.F. eds.) Vol. 3, pp. 159-196, Academic Press,
London and New York.

- Rosso, J.P., Forget, P., and Pichinoty, F. (1973). Les nitrate-reductases bacteriennes: solubilization, purification et properties de l'enzyme A de Micrococcus halodenitrificans. Biochim. Biophys. Acta 321, 443-455.
- Roth, C.W., Hemfling, W.P., Conners, J.N., and Vishniac, W.V. (1973). Thiosulphate and sulphide-dependent pyridine nucleotide reduction and gluconeogenesis in intact *Thiobacillus neapolitanus*. J. Bact. 114, 592-599.
- Rowe, J.J., Sherr, B.F., Payne, W.J., and Eagon, R.G. (1977). A unique nitric oxide-binding complex formed by denitrifying *Pseudomonas aeruginosa*. *Biochim. Biophys. Res. Commun.* 77,253-258.

Roy, A.B., and Trudinger, P.A. (1970). The Biochemistry of Inorganic Compounds of Sulphur. Ist edition. Cambridge University Press, Great Britain.

Ruiz-Herrera, J., and DeMoss, J.A. (1969).
Nitrate reductase complex of Escherichia coli K12:
participation of specific formate dehydrogenase and
cytochrome b<sub>1</sub> components in nitrate reduction.
J. Bact. 99, 720-729.

Sadana, J.C., and McElroy, W.D. (1957). Nitrate reductase from Achromobacter fischeri. Purification and properties: Function of flavins and cytochrome. Arch. Biochem. Biophys. 67, 16-34.

Sapshead, L.M., and Wimpenny, J.W.J. (1972). The influence of oxygen and nitrate on the formation of the cytochrome pigments of the aerobic and anaerobic respiratory chain of *Micrococcus denitrificans*. *Biochim. Biophys. Acta* 267, 388-397.

Sasarman, A., Purvis, P., and Portelance, V. (1974).
Role of menaquinone in nitrate respiration in Staphylococcus aureus.
J. Bact. 117, 911-913.

- Saxena, J., and Aleem, M.I.H. (1972). Generation of reducing power in chemosynthesis. VII. Mechanisms of pyridine nucleotide reduction by thiosulphate in the chemoautotroph Thiobacillus neapolitanus. Arch. Mikrobiol. 84, 317-326. Saxena, J., and Aleem, M.I.H. (1973). Oxidation of sulphur compounds and coupled phosphorylation in the chemolithotroph Thiobacillus neapolitans. Can. J. Biochem. 51, 560-568. Schlegel, H.G. (1960). Die Eisenbakterien. In Encyclopaedia Plant Physiology (Rufland, W. ed.) vol. v/2, pp. 649-663. Scholes, P.B., and Smith, L. (1968). Composition and properties of the membrane-bound respiratory chain system of Micrococcus denitrificans. Biochim. Biophys. Acta 153, 363-375. Schulp, J.A., and Stouthamer, A.H. (1970). The influence of oxygen, glucose and nitrate upon the formation of nitrate reductase and respiratory system in Bacillus licheniformis. J. Gen. Microbiol. 64, 195-203. Schwartz, A.C. (1973). Terpenoid quinones of the anaerobic Propionbacterium shermanii. I (II, III) Tetrahydromenaquinone - 9. Arch. Mikrobiol. 91, 273-279. Sewell, D.L., and Aleem, M.I.H. (1969). Generation of reducing power in chemosynthesis. V. The mechanism of pyridine nucleotide reduction by nitrite in the chemoautotroph Nitrobacter agilis. Biochim. Biophys. Acta 172, 467-475.
- Shafia, F., Brinson, K.R., Heinzman, M.W., and Brady, J.M. (1972).
  Transition of chemolithotroph Ferrobacillus ferrooxidans to
  obligate organotrophy and metabolic capabilities of glucosegrown cells.
  J. Bact. 111, 56-65.

Shimada, H., and Orii, Y. (1975).
The nitric oxide compounds of Pseudomonas aeruginosa nitrite
reductase and their probable participation in the nitrite reduction.
FEBS Letters 54, 237-240.

Shimada, H., and Orii, Y. (1976). Oxidation-reduction behaviour of the haem c and haem d moieties of Pseudomonas aeruginosa nitrite reductase and formation of oxygenated intermediate at haem d. J. Biochem. (Tokyo) 80, 135-140.
## Showe, M.K., and DeMoss, J.A. (1968). Localization and regulation of synthesis of nitrate reductase in *Escherichia coli*.

J. Bact. 95, 1305-1313.

Smith, L., Newton, N., and Scholes, P. (1966). Reaction of cytochrome c oxidase of beef heart and Micrococcus denitrificans with mammalian and bacterial cytochrome c. In Haems and Haemoprotein (Chance, B., Estabrook, R.W., and Yonetani, T. eds.) pp. 395-403.

Smith, A.J., London, J., and Stanier, R.Y. (1967).
Biochemical basis of obligate autotrophy in blue-green
algae and thiobacilli.
J. Bact. 94, 972-983.

Sone, N. (1974).
The redox reactions in propionic acid fermentation. IV.
Participation of menaquinone in the electron transfer system
in Propionibacterium arabinosum.
J. Biochem. 76, 137-145.

Squire, P.G. (1964).
A relationship between the molecular weights of macromolecules
and their elution volumes based on a model for Sephadex gel
filtration.
Arch. Biochem. Biophys. 107, 471-478.

Stadtman, T.C. (1967). Methane fermentation.

Ann. Rev. Microbiol. 21, 121-138.

Stouthamer, A.H. (1976). Biochemistry and genetics of nitrate reductase in bacteria. Advances in Microbial Physiology 14, 315-375.

Stulen, I. (1970).
Interference of NADH with the reaction of nitrite in nitrate
reductase estimation.
Acta Bot. Neerl. 19 (3), 425-430.

Tabita, R., and Lundgren, D.G. (1971). Utilization of glucose and the effect of organic compounds on chemolithotroph Thiobacillus ferrooxidans. J. Bact. 108, 328-333.

Taniguchi, S., and Itagaki, E. (1960). Nitrate reductase of nitrate respiratory type from Escherichia coli. 1. Solubilization and purification from the particulate system with molecular characterization as a metaloprotein. Biochim. Biophys. Acta 44, 263-279.

Taniguchi, S., Sato, R., and Egami, F. (1956). The enzymatic mechanism of nitrate, nitrite metabolisms in bacteria. In Symposium on Inorganic Nitrogen Metabolism (McElroy, W.D., and Glass, B.H. eds.) pp. 87-107, John Hopkins Press, Baltimore. Tano, T., and Imai, K. (1968). Physiological studies on Thiobacilli. Pt VI. Certain properties of NADH-cytochrome c oxidoreductase of Thiobacillus thiooxidans. Agric. Biol. Chem. 32, 401-404.

Tano, T., Asano, H., and Imai, K. (1968). Physiological studies on Thiobacilli. Part III. The oxidation of thiosulphate by particulate fraction of *Thiobacillus thiooxidans*. Agric. Biol. Chem. 32, 140-143.

- Tate, M. (1968). Separation of myoinositol pentaphosphates by moving paper electrophoresis (MPE). Anal. Biochem. 23, 141-149.
- Taylor, B.F., and Hoare, D.S. (1971). Thiobacillus denitrificans as an obligate chemolithotroph. II. Cell suspension and enzymic studies. Arch. Mikrobiol. 80, 262-276.
- Taylor, B.F., Hoare, D.S., and Hoare, S.L. (1971). Thiobacillus denitrificans as an obligate chemolithotroph. I. Isolation and growth studies. Arch. Mikrobiol. 78, 193-204.
- Thauer, R.K., Jungerman, K., and Decker, K. (1977). Energy conservation in chemotrophic anaerobic bacteria. Bacteriol. Rev. 41, 100-180.

Trudinger, P.A. (1967). The metabolism of inorganic sulphur compounds by Thiobacilli. Rev. Pure Appl. Chem. 17, 1-24.

Trudinger, P.A. (1969).
Assimilatory and dissimilatory metabolism of inorganic
sulphur compounds by micro-organisms.
In Advances in Microbial Physiology (Rose, A.H., and
Wilkinnon, J.F. eds.) vol. 3, pp. 111-158. Academic Press,
London and New York.

Trudinger, P.A., and Kelly, D.P. (1968). NADH-oxidation by Thiobacillus neapolitanus and Thiobacillus strain c. J. Bact. 95, 1962-1963.

Van't Riet, J., and Planta, R.J. (1969).
Purification and some properties of the membrane bound
respiratory nitrate reductase of Aerobacter aerugenes.
FEBS Letters 5, 249-252.

## Van't Riet, J., and Planta, R.J. (1975). Purification, structure and properties of the respiratory nitrate reductase of Klebsiella aerogenes. Biochim. Biophys. Acta 379, 81-94.

- Van't Riet, J., Stouthamer, A.H., and Planta, R.J. (1968). Regulation of nitrate assimilation and nitrate respiration in Aerobacter aerogenes. J. Bact. 96, 1455-1464.
- Van't Riet, J., Knook, D.L., and Planta, R.J. (1972). The role of cytochrome b in nitrate assimilation and nitrate respiration in Klebsiella aerogenes. FEBS Letters 23, 44-46.
- Vega, J.M., Garrett, R.H., and Siegel, L.M. (1975). Siroheme: a prosthetic group of Neurospora crassa assimilatory nitrate reductase. J. Biol. Chem. 250, 7980-7989.
- Vishniac, W. (1952).
  The metabolism of Thiobacillus thioparus. l. The oxidation
  of thiosulphate.
  J. Bact. 64, 363-373.
- Vishniac, W., and Santer, M. (1957). The thiobacilli. Bacteriol. Rev. 21, 195-213.
- Wallace, W., and Nicholas, D.J.D. (1969). The biochemistry of nitrifying organisms. Biol. Rev. 44, 359-391.
- Walker, G.C., and Nicholas, D.J.D. (1961). Nitrite reductase from Pseudomonas aeruginosa. Biochim. Biophys. Acta 49, 350-360.
- Wharton, D.C., Gudat, J.C., and Gibson, Q.H. (1973). Cytochrome oxidase from Pseudomonas aeruginosa. II. Reaction with copper protein. Biochim. Biophys. Acta 292, 611-620.
- White, D.C. (1962). Cytochrome and catalase patterns during growth of Haemophilus parainfluenzae. J. Bact. 83, 851-859.
- Winogradsky, S. (1890). Recherches sur les organismes de la nitrification. Ann. Inst. Pasteur 4, 213-231.
- Worcel, A., Godlman, D.S., and Cleland, W.W. (1965). An allosteric reduced nicotinamideadenine nucleotide oxidase from Mycobacterium tuberculosis. J. Biol. Chem. 240, 3399-3407.
- Yamanaka, T. (1964). Identity of Pseudomonas cytochrome oxidase with Pseudomonas nitrite reductase. Nature 204, 253-255.

Yamanaka, T., and Okunuki, K. (1963A). Crystalline Pseudomonas cytochrome oxidase. 1. Enzymic properties with special reference to the biological specificity. Biochim. Biophys. Acta 67, 379-393.

Yamanaka, T., and Okunuki, K. (1963B). Crystalline Pseudomonas cytochrome oxidase. II. Spectral properties of the enzyme. Biochim. Biophys. Acta 67, 394-406.

Yamanaka, T., and Okunuki, K. (1963C). Crystalline Pseudomonas cytochrome oxidase. III. Properties of the prosthetic groups. Biochim. Biophys. Acta 67, 407-416.

Yamanaka, T., and Okunuki, K. (1974). Cytochromes. In Microbial Iron Metabolism (Neilands, J.B. ed) pp. 349-400, Academic Press, New York.

## Yamanaka, T., Ota, A., and Okunuki, K. (1961). A nitrite reducing system reconstituted with purified cytochrome components of *Pseudomonas aeruginosa*. *Biochim. Biophys. Acta 53*, 294-308.

Sawhney, V., & Nicholas, D. J. D. (1977). Sulphite- and NADH-dependent Nitrate Reductase from Thiobacillus denitrificans. *Microbiology*, *100*(1), 49-58.

## NOTE:

This publication is included in the print copy of the thesis held in the University of Adelaide Library.